

WHAT IS THE IMPACT OF ICT ON MATHEMATICS EDUCATION?

By Daniel Ata-Baah

A DISSERTATION

Presented to the Department of Mathematics Education Program at Selinus University

Faculty Arts & Humanities in fulfilment of the requirements for the degree of DOCTOR OF PHILOSOPHY In Mathematics Education

DECLARATION

I do hereby attest that I am the sole author of this project/thesis and that its contents are only the result of the readings and research I have done. Permission has been obtained from persons and institutions mentioned to include their interviews and their case studies.

ACKNOWLEDGEMENT

My special appreciation goes to the Almighty Jesus for the gift of life and hope. Also my special thanks to Dr. Mari Cruice, from the University of Roehampton for all her time and continuous encouragement to help me to embark on the Doctorate programme. I am thankful to Crisan Cosette at the University College of London, who really helped me to understand the role of digital technology and inspired my curiosity in wanting to explore more into the role of ICT in mathematics education. Also, my special appreciation are extended to Mr Neil Barley, Head of Mathematics Department, Lilian Bailey High School, whose lectures elicited valuable insights about the perceptions of ICT-integrated pedagogy held by most teachers. Additionally, my appreciation and gratitude are extended to my family and two boys Aaron and Moses; the thought of them made me press on to complete this work despite the hard and difficult times I went through in embarking on this research work.

I dedicate this thesis to Aaron and Moses and wish them success in their educational pursuit.

ABSTRACT

The analysis is based on a detailed investigation of primary data collected through case studies, questionnaires and semi-structured interviews in secondary schools in south London. The pedagogy of mathematics education is explored in order to understand the issues surrounding ICT integration. This study shows that despite the perceived advantages of ICT integrated pedagogy, there are barriers that are closely connected to teachers' pedagogical beliefs at work. The lack of ongoing professional training programmes, and curriculum expectations were admitted by most educators as among the factors that affect their attitudes towards ICT pedagogy. Further empirical research is suggested to advance the exploration of the role of subject leaders in ICT-integrated pedagogy. This research provides an in-depth analysis of the exploration of own practice and mathematics teachers perceptions on an integrated ICT into the teaching. The analysis on own practice was presented through the lenses of TPACK and the Knowledge Quartet. Some works that have been used in the past, such as the works of Rowland, Ruthven, and Mishra are used in this study, to provide the basis for comprehension of the discussion at hand.

С	on	ten	ts
-	~ ~ ~	~~~	

DECLARATION	2
ACKNOWLEDGEMENT	3
ABSTRACT	4
CHAPTER 1: INTRODUCTION	10
1.1 Introduction	10
1.2 Aim of study	10
1.3 Significance and professional relevance	11
1.4 Research Justification	11
1.5 Research Questions	12
1.6 Structure of the study	12
1.7 Definition of terms	13
1.8 Chapter summary	13
CHAPTER 2: LITERATURE REVIEW	14
2.1 Introduction	
2.2 Reasons for Education in schools	
2.2.1 Education for individual reasons	14
2.2.2 Education for social harmony	
2.2.3 Education to develop a human capital	15
2.2.4 Education as a need to standardized learning	16
2.2.5 Education as a national curriculum	16
2.3 The mathematics education	17
2.3.1The mathematics curriculum in UK	18
2.3.2 What is a mathematical proof/truth?	19
2.3.3 Visualisation and proofs	22
2.3.4 Theorems and proofs	
2.3.5 Inductive reasoning	23
2.3.6 Deductive reasoning	23
2.4 The benefit of ICT in education	24
2.4.1 Excitement and confidence	25
2.4.2 Aid lesson planning with wide range of resources	25
2.4.3 Improved the attainment of children	
2.4.4 Enhance interactivity and exploration	27
2.5 Barriers to integration of ICT in pedagogy	27
2.5.1 Lack of school and curriculum resources	
2.5.2 Software and Hardware problems	29

2.5.3 Lack of experience with ICT tools	
2.5.4 Negative attitudes	
2.5.5 Lack of training	
2.6 ICT facilitators in education	
2.6.1 The availability of ICT tools	
2.6.2 Experience and knowledge of using ICT	
2.6.3 School leadership	
2.6.4 Opportunities for training	
2.7 ICT and mathematics education	34
2.7.1 Constructive reasoning	
2.7.2 Conjectures and efficient learning	
2.7.3 Guaranteed feedback and interactive support	
2.8 GeoGebra as a digital tool in mathematics education	
2.8.1 Merits of using GeoGebra	
2.8.2 Demerits of GeoGebra	
2.9 Technological Pedagogical and Content knowledge (TPACK)	
2.9.1 Content knowledge (C)	
2.9.2 Pedagogical content knowledge (PCK)	
2.9.3 Limitation of TPACK	
2.10 Chapter Summary	41
CHAPTER 3 RESEARCH METHODOLOGY	42
CHAPTER 3 RESEARCH METHODOLOGY	42
CHAPTER 3 RESEARCH METHODOLOGY 3.1 Introduction 3.2. Research design	42 42 42
CHAPTER 3 RESEARCH METHODOLOGY 3.1 Introduction 3.2. Research design 3.3 Lesson Design: TPACK and KQ	42 42 42 43
CHAPTER 3 RESEARCH METHODOLOGY 3.1 Introduction 3.2. Research design 3.3 Lesson Design: TPACK and KQ 3.4 Design tool	42 42 42 43 43
CHAPTER 3 RESEARCH METHODOLOGY 3.1 Introduction 3.2. Research design 3.3 Lesson Design: TPACK and KQ 3.4 Design tool 3.5 Interpretivism	42 42 42 42 43 43 43 43
CHAPTER 3 RESEARCH METHODOLOGY 3.1 Introduction 3.2. Research design 3.3 Lesson Design: TPACK and KQ 3.4 Design tool 3.5 Interpretivism 3.6 Research approach	42 42 42 43 43 43 44 44
CHAPTER 3 RESEARCH METHODOLOGY. 3.1 Introduction. 3.2. Research design 3.3 Lesson Design: TPACK and KQ 3.4 Design tool. 3.5 Interpretivism 3.6 Research approach 3.7 Methodological approach	42 42 42 42 43 43 43 43 44 45
CHAPTER 3 RESEARCH METHODOLOGY. 3.1 Introduction	42 42 42 43 43 43 44 44 45 45
CHAPTER 3 RESEARCH METHODOLOGY. 3.1 Introduction	42 42 42 43 43 43 44 44 45 45 45 46
CHAPTER 3 RESEARCH METHODOLOGY. 3.1 Introduction. 3.2. Research design 3.3 Lesson Design: TPACK and KQ. 3.4 Design tool 3.5 Interpretivism 3.6 Research approach 3.7 Methodological approach. 3.8 Data collection 3.9 Questionnaires. 3.10 Participants	42 42 42 43 43 43 44 45 45 45 45 45 45 47
CHAPTER 3 RESEARCH METHODOLOGY. 3.1 Introduction	42 42 42 43 43 43 44 45 45 45 45 45 45 47 47
CHAPTER 3 RESEARCH METHODOLOGY. 3.1 Introduction. 3.2. Research design 3.3 Lesson Design: TPACK and KQ 3.4 Design tool 3.5 Interpretivism 3.6 Research approach 3.7 Methodological approach 3.8 Data collection 3.9 Questionnaires. 3.10 Participants 3.10.1 Students 3.10.2 Teachers	42 42 42 43 43 43 44 44 45 45 45 45 45 46 47 47 47 48
CHAPTER 3 RESEARCH METHODOLOGY. 3.1 Introduction 3.2. Research design 3.3 Lesson Design: TPACK and KQ 3.4 Design tool 3.5 Interpretivism 3.6 Research approach 3.7 Methodological approach 3.8 Data collection 3.9 Questionnaires 3.10 Participants 3.10.1 Students 3.10.2 Teachers 3.11 Materials/research tools	42 42 42 43 43 43 44 44 45 45 45 46 47 47 47 47 47 50
CHAPTER 3 RESEARCH METHODOLOGY. 3.1 Introduction. 3.2. Research design 3.3 Lesson Design: TPACK and KQ 3.4 Design tool 3.5 Interpretivism 3.6 Research approach 3.7 Methodological approach. 3.8 Data collection 3.9 Questionnaires. 3.10 Participants. 3.11 Materials/research tools 3.12 Recording interview data	42 42 42 43 43 43 43 44 45 45 45 45 45 47 47 47 47 48 48 50 50
CHAPTER 3 RESEARCH METHODOLOGY. 3.1 Introduction 3.2. Research design 3.3. Lesson Design: TPACK and KQ 3.4 Design tool 3.5 Interpretivism 3.6 Research approach 3.7 Methodological approach 3.8 Data collection 3.9 Questionnaires 3.10 Participants 3.10.1 Students 3.11 Materials/research tools 3.12 Recording interview data	42 42 42 43 43 43 44 45 45 45 45 45 45 45 45 45 45 45 45
CHAPTER 3 RESEARCH METHODOLOGY 3.1 Introduction 3.2. Research design 3.3 Lesson Design: TPACK and KQ 3.4 Design tool 3.5 Interpretivism 3.6 Research approach 3.7 Methodological approach 3.8 Data collection 3.9 Questionnaires 3.10.1 Students 3.10.2 Teachers 3.11 Materials/research tools 3.13 Recording lesson episodes' data	42 42 42 43 43 43 43 44 45 45 46 47 48 50 50 51
CHAPTER 3 RESEARCH METHODOLOGY. 3.1 Introduction 3.2. Research design 3.3 Lesson Design: TPACK and KQ 3.4 Design tool 3.5 Interpretivism 3.6 Research approach 3.7 Methodological approach 3.8 Data collection 3.9 Questionnaires 3.10 Participants 3.11 Materials/research tools 3.12 Recording interview data 3.13 Recording lesson episodes' data 3.14 Case studies	42 42 42 43 43 43 44 45 45 46 47 48 50 50 51

3.15.1 Triangulate data	
3.15.2 Coding and analysis teachers responses	53
3.16 Research ethics	54
3.17 Limitations of the research	
3.18 Chapter summary	55
CHAPTER 4: DATA ANALYSIS	56
4.1 Introduction	
4.2 Rationale of analysis for the interviews	
4.3 Possible outcomes	
4.4 Dissemination	
4.5 Case study 1: Circle theorem	
4.6 Case study 2: Linear functions	
4.6.1 Lesson Episode 1: Drawing Graphs	
4. 6. 2 Lesson Episode 2: Slope and y-intercepts	
4.7 Teachers Interview questions	
4.7.1 Part 1/Code 1: Benefits of ICT pedagogy	
4.7.2 Part 2/Code 2: Barriers to ICT pedagogy	
4.7.3 Part 3/Code 3: Facilitators of ICT pedagogy	
4.8 Teachers: Questionnaires	
4.9 Chapter summary	67
CHAPTER 5: EVALUATIONS AND FINDINGS	68
5.1 Introduction	
5.2 Rationale for the choice of GeoGebra	
5.3 Lesson design	
5.4 Frame work for ICT lesson	69
5.5 Lesson plans	
5.5.1 Experience of teacher with ICT	
5.5.2 Objectives of the lesson	
5.4.3 Consulted wide range of resources	
5.4.4 Experience of students with technology	72
5.5 Case study 1: lesson on Circle theorem	
5.5.1 Review previous lesson	
5.5.2 Misconceptions were corrected	72
5.5.3 Access to online resources	
5.5.4 Collective learning	
5.5.5 Facilitated assessment	
5.5.6 Increased learning confidence	

5.5.7 Exploration	74
5.6 Lesson report	75
5.6.1 Task 1: The angle at the centre of a circle is twice the angle at the circumfer of the circle from the same arc.	rence 75
5.6.2 Task 2: Angles inscribed in a semicircle are right angles	75
5.6.3 Task 3: Angles in the same segment are equal	76
5.6.4 Task 4: Cyclic Quadrilateral Theorem	77
5.6.5 Conclusion	77
5.7 Case study 2: Lesson in Linear functions	78
5.7.1 Introduction	78
5.7.2 Task 1: Introduction lesson to linear functions using GeoGebra	79
5.7.3 Task 2: To sketch a linear graph when given the gradient and point of interce	ption 79
5.7.4 Task 3: To construct a table of values for given linear functions and dr linear graph	72 raw a 80
5.7.5 Lesson report	80
5.7.6 Pupil's response	81
5.7.7 Impact on learning functions with ICT on mathematics	81
5.7.8 Conclusion	84
5.8. Research question 1: What are the benefits of ICT in Mathematics education?	84
5.8.1 Sustained student interest	84
5.8.2 Aid lesson planning with wide range of resources	85
5.8.3 Improved the attainment of children	85
5.8.4 ICT enhance interactivity and exploration	85
5.9 Research question 2: What are the barriers to ICT in mathematics educations?	86
5.9.1 Lack of curriculum resources	86
5.9.2 Software and hardware problems	86
5.9.3 Lack of experience with ICT tools	87
5.9.4 Negative attitudes	87
5.10 Research question 3: What are the facilitators to ICT into mathematics education	?87
5.10.1 The availability of ICT tools	87
5.10.2 Experience and knowledge of using ICT	88
5.10.3 School leadership	88
5.10.4 Opportunities for training	89
5.11 Chapter Summary	89
CHAPTER 6: CONCLUSION	90
6.1 Introduction	90
6.2 Effect digital tool	90
6.3 TPACK Framework	91

6.4 Lesson learned	
6.5 Summary of findings	
6.6 Limitation and difficulties	
6.7 Recommendations for further research	
Bibliography	95
Appendices	
Appendix 1: Case study 1	
Appendix 2: Activity Sequence:	
Appendix 3: Task: Circle Theorem	
Appendix 4: Case study 2	
Appendix 5: Coding using TPACK	
Appendix 6: Snapshot from lessons	
Appendix 7: Interview scripts -students	
Appendix 8: Letter to Head teacher	
Appendix 9: Letter to Teacher	
Appendix 10: Consent form	
Appendix 11: Participation Consent Form	
Appendix 12: Participant Debrief	
Appendix 13: Health & Safety Risk Assessment	
Appendix 14: Interviews and Documentary Analysis Schedules	
Appendix 15: Interview Transcript 1 – Teacher 1	
Appendix 16: Interview Transcript 2 – Teacher 2	
Appendix17: Interview Transcript 3 _Teacher 3	
Appendix 18: Interview Transcript 4 – Teacher 4	
Appendix 19: Interview Transcript 5 – Teacher 5	
Appendix 20: Teachers questionnaires	
Tables	
Table 1 Mathematics teachers: interviews	
Table 2 Mathematics teachers: Questionnaires	
Table 3: Computer sharing	
Table 4: Summary data analysis using SPSS	

CHAPTER 1: INTRODUCTION

1.1 Introduction

This section gives an over view of the aim of the study, significance and professional relevance of embarking on this research. Also discussed here is the justification of undertaking this research through the chosen research questions. Finally, the structure of the study is subsequently revealed.

1.2 Aim of study

My appreciation in the use of ICT in lessons has made me question the lack of ICT in mathematics education. It is this frustration that has led me to pursue this empirical investigation to establish the role of ICT-integrated pedagogy in lessons through case studies, questionnaires and interviews. In the midst of falling educational standards as reported by UNESCO (2005), ICT-integrated lessons can raise attainment. Also, ICT integration pedagogy is important to facilitate a generation of critical thinking ability, decision making ability and effective communication (Valtonen et al., 2015).

The topic for the research was chosen in order to best understand the implications to enhance professional efficiency and to aid the development of mathematics teachers. Where ICT is integrated well, it has been shown to enhance pupils' achievements and understanding mathematics. Using ICT is now deemed by parents and educators as 'an integral part of providing a high-quality education" (U.S. DOE, 2003: 3).

Additionally, the central idea of this research paper is to add to the existing research through exploration on the impact of integrating ICT in Mathematics education. These aims will be achieved in the following ways;

First, I will explore the role of mathematics education and the role of ICT in education. Then, I will present a theoretical overview of teachers' pedagogical beliefs as a vital starting point. Afterwards, I aim to use own experience and views of mathematics teachers across selected schools in south London to established the impact of ICT integration in mathematics lessons and propose understanding of the implications to schools for future research.

1.3 Significance and professional relevance

The use of different ICT tools has significantly changed teaching practice in secondary schools in western countries (UNESCO, 2002), as ICT integration can make learning more interesting and easier for students (Buabeng-Andoh, 2012). The use of various ICT tools can also increase the pace of learning for all students. The knowledge of the role of ICT will in no doubt contribute to professional practice and impact on pedagogical choices. With this in mind, the research will endeavour to appreciate existing study on the topic to review own perception and significance on the impact of ICT tools on secondary mathematics pedagogy.

Koh and Chai (2014) are of the view that educational institutes in their current form are not able to meet the expectations of teachers in terms of ICT pedagogy knowledge. As a subject leader and teacher, the awareness of the benefits of ICT requires an immediate response into ways in which ICT can be embedded into teaching and learning as directed by the school curriculum.

Thus, from a professional point of view, the integration of ICT into pedagogy will facilitate the use of modern technologies to improve teaching. From my professional experience, it is evident that the use of ICT equips teachers to reduce the effort and time taken for the management of students' learning (Buabeng-Andoh, 2012). Also, the use of ICT can ease the information collection process and the overall process of lesson planning. Furthermore, the use of new technologies in ICT pedagogy likewise enables like-minded professionals to become creative and develop the ability to manage the process of acquiring knowledge (Livingstone, 2012).

1.4 Research Justification

The aim of this research is to explore the impact of integrating ICT into the pedagogy of secondary mathematics education. This will be done by bringing together the findings and the key points from available literature on ICT in Mathematics education. The initiatives related to attitudes that promote effective ICT pedagogy will be reviewed. In addition, the study will provide information on for schools to integrate ICT pedagogy in mathematics education. Finally, it will offer insight into the differences and exclusive characteristics of the learning needs of practitioners.

1.5 Research Questions

The effective application of ICT in education is influenced by the perceptions of teachers (Betz and Fassinger, 2011). Teachers' perceptions have become vital in using ICT tools in an era in which modern technologies are used to advance pedagogies in schools (UNESCO, 2005). The research questions were developed to explore into detail on the main research question, 'what is the impact of ICT use in mathematics education?' Although, Higgins et al., (2005) states that the benefit of teaching through ICT enhances learning, however, commenting on the obstacles of ICT-integrated lessons, Moseley et al., (1999) cited in Higgins (2005) lament the reluctance of ICT use by teachers due to the lack of ICT experience and poor attitudes. The attitudes of teachers towards ICT integration have also been associated with the perception of barriers to using ICT (Newhouse, 2001). Further, Lim and Khine (2006) argue that the problem is worsening with the absence of the required expertise by teachers to facilitate the delivery of ICT integration in schools.

The sub division of the research questions again seek to offer more insights into the entire analysis. The emerging knowledge from this study optimally is aim to enlighten schools and policy makers on the implications of using ICT in mathematics education.

The following research questions emerged for exploration in this study:

- What are the benefits of ICT in Mathematics education?
- What are the barriers to ICT in mathematics education?
- What factors will facilitate ICT in Mathematics education?

1.6 Structure of the study

This dissertation consists of five further chapters as follows: Chapter 2 presents a review of the literature on effective ICT pedagogy. Chapter 3 outlines the research methodology used and the chosen approach with its justification. Chapter 4 elaborates on the justification of the chosen data analysis method and presents findings from both the questionnaires and the semi-structured interviews. Chapter 5 provides a discussion and analysis of findings, and Chapter 6 contains the final conclusions with recommendations.

1.7 Definition of terms

- Teacher knowledge: This entails the technology put into practice by the teachers in solving problems or inventing useful tools in teaching.
- Teachers' 'use' of technology: This entails the application of technological knowledge by the teachers in education to achieve a specified goal.
- Information Technology: The use of software and computer hardware for teaching.
- Communications Technology: All Information, Communication, and Technology (ICT) equipment for communication in education.
- Implementation: The application and operation of ICT equipment in schools.
- Technology: This is the deployment of scientific and technical knowledge aimed at practical purposes, more specifically in industries and in institutions.

The literature review that follows explores publications in the field of educational technology, including research papers, journals, reports, and published databases, and draws conclusions regarding current studies on ICT integration in mathematics education.

1.8 Chapter summary

The section provided the aims of the study and the significance for undertaking the research. Furthermore, the rationales behind the research questions with the structure of the study were addressed to provide a course of the study. Also discussed briefly were the definition of terms associated with information communication and technology to aid the understanding of the research.

CHAPTER 2: LITERATURE REVIEW

2.1 Introduction

I believe that among the aims of the mathematics education is the intention to provide the skills needed to succeed on the modern job market. Morden technologies have a role to play in making this dream a reality. In this section, first, I will engage with the theory related to education and the curriculum in schools. Second, I will recognise the role of ICT within the mathematics pedagogy and finally, I will draw out any known framework aiding the effective design of an ICT in mathematics lesson with an associated technological device.

The following research questions guides this study:

- What are the benefits of ICT in Mathematics education?
- What are the barriers to ICT in mathematics education?
- What factors will facilitate ICT in Mathematics education?

2.2 Reasons for Education in schools

Historically, the vision of education existed for rulers who continued through to most of the twentieth century. After 50 years of the British comprehensive school education, family welfare still remains an important determinant of how well a child does in school (Hatcher in Cole, 2006). This sentiment of social economic factors influencing the academic progress of learners is obvious in today's educational system. The fact still remains that the greatest determinant of a school's success is the proportion of children from poor social backgrounds (Whitty, 2002). It is no surprise that studies of educational success in the UK consistently show correlations with social background, suggesting that while the aim of education can be achieved through the curriculum provided by the schools, in most cases, the effect of the education system has been producing the same knowledge over several decades (Ross, 2000).

2.2.1 Education for individual reasons

The aim of education, according to Capel et al., (2013) is seen as to provide individuals with the key processes needed to succeed in everyday life. The aim of education has changed over time and a historical perspective by Harris (1995) shows that 'education' contents are changing and becoming personalised, despite the influence of politics on the curriculum. For Corrigan, et al., (2007), education provides the mechanism where individuals are offered the opportunities, knowledge and required skills to be a self-independent. The ultimate purpose of education however, according to Tyler (1949), is not to 'cover' particular content but should be to define attitudinal aims that ensure the opportunity of individual's progress. Reiss (2007) also indicates that education seeks to develop the benefits of individuals as well as the community. This aim of education has brought about much divide between the academic community and politicians on whether the aim of education is being fulfilled. In light of this tension, the aim of education has been used to describe all activities that define the human existence, including happiness, opportunities and the ability to make decisions (Michael and White, 2014).

2.2.2 Education for social harmony

Plato advocates the aim of education as the maintenance of social harmony through communal knowledge (Wheelahan, 2015). For Hutchins (1968) the purpose of education is to propagate the cultural truth in society and according to Rousseau (1968) the need for education is to generate social freedom to understand the meaning of life and the environment. Truly, is it not the aim of education to separate the intelligent from the less intelligent and consequently offer the intellectuals the ability to rule the less intelligent through the power offered by the curriculum? This sentiment was expressed by Bernstein (1977) when he pointed out that the prevailing aim of the education system had been to redistribute knowledge to enhance the principles of social control.

2.2.3 Education to develop a human capital

The human capital is defined by the OECD as the knowledge, skills, competencies and attributes embodied in individuals that facilitate the creation of personal, social and economic well-being (OECD, 2007, p. 29). Such development of human capital led to over half of the percentage of economic growth from the previous decades in countries such as Germany, Italy, Greece and the Netherlands in the 1980's (OECD, 2000). On a whole, each extra year of full-time education in the OECD countries corresponded to about a 10 per cent rise in human capital that was associated with about a 6 per cent increase in output per capita (OECD, 2001). The economic success of countries is based on the efficient utilization of both

tangible and intangible resources, including the skilled labour force provided through the educational (Capel et al., 2013).

2.2.4 Education as a need to standardized learning

There is more progress in advocating for equal aims for all as education is too important to society and cannot be understated and restricted to the few (Leask el al., 2013). Many developments over the years to promote equal opportunity of education for all, regardless of race, sex, gender and religious affiliations, are embarked by various governments in the pursuit of eradicating the historical divisions in education. For example, in 1990 the Department of Education and Science (DES, 1978) introduced the integration of SEN children into mainstream education, after decades of separation in special education provisions (Leask el al., 2013).

The debate still continues whether the policy on equal aims for everyone in education is still possible as schools continue to embark on segregated policy in the name of appreciating the diverse needs of learners. Schools continue to promote single sex education within a mixed school, to promote achievements and segregate ability at the point of admissions. The critics are quick to conclude that such a system will promote education integration, in the midst of acknowledging the peculiarity of learners needs. Ability selection has further characteristics of endorsing segregation in the class system. Although the previous regimes criticised the segregation practices of the past, and promised to reform the equality aim of education, most schools have gone backwards in the name of promoting standards in education by acknowledging the differences in learners (Leask et al., 2013).

2.2.5 Education as a national curriculum

Among the aims of the national curriculum in England is to make children "learn and achieve; and to promote pupils' spiritual, moral, social, cultural and emotional well-being" (DfE, 2013, p. 6). Added aims in the 2008 National Curriculum generally became the starting point of the curriculum design that ensured that all children enjoyed learning and would become successful learners. It especially directed that young people became progressive and also safe, confident and learned in a healthy-environment (DfE, 2012b). We can take a closer look at what the aim of education is by understanding the reasons for the creation of the

curriculum through which the aim of education is realised. Capel et al., (2013) questioned whether all the key concepts to be gained through education could be used for why education can be captured by just the acquisition of skills. The aims of education defined how the curriculum should be shaped and formed (Reiss & White, 2014).

According to Young (2014) the curriculum should be defined as what is possible and not act as a restriction on students learning, in relation to discrete teaching of subjects and/or cross curricular teaching. Young proposed again that the current crises in curriculum theory can be overcome, not only by providing a curriculum that does not inform learners of past knowledge, but by also informing them of present and future knowledge (Young, 2013). However, if the curriculum as seen by Ross (2000) was content-based and consisted of present and future knowledge, who could deliver such content based curriculum? Since neither the content nor knowledge-based approach was considered as an appropriate medium of education (Skilbeck, 1976).

The importance of the curriculum, according to (Dewey, 1897), was not to identify it with a specific subject but should consist of the experiences of the children own social activities. In developing a curriculum Michael and White (2014) questioned whether the aim of the curriculum should guide the design of the curriculum. However, Richard Peters (1959) reiterated that the need of a curriculum cannot be sought, hence it is aimless. According to Aberdeen Standish, the notion that education is aimless makes the concept of defining a curriculum with an aim unattainable (*Standish*, 1999; Reiss, & White, 2014). The curriculum should aim to provide a medium through which the educational system can appreciate the previous knowledge acquired by learners and progress learners, with fresh knowledge to provide children with differentiated understanding including emotions support and the sense of the community (Schleicher, 2015).

2.3 The mathematics education

The mathematical education paradigm is not speculative to just a body of an isolated knowledge but are associated with the processes of enquiry into the representations of both personal and published knowledge. The mathematical philosophy on theorems and Proofs is not just the analysis of merely significant units but complex connections and collaborations

between individuals, culture, social structures, unrestrained and other practices (Brown, 1995; Ernest, 1991).

This section focuses the current mathematics education in UK and then discusses the role of theorems and Proofs, to gain mathematics knowledge. It uses the example on the Pythagoras theorem to demonstrate how mathematics knowledge is perceived. The subsequent section discusses what constitutes a mathematical proof/truth and a review on the role of inductive reasoning and deductive reasoning in mathematics education, the section concludes with the use of diagrams and visualisation to gain mathematics knowledge.

2.3.1The mathematics curriculum in UK

The secondary school mathematics curriculum in England aims to develop a student's ability to theorise, inquire, reason and communicate mathematically (DFE, 2014). The mathematical knowledge provides the foundation through which the world can be understood by offering skills needed for enjoyment (DFE, 2014). It is assumed that the knowledge gained from studying the curriculum will enable young people to succeed in post-16 studies. Stephen Ball (1990) and Paul Ernest (1991) insist that a curriculum in mathematics should consist of a more traditional set of facts and skills that concentrates on ensuring that the basic knowledge on written algorithms for the primary mathematical operations are adhered to. Focusing largely on the skills as an 'industrial trainer' was commonly viewed by many prominent politicians who insisted that the mathematics curriculum should enable learners to appreciate knowing the times-tables and be able to perform sums as the core objective of the mathematical module (Brown, et al., 2000). As pointed out by Nick Gibb, a former conservative minister, the need to perform mathematical procedures and the ability to recollect the times-tables was the needed skills that the mathematics curriculum provided at the primary level.

Michael Gove (2010), the former educational secretary, viewed the ability to learn concepts in the mathematics curriculum as absurd and insisted it was insignificant to learning maths (Brown, et al., 2000). Bourdieu and Passeron (1990) did not advocate and support the initiative of an alternate curriculum, in the midst of addressing the crises in the present curriculum. Instead, they advocated that it should provide pupils with the ability to gain knowledge in transferable skills. Students' performance on mathematics has significantly impacted on a school's resources to ensure higher attainment. The government initiative to promote academic rigour came after the publication of the performance by OCED countries, which saw a decline in England's position suggesting that schools perhaps needed to promote more challenging curriculum contents. The clear absence of the government explicitly defining the expectation of 'an academic rigour' in the national curriculum, as pointed out by many reviews, suggests contention (White & Bramall, 2000).

2.3.2 What is a mathematical proof/truth?

The meaning of mathematical proof has been debated by many mathematical authors over the centuries. According to Todd CadwalladerOlsker in the article *What Do We Mean by Mathematical Proof?* (2011), mathematical proofs are activities embarked on by individuals or institutions to justify a mathematical concept. The author further alleged that there is the widely held view as to whether mathematical proofs are 'discovered' or 'created'; in either case, it is people who discover or create mathematics. Hersh (1993) and others have contended with the formal meaning of whether a mathematical proof is important to the mathematics pedagogy. According to CadwalladerOlsker (2011), the issue of mathematical proofs 'continues to be negotiated and to evolve' (p. 34). Here, the notion is expressed that all mathematical propositions are not known to be true or false until they have been proved.

In the view of Rota, a mathematical proof involves a sequence of steps that lead to the desired mathematical conclusion. Again, he emphasises realisation that the rules and sequences of steps to be followed to achieve the desired results have not changed over the centuries (Rota, 1997). Rota agrees with what most authors indicate to be the aim of all proofs: to follow a series of logical steps to justify a mathematical fact. Nevertheless, Rota indicates that this is unrealistic in defining what most mathematicians actually think about mathematical proofs.

There are various selections involved in conjecturing for instance the Pythagorean Theorem and establishing its proof. To guide the choices that mathematicians make there are background results published in literature on the theoretical ideas, which may be detailed or may be vague. However, as time progresses various literatures are produced on the same topic and definitions of what was once an important theorem also change. For instance, while most mathematicians working on proving the Pythagorean Theorem are guided by the ideas of Euclid, some use geometry and others use the guiding principles of circles. For example, Salame et al., (2016) used geometry to prove Pythagoras' theorem, as shown below. The most elementary proof of the Pythagorean Theorem can be done using a square with equal sides and lengths.



The figure below proves that the inside angle θ of the quadrilateral is a right angle. Proportionally, all the inside angles are right angles.



Therefore, the quadrilateral forms a square with an area of c^2 . The fact that the original square is $4 \times$ in the area means it is equal to the area of the quadrilateral. Which means $(a + b)^2 - 4 \times c^2$



area is equal to c^2 . In this case

But the area of the triangle of width *a* and height *b* is $\frac{1}{2}ab$. Therefore:

 $(a + b) 2 - 4 \times \frac{1}{2} ab = c^{2}$ $a^{2} + 2 ab + b^{2} - 2 ab = c^{2}$ $a^{2} + b^{2} = c^{2}$

Mathematicians agree that proofs are an important element of learning mathematics and should be part of all students' mathematical experiences at all levels (Waring, 2000) in the early school years (Greenes, 2003). Waring (2000) and Greenes (2003) noted that students' work in the early years emphasises proof as an explanation of the structure of a known pattern. Also noted was the ability for teachers to use proof activities to advance mathematical understanding, particularly using proofs consisting of patterns to advance students' reasoning as they progress through to the higher school years. In the light of the different stages of proofs that advance the understanding of mathematics education, Waring (2000) mentioned three key distinguishers, starting from the moment of appreciation of the

reason for a proof. Secondly, acknowledging their understanding of the type of proof, and finally, gaining the skills required to construct proofs (Greenes, 2003).

2.3.3 Visualisation and proofs

The ancient Greek mathematicians used proofs to demonstrate a mathematical understanding in a purely graphic method (Miller, 2012). Piaget (1936) and Van Hiele's (1986) theories of learning geometry, affirming the hierarchical stages that justifies the knowledge in mathematics and describes the stages that a learner go through in learning it. According to Van Hiele's (1986) a learner uses different stages including visualisation by children to arrive at a solution. Comparing the theorem to the proof of for example, the circle theorem, using a dynamic tool can facilitate mathematical understanding through the visualisation function that makes it possible by dragging the points on the drawn shape to align with the desired knowledge (Rodd, 2010). The act of visualisation in geometrical reasoning to represent proofs as an absolute truth is subject to debate and discussion (Giaquinto, 2007). Although visualisation offers ways of thinking about investigations in geometry, the process involves the use of heavy effects in visual reasoning that can be ambiguous for learners. Again, an individual's mood psychologically suggests that visualisation employs effects in different ways, including skills in number and algebra, difficult concepts for most learners to establish their relations with geometry. The 'positive affect' here is that when a learner is receptive they can broaden their perspective and so facilitate creative problem solving. The converse is true when the 'negative affect' contributes to a person attending to detail and so facilitates on difficult computations and indulge in efficient task accuracies (Linnenbrink & Pintrich, 2004).

2.3.4 Theorems and proofs

According to Rene Thom (1971, p. 697), 'A theorem is above all the object of a vision'. In Thom's aphorism, he defines a theorem as 'that which is true' and understanding/knowledge as the ability to be translated into certainty ('object') and revelation ('vision'). The nature of truth associated with theorems can be visually expressed (Davis, 1993). Despite the claim by many mathematicians that theorems are often viewed as the ultimate truth of a key concept in mathematics education, Thom asserts that the knowledge of a theorem can be vague and capricious.

Mathematical proofs are usually offered in a progression method, moving from one stage to the next, unidirectional from propositions to conclusion (Leron, 1983). Although this approach secures the required validity of proofs, it is, on the other hand, inappropriate when the proof involves ambiguous mathematical communication. Mathematicians in arriving at a proof have used various confusing approaches involving a process of justification and modification of ideas (Liljedahl, 2007). Communication forms a vital part in the process of any mathematical proof. In proofing, the mathematician 'provides the reasoning and the rational underpinnings for why the mathematics behind the whole process is not only valid but also worthy of discussion' (Liljedahl, 2007).

Many writers, including Liljedahl (2007), have indicated that the problems of learners associated with proofs must not be attributed to slow cognitive development or lack of logical reasoning power, but the inability to appreciate the need for proving theorems. According to Liljedahl (2007), proving for instance the Pythagoras' theorem is a cognitive development of mathematical concepts.

2.3.5 Inductive reasoning

All students have the potential to participate in some form of mathematical reasoning. The nature of mathematical reasoning involved in mathematical proofs is largely categorised under either inductive or deductive reasoning. Young learners, according to Greenes (2003), indulge in inductive reasoning, which involves the ability for learners to make generalisations by looking for specific patterns. Such reasoning allows students to begin solving problems, for example through drawing pictures and using accustomed methods to make conjectures (Greenes, 2003). However, as pointed out by *Professional Standards for Teaching Mathematics* (NCTM) in 1991, although inductive reasoning aids problem solvers to develop an understanding of a problem, it may be inadequate for learners to arrive at the correct and expected solution. There is the need to verify simplifications acquired over inductive reasoning proven through deductive reasoning.

2.3.6 Deductive reasoning

Inductive reasoning depends on observing patterns and the ability to make conjectures. However, in sharp contrast, deductive reasoning depends on logic. Using proofs to solve mathematical problems enables learners to make a logical argument by drawing on logical conclusions, through generalisations based on logic applied to specific situations. For example, the proof of the Pythagorean Theorem by students may consist of an explanation through a deductive process (Greenes, 2003).

An update of the *Professional Standards for Teaching Mathematics* (NCTM, 2000) advocates teachers to use proofs to advance the mathematical reasoning of students. Using mathematical proofs to aid mathematics understanding offers learner's opportunities for developing their thinking and learning. Proofs that require students to execute a memorised procedure improve their chances of operating at a higher level of mathematical thinking, especially where the task engages learners to think actively about the concepts in the proof and make connections among them (Greenes, 2003). According to (Doyle, 1988), the tasks that students encounter through proving a mathematical theorem build their experiences and their ability to personally engage in mathematics pedagogy.

Again, the illustration demonstrated in the 1999 video study accomplished by the Third International Mathematics and Science Study (TIMSS), revealed a similar trend in the relationship between low-level tasks, where learners demonstrate mathematical procedures consisting of basic computations, and high-level tasks, which entail the making of connections centred on discovering and concepts (NCES, 2003; Stiegler & Hiebert, 2004). In an assertion, the report concluded that the tasks that promoted high-level cognitive demands on students had a maximum prospective to develop learners' ability to reason and conjecture mathematically and the ability to comprehend what defines a mathematical proof.

2.4 The benefit of ICT in education

The benefit of ICT as widely used, encompasses the view that ICT refers not only to 'computers and computing related activities' but all technological activities. The shift in definition towards the end of the 1980s from 'computers' to 'IT' represented not only computer technology, but also the integration of all related technology with the capacity to store and retrieve information (Pelgrum & Law, 2003). In this study, the term ICT is used in accordance with the United Nations report (1999) description of ICT as consisting of telecommunications equipment, information technology equipment and services, information and communication activities, internet services, and network-based information services.

2.4.1 Excitement and confidence

Children's excitement when learning with computers allows them to devote more attention to understanding key process (Ruthven & Hennessy, 2003). Integrating ICT in learning brings a new dimension to teaching and learning, where learners go to the computer environment with the expectancy of doing something new. The use of ICT brings variety to classroom learning, which enhances learning appeal and promotes different learning styles (Oldknow & Taylor, 2003; Ruthven & Hennessy, 2003). According to Ruthven and Hennessy (2003), when the different leaning needs are accounted for while ICT lessons are introduced increases pupils' motivational to learn new ideas. The use of ICT has the additional benefit of building confidence and persistence among pupils. The new curiosity makes learners more active and encourages them to investigate and explore new knowlwdge, so mathematical software can be relied on as a tool that improves topical understanding in mathematics, ensuring more enthusiasm to work (Ruthven et al., 2008). Again Ramsey et al., (1991) cites that students can enhance their own knowledge through enthusiasm and not remain passive recipients of information when ICT is integrated through a student-centred approach to learning.

2.4.2 Aid lesson planning with wide range of resources

The use of ICT aids teachers planning of lesson in numerous ways. According to Tinio (2002) through the use of ICT teachers gain access to online materials saving precious time and cost for creating new resources. The use of ICT enables teachers to either use or improve on either own or another teachers resources. Through the use of ICT, teachers can easily create a reservoir of variety of resources to be used by other teachers on similar topics of interest. The ability for teachers to plan to use a wide range of resources in lessons helps to accommodate for the different learning needs in the classroom when ICT is used. Through the use of the internet, teachers are able to provide learners the learning experiences beyond the classroom environment, all over the word to enhance learning (World Bank report, 2004).

The use of technology as indicated by Wong et al., (2006) is vital to support direct teaching and learning in the classroom, possible through audio and visual capabilities. The use of ICT has the powerful benefit of enabling teachers to plan different learning scenarios favourable for both collaborative and independent learning (Cabero, 2001). These learning styles are possible due to the ability of teachers to promote reception of information and interactions through the use of ICT. The use of ICT therefore, accelerates the nature of lesson delivery in

mathematics education since learners can gain access to wide range of knowledgeable resources from anywhere with no geographical constraints. This benefit of using ICT to prepare these lessons resources in this approach aids the way lessons are prepared and taught, the learning is more productive, efficient and student oriented.

2.4.3 Improved the attainment of children

Many European countries, including Sweden, continue to stress the importance of integrating ICT pedagogy to enhance the quality of teaching (Abbott, 2001). Vygotsky's theory, originally written in Russian in the late 1930s, placed emphasis on the learning gained by pupils through ICT pedagogy. Where a child's learning is hampered with misunderstanding, Vygotsky (1978) emphasised the benefit of using ICT to provide guarded questions to progress the learning improved the attainment of children. Vygotsky (1978) acknowledged that children's learning of social rules and culture is enhanced by ICT-integrated learning, beneficial for educational pedegogy. Ekanayake and Wishart (2015) also argued that introducing students to ICT tools was important to help them to understand the corporate environment. Finally, the use of ICT and new technologies has increased educational attainment through independent learning (Ertmer et al., 2012). In the view of Robertson and and Al-Zahrani (2012) and Young (2014), the increase attainment is achieved because students develop more self-learning abilities through the use of ICT.

Over the last decade, digital technologies (DT) have supported the pedagogy of mathematics in schools. The impact of technology on students' mathematical thinking cannot be over exaggerated, and the debate on the effect of ICT on students' thinking and reasoning in mathematics has gained enormous attention. Olive and Makar (2010), who examined the effect of ICT on students' learning, reiterate the need to incorporate ICT to promote the efficiency of learning mathematics. According to Goos (2010), the knowledge gains in learning mathematics when digital tools are used generally outweigh the benefits of learning through pen and paper. Again Goos indicates that by using ICT, it is possible to make predictions and test conjectures when completing a mathematical task (Goos, 2010). Conferring with Zbiek et al., (2007) and Heid and Blume (2008), promoting ICT pedagogy, including the use of GeoGebra, has the possibility to support mathematical thinking to deepen students' mathematical content knowledge.

2.4.4 Enhance interactivity and exploration

Ager further claims that ICT is important to motivate and provide interactive programmes to support children's learning (Ager, 2000). The theory of Multiple Intelligences propounded by Gardner also suggests that learning takes place in variety of ways when using ICT (Gardner, 1993). In support of Vygotsky's theory, Gardner suggests that the development of different types of intelligence is possible through ICT. The ICT Test Bed Evaluation conducted in 2006 provides again useful evidence that demonstrate that many teachers opted for ICT to support innovative pedagogy to enhance on the interactivity of learners (Becta, 2006).

The use of ICT tools significantly broadens the learning opportunities for most learners (Ekanayake & Wishart, 2015). Using ICT improves the quality of education (Al-Ansari, 2006), and can accelerate, innovate, and enrich students' learning (Davis & Tearle, 1999; Lemke & Coughlin, 1998, cited by Yusuf, 2005). The visual and interactive tools in computers can make learning exciting for users (Phillips & Scrimshaw, 1997), as they facilitate rapid feedback during discussions and offer sets of challenges to enhance learning (Mann & Tall, 1992).

2.5 Barriers to integration of ICT in pedagogy

The study by TIMSS (2007), reports about the lack of the use of computers in mathematics lessons in countries where the use of computers is relatively high (Mullis et al., 2009). A further report done in the UK by ImpaCT2 in 2003 on students in Key Stage 3 maths lessons stated that 67% of Key Stage 3 students hardly used ICT. Many teachers continue to resist the use of ICT to promote mathematical pedagogy in lessons. While this assertion is true in most cases, the use of ICT in lessons remains problematic. Furthermore, using ICT tools does not necessarily increase attainment (Ellis & Loveless, 2001).

Attainment is not obvious in all National Curriculum subjects when ICT was integrated, especially in Mathematics, Science, and English (Becta, 2004). However, the claim of attainment is seen to be the highest in primary schools rather than secondary schools is absurd (Becta, 2004). Therefore, the claim of the impact on attainment through ICT is 'inconsistent and restricted by the amount of ICT use' (Becta, 2004:5). Kennewell (2000) echoes the view of Joy and Garcia (2000), who argue that learning gains are not achieved

solely as a result of ICT integration, but through the amalgamation of ICT with specific pedagogical practices.

The following were among the identified barriers to the use of ICT in mathematics education.

2.5.1 Lack of school and curriculum resources

Kuchler (1998) conducted a study to determine the challenges faced by teachers who employ technological instructional resources in mathematics. The study established that some of these obstacles included the lack of adequate time to set and prepare for a lesson as well as the lack of differentiated resources. Added to this problem is the lack of adequate resources to integrate ICT, as a considerable amount of teachers' time is often spent on finding appropriate resources (Cox, 1993; Johnston-Wilder, 2005).

Despite the benefits of integrated ICT pedagogy, there are many who oppose the introduction of the idea in the school curriculum. In UK, it is widely known that the introduction of the national curriculum in 1988 did not encourage the integration of ICT, as the subjects in the national curriculum remain very traditional (Abbott, 2001).

The initial lack of interest by Her Majesty's Inspectors (HMI) at the early stages of ICT integration into school curriculums reduced many schools' motivation to integrate ICT pedagogy (Abbott, 2001). Graham (1993) agreeing with Somekh (2007) states that the restrictions of the curriculum resources greatly limit the potential for ICT tools to radically transform the learning that takes place in schools.

Abbott (2001) similarly laments that the current school curriculum of some subjects makes it difficult to embed ICT. In addition to the difficulty of integrating ICT into the curriculum, there is the debate as to whether ICT should be taught as a standalone subject rather than through integration (Graham, 1993). Abbott (2001) argues against the approach of teaching ICT as a separate subject and advocates for integration. Despite the importance of 'purposeful assessment of attainment', an integrated ICT curriculum does not easily lend itself to assessment (Kennewell et al., 2000: p.3). Major public examinations are still tested traditionally and do not test pupils' skills acquired through ICT (Kennewell et al., 2000). According to Gillespie (2006), computer-based assessment resources take considerable time, expertise and money to prepare hence there is the reluctance on the part of teachers to promote the practice.

2.5.2 Software and Hardware problems

Kreijns et al., (2013) argue that the lack of hardware support and poor access to technical support present barriers to the successful integration of ICT tools. According to Buabeng-Andoh (2012), eighty percent of teachers' resistance to change and negative attitudes to ICT is due to poor software use. Worldwide surveys in secondary schools also show that the majority of teachers perceive ICT software as the greatest obstacle to ICT integration (Leask & Pachler, 2013). The absence of technical support for software to integrate computer tools successfully is often due to schools' financial constraints, and this further discourages teachers from using ICT (Cox, 1993). Gillespie (2006) argues that where software exists, the over-dependence of the applications has led to slower and unreliable access to internet connections. Often schools' software applications are outdated with no proper maintenance programme in place. The substantial cost of ensuring that children are safe with software online also puts greater pressure on a school's financial budget, as it requires expensive software with continuous technological support in cyberspace (Gillespie, 2006). Most teachers are currently struggling with new software applications due to the high pace at which modern technologies are changing (Barak, 2007; Koehler & Mishra, 2009). Koehler and Mishra (2009) and Barak (2007) argue that the inability of teachers to learn and familiarise themselves with newer digital technologies and computer hardware that keep changing makes it difficult for teachers to keep up with the expected ICT competencies.

2.5.3 Lack of experience with ICT tools

School management teams face considerable difficulties in embarking on school policies for ICT-integrated lessons (Koh & Chai, 2014). A survey conducted by Becta in 2004 revealed that integrating ICT tools into the pedagogy of secondary teachers was becoming difficult due to lack of experience. Mama and Hennessy (2013) also confirm the assertion that the incompetence of teachers is directly linked to teachers' lack of experience. According to Sorensen et al., (2007) the lack of confidence for teachers to use ICT is due to lack of experience. Also, impacting on the experience of teacher's is the limited knowledge of the application of the software, which acts as a hindrance in the use of ICT in the classroom (Koehler & Mishra, 2009). Where hardware devices are used to enhance learning, the inefficient knowledge on how to utilise the software by teachers can impede its integration Clark-Wilson et al., 2014). The inability of teachers to gain the needed experience to deliver an ICT based lessons are associated with a teacher's general interest in ICT. Although many

factors affects a teachers experience, among them are the amount of time a teacher allocates to the learning of new ICT tools.

2.5.4 Negative attitudes

The Stevenson Report (1997) in UK identified that due to the resistance towards ICT, its use in schools was 'described as patchy' (Leask, 2001:p.62). Teachers' resistance to educational technologies has been noted for many years, and is evident in their attitudes (Woodrow, 1990). The poor attitudes and approaches to using ICT to facilitate pedagogy are due to lack of 'preparedness to exploit new technologies' (Ellis & Loveless, 2001:p.210). This means that teacher's poor attitude towards ICT-related pedagogies and practices can influence approaches towards ICT usage (Mumtaz, 2000; Watson, 1993). Arguably, changing such attitudes towards practice entails considerable difficulties, largely due to the lack of tailored support, and requires understanding of the experiences of teachers (Ellis & Loveless, 2001).

Negative attitude is one of the known factors affecting ICT integration in schools. Here, 'teachers' Scepticism, lack of confidence and competence with ICT remain centrally important in the pedagogical adoption of ICT' (Somekh, 2008:p.209). Although most studies refute that the attitude of teachers determines how well ICT is deployed in lessons, it continues to be a contentious debate since the use of ICT is 'valuable and well-functioning instructional tool' in education (Becker, 2000, p.29). Individual opinions concerning the use of ICT are often less assumed and remain a difficulty to resolve (Becker, 2000). The changes of attitude required for ICT integration are difficult to instigate, and knowing how to facilitate and support these changes are major concern to institutions (Garet et al., 2001).

Despite the advantages of using ICT in schools to raise attainment, little evidence is known about the relationship that exists between teachers' attitude and the full integration of ICT into schools' curricula. Schiller (2002) affirms that teachers have an important role to play in promoting and supporting the integration of ICT. By understanding the factors that affect the attitudes of teachers, it is possible to bring about the educational changes that are needed in the modern age (Mingaine, 2013). According to Mingaine (2013), teachers' attitude of ICT is important in supporting staff efficiency in relation to integrated learning in schools. Furthermore, He indicated that the understanding of educator's views will increase the understanding of education reforms that supports teachers in their pedagogy.

2.5.5 Lack of training

It was established that teachers lack the professional expertise required for the employment of ICT tools and resources. This is coupled with the shortage of ICT-trained specialists in schools (Cox, 1993). Many critics blame the teacher training for not equipping teachers with the right ICT skills to teach (Cox et al., 1988). In addition, Fu (2013) and Mumtaz (2000) found that lack of training time hinders the use of ICT in schools. The absence of the time to experiment with new skills through ICT initiatives makes it difficult to incorporate ICT applications in classrooms (Fu, 2013; Johnston-Wilder, 2005). According to Gillespie (2006), the prioritisation of schools' initiatives on improving pedagogy over staff professional development has led to less time for experimenting with new ideas. Sorensen et al. (2007) agree with both Koehler and Mishra (2009) and Barak (2007) in indicating that the absence of skilful knowledge in using ICT in lessons discourages teachers to use the software. Generally, the inability of teachers to acquire the needed skills through training or professional development to embed ICT into lessons is blamed on the absence of school clear policies and leadership (Koehler and Mishra, 2009).

2.6 ICT facilitators in education

The reviews reported on many initiatives that teachers perceived to be the main ICT integration facilitators. These supports aim at either removing the obstacles or to encourage teachers to use numerous ICT tools to promote their classroom pedagogy. The section discusses the following support; the availability of ICT tools, experience and knowledge of using ICT, school leadership and the opportunities for teacher training programmes.

2.6.1 The availability of ICT tools

Among the many facilitators of ICT is access to both hardware and software tools. According to Norris et al., (2003), suitable access to ICT tools is a key factor in the effective learning process. For example, interactive white boards facilitate 'visual education in schools' (Ellis & Loveless, 2001:p.210). The availability of digital resources such as digital libraries that makes it possible for teachers and students to access needed pedagogical materials facilities the use of ICT (Cholin, 2005). This availability of such tools when present in the school enhances on teachers confident of sharing good practice. However, many reviews argue that heavy reliance on ICT tools to facilitate learning could lead to the 'loss of hard-won fluent

skills' in education (Ellis and Loveless, 2001:175). In some schools where there are shortages of computer suits or long teacher waiting queues on the use of ICT devices, teachers are reluctant to plan and teach using ICT. In situations where internet connections are required for a lesson, often the broad band connections to online resources take ages to download. Such situations often leave both teachers and students frustrated by the long wait for the network connection, hence very easy for teachers to abandon such lessons.

2.6.2 Experience and knowledge of using ICT

The experience of teachers on the use of ICT greatly affects their use in the classroom. The experience of a teacher in the use of ICT acts as a catalyst for change. Experiences acquired of the classroom are by its nature an ingredient to make teachers engrossed in the use of ICT. According to Plomp et al., (2007) teachers who have had the experience of using ICT are by themselves innovators; hence such prior experiences can motivate the use of ICT in practice. An experience user of ICT tools facilities a teacher's creativity through exploration in practice. Notably, ICT is unlikely to be used in practice if teachers have less experience in the required skills. To support the use of ICT pedagogy opportunities should be offered to teachers to either gain deeper knowledge or shaper practical experiences. The case studies conducted by Harris (2002) on three secondary and three primary schools focussing on the benefit of inventive educational practices in ICT, indicated that experienced teachers explores new avenues towards classroom pedagogy. Teachers will feel hesitant to advocate for an ICT integrated lesson where poor experiences have occurred in the past. The use of support staff to assist personnel's to overcome poor classroom experiences can motivate teachers to use ICT.

2.6.3 School leadership

Research by Hammersley-Fletcher and Brundrett (2005) indicates that professional development plays a role in enhancing institutional changes in teachers' pedagogy. A key subject leadership role is to provide 'investment in ongoing professional development, for teachers' (Fullan, 2006:p.4). Glatter (2006) blames institutions for neglecting the challenging needs of teachers by refusing to acknowledge their training needs to be provided by the school leadership. Even though Barker et al. (2010) believe that the desired training of staff enhances their responsibility, the absence of the needed resources to provide the training remains a concern. School leaders must provide the ethos and culture in which ICT

integration can thrive. The refusal to identify the prevailing teachers ICT needs, according to Fullan (2006) and Tondeur et al., (2008) is the inability of school leaders to promote a shift in institutional culture. It is also worth noting that when school leaders provide teachers with right and appropriate training, the positive attitudes towards the use of ICT are encouraged (Dupagne & Krendl, 1992).

However, there is a dilemma regarding whose responsibility it is to provide the needed training to facilitate changes, with the failure of institutional training blamed on schools' Human Resources (HMR). Facilitating ICT changes requires creating 'conditions in which all members of the organisation can give their best' (Hallet, 2014:p.17). According to Morgeson et. al., (2010) and Merkley et al., (1997), facilitation is achieved through coaching and mentoring. Using the self-efficacy model, leaders facilitate changes by promoting confidence and self-esteem (Martin-Alcazar et al., 2005; Truss et al., 1997). The ability of leaders to promote opportunities for teachers to 'share good practice' builds their confidence in using ICT (Gillespie, 2006:p.73).

2.6.4 Opportunities for training

At the start of the 21st century, a UK government initiative emphasised the important role of ICT in children's education. This initiative was based on the principle that ICT was vital in education, and insisted that all teachers be trained to use the pedagogy (Stevenson, 1997). A separate survey by DfEE (1998) indicated a strong commitment by the government to support the use of ICT-integrated lessons: 'The government is fully committed to ensuring that all schools and teachers are in a position to deploy new information and communication technology (ICT) to raise educational standards' (DfEE, 1998:p.3).

According to Mann and Tall (1992), providing training can facilitate the use of ICT tools. While having an ICT specialist in the school to assist training is vital, training can also be done through teams, as they provide the 'right opportunity' to facilitate teachers to use ICT (Woodcock 1985:p.87). Working in teams has the benefit of facilitating ICT integration because 'teams outperform individuals acting alone' (Katzenbach & Smith cited in Bush & Middlewood, 2005:p.107). However, developing teams to facilitate any initiatives requires 'time as the team develops and the environment evolves' (Morgeson et. al., 2010:p.12).

Teachers' pedagogical perceptions are tied to their professional attitude, and their perceptions change over time through continuous professional development (Wood & Bennett, 2000).

Although the training done by internal specialists is deemed to be more effective as they are familiar with the school's curriculum and can offer role models for teachers, there is ample research to show that their services do not meet the real needs of teachers (Crook, 1994; Sandoltz et al., 1997).

2.7 ICT and mathematics education

Paul Drijvers (2013), in his work *Digital Technology in Mathematics Education: Why it works (or doesn't)*, noted that digital technology integration confronts the teachers, researchers and educators with a diversified range of questions. What is the ICT potential in teaching and learning mathematics? Which are the conclusive factors in enabling its application in mathematics education? According to Artigue (2002) the failure of teachers to use the required mathematical software applications can limit students' understanding of mathematics.

2.7.1 Constructive reasoning

The importance of incorporating digital technologies to bring about the needed pedagogical and curriculum changes is emphasised in the research carried by McFarlane and Sakellariou (2002) and Rogers (2004). According to the research, students' constructive reasoning has the ability to beneficially develop when technology is incorporated into the learning. Using ICT increases students' constructive reasoning, leading to higher achievement in mathematics (Hanna & Jahnke, 1996). Again, the use of ICT offers the pupils the ability to model and repeat the key process many times to facilitate understanding of mathematical processes. The manipulations performed through ICT are reliable and generally produce accurate results when used for computations (Ruthven & Hennessy, 2003). Oldknow and Taylor (2003), in support of Hanna and Jahnke (1996), argue that pupils' mathematical reasoning is enhanced when different representations of key concepts are seen for the same concept multiple times.

2.7.2 Conjectures and efficient learning

Using digital technologies improves students' performance and advances conjectures (Murray, 1996). According to Murray (2016), to promote conjecture learning, there is the necessity to improve the teaching of mathematics through alternative pedagogical approaches that integrate technology. Since the process of conjecturing to gain understanding depends on time and attention, using technology for learning increases the attention that learners devote

to the learning process (Murray, 1996). As Willougby and Wood (2008) rightly point out, learners learn without actually realising the amount of attention devoted to understanding key concepts when computer software is involved.

The use of digital tools increases the efficiency of pupils' learning and improves the value of work that they produce (Ruthven & Hennessy, 2003), allowing the more routine classwork to be accomplished reliably and in a timely manner. The emphasis on speed, ease, and time saving enables much work to be done within a short space of time (Ruthven & Hennessy, 2003), and this improves students' learning involvement in mathematics (Jones, 2011). It is argued that using digital technologies improves speed and accuracy when accomplishing learning gaols, just as mentioned by (Ruthven & Hennessy, 2003; Sorensen et al., 2007; Pierce & Stacey, 2010). This is in contrast to the use of pen-and-paper learning that often leads to many errors and can be time consuming when calculations are involved. The use of digital technologies in geometry provides learners with the ability to interact with geometric objects through 'dragging', which allows them to investigate for example, the properties of quadrilaterals to gain understanding within a shorter time than using pen and paper (Ruthven & Hennessy, 2003)

2.7.3 Guaranteed feedback and interactive support

The use of ICT supports feedback on various processes, testing, and improvement (Ruthven & Hennessy, 2003) An investigation by Trifonas (2008) demonstrated that it is possible to raise attainment in mathematics when students are active with their learning using technological innovations, due to the guaranteed feedback on completed work. The ability to use ICT ensures that instructions on the computer are executed instantly empowering learners to appreciate the differences in learning objects and outcomes. Through the tool, effective comparisons can be made on the same platform to see the differences in the learning for instant feedback (Oldknow & Taylor, 2003; Ruthven & Hennessy, 2003). The ability of the software to produce ongoing feedback motivates children to remain focused on the task. The fast and reliable non-judgemental, impartial feedback provided by ICT encourages learners to form their own conjectures and ideas about the learning (Oldknow & Taylor, 2003). This is crucial for the cognitive development of learners (Baki et al., 2011).

Remarkably, software tools have the ability to provide interactive support for problem solving in mathematics through 'trial and improvement', enabling children to conjecture. The

facility to interact with the software provides the capacity for students to investigate and extends thinking time (Oldknow & Taylor, 2003; Ruthven & Hennessy, 2003). According to Ruthven and Hennessy (2003), learners will go the extra mile with their learning when technology is incorporated through interactivity to gain understanding of key concepts. This is possible because by going through more than one procedure, students discover the repeated patterns, and more demanding task is solved quickly (Ruthven & Hennessy, 2003). This use of technology agrees with a move to focus learners' mathematical thinking achieved through instant feedback and promote activities away from just measuring and drawing towards designs investigation and conjecture testing (Sherman, 2014).

2.8 GeoGebra as a digital tool in mathematics education

According to the Key Curriculum Press (2003), the most widely used ICT tool in mathematics lessons is the Geometer's Sketchpad. Even though the Geometer's Sketchpad was the preferred software in mathematics amongst educators (Becker et al., 1999), the use of GeoGebra has the ability to develop inventive activities and geometrical thinking, which makes it an excellent choice by most practitioners (Venkataraman, 2012). At a time when the technology is constantly evolving, it is critical for teachers to incorporate the different dynamics in content delivery, most especially the graphing software.

Currently, GeoGebra (derived from Algebra and Geometry) is one among the highest innovations, open-code ICT tool that is freely downloadable and functions on a diverse spectrum in the system platforms (Ruthven et al., 2008; Jones el al., 2009).

GeoGebra has in the last decade gained popularity among researchers and teachers globally, since it is easy to use and effective in the dynamic mathematics software, which integrates numerous aspects of various mathematical packages. Additionally, it has enhanced the development of an extensive user community owing to its open-source nature.

2.8.1 Merits of using GeoGebra

The open-code maths software offers algebra, geometry and characteristics of calculus in connected environments. In addition, the software offers extensive dynamic geometry concepts in mathematical analysis. The use of GeoGebra has the ability to bring meaning to abstract concepts, through visualisation. GeoGebra aids students in grasping abstract problem-oriented and experimental learning of mathematics. The students can consequently
utilise a system of computer algebra in interactive geometric systems, which helps in maximising the cognitive abilities in the best possible ways.

When compared to graph calculators such as the MathPapa, the GeoGebra software is more user-friendly and more efficient in offering compatible interface, commands and multilingual menus. This feature in learning mathematics through GeoGebra reinforces the principle of being learner-centred as well as experiential learning (Kolb, 2015). Additionally, the use of GeoGebra motivates the students' mathematical concepts through guided learning discoveries and a wide scope of presentations (Bester & Brand, 2013). According to Willoughby and Wood (2008), the use of GeoGebra offers learners the capability to go the extra mile of learning without being aware of the time actually spent on acquiring the skills.

The use of sliders and dragging in GeoGebra allows learners to use visualisation to understand key concepts through multiple views (Venkataraman, 2012). For example, students can generate adjustments via manipulation of free objects in different views. The use of GeoGebra provides input of algebra enables the users in generating new objects or modifying already existing ones. Teachers use GeoGebra to stimulate learner's creativity through investigations (Brennan & Resnick, 2012).

2.8.2 Demerits of GeoGebra

Students lacking previous experience in programming will find it difficult to use the input bit with the algebraic commands. Though the simplified commands are not hard to understand, the students will find it confusing. Secondly, the existing research on the use of GeoGebra reports limited research on the impacts on the true achievements in mathematics. Cheung and Slavin (2013) cites that the use of ICT tools including GeoGebra has minimum significance on learner's achievement. This perhaps suggests that to impact learner's achievements other methods of teaching should be considered.

Also although when GeoGebra is used, the application of multiple views enhances learners understanding in mathematics through visualisation, there are limited research on whether the use of the software alone has the ability to accelerate learner's creative reasoning associated with algebra, arithmetic, statistics and analysis (Olsson, 2019). In a realistic perspective, the software possesses in-built support in animations but this is limited in scope. Thus, inclusivity of the modules for animating in GeoGebra has turned out to be a critical technical aspect in any futuristic versions.

2.9 Technological Pedagogical and Content knowledge (TPACK)

The TPACK frame work is used in many studies by teachers to plan and analyse the effective use of ICT in lessons. The use of TPACK allows the combination of factors that a teacher considers enhancing in students' knowledge in learning when using ICT (Schmidt et al., 2009; Mishra & Koehler, 2006). Koehler and Mishra (2009) classify knowledge in three fundamental forms when ICT is used in lessons: Technological Knowledge (TK), Pedagogical Knowledge (PK), and Content Knowledge (CK). Since teachers are very often faced with lots of complexities in technology integration (Monaghan 2004), it is critical to adapt the framework to plan efficiently. The use of TPACK framework aids to conceptualise the teacher knowledge required in integrating ICT in the classrooms (Brennan & Resnick, 2012).

Below is an illustration of the TPACK framework:



Figure 1: TPACK framework

The use of TPACK aids the teacher to plan for both the integrated and discrete knowledge in the selection of ICT curricular resources. To be more specific, the process of choosing and the using of resources requires a hint of the relevant sources; skills in effectively using the sources; and an idea of the features of the relevant resources. Additionally, knowledge of the specific hardware and software is required in enabling the teachers to comprehend the options from which they could select and efficiently use in lessons (Clark-Wilson et al., 2014).

In the circle in figure 1, the areas that overlap are three kinds of knowledge combination, which can be explained as follows:

- Pedagogical content knowledge (PCK): this defines the knowledge that teachers possess concerning their content (Shulman, 1986)
- Technological pedagogical knowledge (TPK): this is a skill set that enable teachers to identify the best technology in supporting a specific educational approach (Mishra & Kohler, 2006).
- Technological content knowledge (TCK): this denotes the skills that were acknowledged in which the teachers recognise the finest set of skills in using technology to support students to learn the subject content (Mishra & Kohler, 2006).

Our main area of interest will be on the content knowledge, since most of our discussion on integration of the dynamic graphing software will be firmly established from content knowledge.

2.9.1 Content knowledge (C)

Content knowledge is associated with the definite matter of interest that is to be learned or taught (Rowland et al., 2009). The question that comes to mind is how the various kinds of 'content' knowledge are applied by mathematics teachers alongside incorporation of the dynamic software in the teaching of a mathematical topic?

2.9.2 Pedagogical content knowledge (PCK)

Intuitively, it is with no argument to believe that a good teacher should have detailed subject knowledge. Many policy makers and research over time has described accurately what knowledge a teacher should have to be successful (Ball, 1991)? To answer this same question, Ball (1991) argue that inadequate insight in the understanding of the subject matter adds to poor numeracy skills and can contribute to a lack of curiosity in teaching maths. The perception of learning mathematics is further reinforced through the heavy dependence on textbooks listing 'hints' rather than conceptual development. No wonder many studies on this subject still consider what constitutes a reputable teacher to be implicit and reflect teaching as a vocation based on the application of common sense with less involvement of professional education.

Schulman (1986) theorised pedagogical content knowledge (PCK) as an exclusive realm of teacher knowledge including numerous facets of mathematical knowledge relevant for teaching. Regarding PCK, he proposed to justify the principal role of subject content

knowledge in the quality of teaching together with pedagogical knowledge. In Shulman's (1986) perception, PCK is associated with both having the knowledge of the subject content and the ability to support others to understand it. Krauss et al., (2008), by refining the views on Shulman perception of PCK, identified three key scopes of PCK: the teacher's mathematical knowledge; the interpretation of students' knowledge and thinking; and knowledge of numerous intelligences on mathematical problems. Ball (1991), in their definition, also takes account of the subject knowledge, teachers view comprising the structure of the curriculum and the gradual introduction of concepts.

Eisenhart et al., (1993) did not find Shulman's PCK concept supportive and dismisses the difference between pedagogical content knowledge and subject. Eisenhart et al., (1993) uses conceptual knowledge and procedure as constituents of teacher knowledge. Cochran et al., (1993) use pedagogical content knowing (PCK) to emphasise on PCK, an inherent knowledge associated with a specified teaching context. Shulman's thought was difficult to justify with empirical data and is again seen as a theoretical concept. To overcome the difficulties with Shulman's constructs, Rowland et al., (2005) introduced the theory to advance an empirically founded taxonomy, the 'Knowledge Quartet', which differentiates mathematical contents into foundation, transformation, connection and contingency.

2.9.3 Limitation of TPACK

The first supposition of the use of TPACK is that technology, pedagogical and content knowledge happen in segregation and can be assumed through separate overlapping circles as a symbol of empirical realism. This further assumes that technological, content and pedagogical knowledge happen in a distinct process. However, the content expressed in the form of learning is impossible in the absence of pedagogy and the ability to present explanations of key concepts is naturally represented in pedagogy (Archambault & Barnett, 2010). This makes Shulman's framework seem unreasonably complicated when the common association that exists among content knowledge structure and the equivalent pedagogical knowledge are accommodated in a single frame (Archambault & Barnett, 2010).

Added to this is the second assumption of TPACK on the content knowledge (CK). According to Shulman, the authority of content knowledge depicts one of the key constituents of digital education and must be viewed as one of the key concerns of education. This held view is not justified and expounded with convincing reasons. According to Segall (2004), the

act of recent education is rather alarmed on the attainment of capabilities, individual resources and development skills, and not content as such, although content is defined by key concepts and cannot easily be disconnected from informal epistemological admittance.

Another limitation of TPACK was the assumption of an ideal absent of actors and goals in mind when the framework is in practice. This final assumption perceived that it is confusing by distinctively locating boundaries around T, P and CK as clear regions when planning teaching and learning activities with a digital media learning environment (McEwan & Bull, 1991; Archambault & Barnett, 2010). The assumption that key elements of learning procedures are also separated from, for example, key educators including teachers makes TPACK a less student-centred model, which is a serious concern for educators to ponder on when deciding whether to use it.

2.10 Chapter Summary

This section discussed the reasons for education in schools in relation to the mathematics education curriculum in UK. To understand the impact of ICT in education, we discussed the role of ICT in education in particular in mathematics education. The chapter was concluded with the review on the use of ICT tool GeoGebra, as digital tool to teach mathematics together with an identified Framework for ICT lessons.

CHAPTER 3 RESEARCH METHODOLOGY

3.1 Introduction

This chapter identifies the research method employed for this study including the type of data that was gathered and how it was analysed to generate the results for the study. This section begins with the justification of the rationale and purpose of the chosen methodology. The research approach is further discussed together with the chosen design. Also mentioned here is the examination of the data together with the selected ethics for the research. The research employed mixed methodology approach; quantitative and qualitative research designs.

The following research questions guide this study:

- What are the benefits of ICT in Mathematics education?
- What are the barriers to ICT in mathematics education?
- What factors will facilitate ICT in Mathematics education?

3.2. Research design

The descriptive enquiry design dominates this study since it defines the characteristics of the respondents. The method is suitable because it allows the researcher the benefit of conducting a detailed analysis as well as appreciates the philosophies of the participants using their own experiences and words (Bernard & Bernard, 2012). This method allows the researcher to make deductions dependent on their own opinions of the practice (Rubin & Babbie, 2012). In sharp contrast with the design that is exploratory, the researcher discovers new ideas in the study and connects the origin and the outcome of these issues (Bernard and Bernard, 2012). When descriptive design is used, the researcher is able to further study the research question before establishing any conclusions (Kreijns et al., 2013).

Two aspects guided the research design employed and these are the use of TPACK and Knowledge Quartet. These two frameworks have been explained and incorporated into the research design as follows;

3.3 Lesson Design: TPACK and KQ

The TPACK framework guided the design in aiding to express the knowledge content as well as the application of the technology. The PCK (pedagogical content knowledge) framework offers the teacher the opportunity to use the subject knowledge in designing the lesson. This was achieved by consulting the schemes of work to understand the key concepts to be presented in the lesson. Furthermore, there was the need to conduct an initial audit on the use of the technology within the subject knowledge before introducing new concepts. Finally, a consideration was given to how to explain the subject content to empower learners to acquire the needed mathematical knowledge.

Another aspect of TPACK framework that was deployed was the technological pedagogy knowledge (TPK). The knowledge of TPK reminded me to think clearly on own actions as a teacher and communicator of information to learners. Teachers in this position are at the forefront of correcting errors in teaching and education practice (Hadjidemetriou & Williams, 2002). To support both individual and whole class discussion learning, I embedded in my lessons questions to promote learners' thinking skills, vital to extend learners thought processes (Malibar & Pountney, 2002; Ruthven & Hennessy, 2003).

In planning for PK and PCK, students were equipped to carefully use a correct shared dialogue. Throughout the planning and the delivery of the lesson, opportunities were offered to students to discuss their work with other students and receive feedback on their completed work. Recognising the diverse needs of each learner enabled me to physically approach students to obtain feedback on the completed task. Much learning takes place when immediate assistance is at hand whilst completing a mathematical task (Saudelli & Ciampa, 2016). The designed worksheet was differentiated by task to ensure that each student benefited from the lesson. Avoiding generalisations and accommodating for individual learners' needs encourages learners to thrive in group studies (Mavrikis et al., 2012). Specific questions such as 'Why' and 'what', open-ended questions, were utilised to prompt class discussions and debate about the completed work.

3.4 Design tool

I used GeoGebra as a device to improve the delivery of my pedagogy; I acquainted myself with the expectation of the functions to be able to support learning. Also, I ensured that the

software was running on all the computers before the lesson. This knowledge and awareness of ensuring that my technological understanding of how to use GeoGebra to teach was expressed in all the lessons, a vital knowledge to ensure an effective technological integrated learning (Stupel & Ben-Chaim, 2017). Saudelli and Ciampa (2016) state that many unexpected challenges in using technology can be avoided when, during lesson planning, teachers anticipate and plan for them.

The knowledge of learners' background for John (2006) was vital for an effective lesson planning. The knowledge of students' backgrounds for my lesson planning included knowledge of their experiences and history of ICT used. I further consulted the school data on all the learners to be able to plan the different preferred learning styles in the class. In particular, the knowledge of students' previous digital use including the use of GeoGebra was considered for my planning (Godwin & Sutherland, 2004). The background information of most of the students indicated little use of ICT so the starter activities were focused on support tasks to ensure a scaffolding approach into learning. Added to this knowledge was the need to promote connectivity and collaborative learning. According to Gerlach (1994) learning takes place naturally when learners are offered the opportunity to discuss in groups.

3.5 Interpretivism

The research is structured within an interpretative paradigm. The context of study is within an educational setting. The interpretive paradigm is the most appropriate because the study aims to understand the perception of the participants (Willis, 1995). This approach stresses on knowledge socially acquired from participants (Sexton, 2003).

The reason why the method has been considered and used in this research is, firstly, because all the analysis and interpretations made in this study are primarily based on the researcher's personal experience and the perception of teachers through questionnaires with integrating ICT in lessons. Secondly, interviews were conducted to support data from the case studies and questionnaires. Also data was gathered from the literature to support the extracted primarily data.

The interpretive paradigm used in this research is reinforced by the interpretation attached to the collection of information on own practice to interpret it (Sexton, 2003). The interpretation

process consists of making meaning of the information obtained by drawing inferences and by establishing inferences between some abstract patterns (Aikenhead, 1996).

3.6 Research approach

First, primary data was collected as a means of understanding own practice with regards to digital integration in teaching circle theorems and linear functions. Thereafter, the data on questionnaires and interviews were gathered. Primary data has the ability to describe the events that really took place in the lesson (Wallace, 1991). The main sources of data collection for researchers on own practice, according to Richards and Lockart (1996), are lesson plans, reports, audio or video recordings, observations, and questionnaires on teachers' own practice. These sources allow a teacher to critically analyse their own practice in reference to subject literature. Using different episodes and case studies in the lesson allow the teacher to reflect on own practice with a view to improving (Richards & Farell, 2005).

According to Kvale (1996), research conducted within the context of the qualitative method uses interviews to understand world realities. The data collected through the interviews will enable me to make generalisations of the entire population of students in my class. Using the semi-structured interviews data and teachers will provide the researcher the information needed for the analysis (Brinkmann et al., 2015). Using interviews with has the added advantage of allowing the researcher the time to probe the participants in depth for the required information. However, the limitation here is that it requires plenty of time to conduct interviews and to analyse the transcribed data and it can be expensive and time consuming when a large amount of data is to be analysed (Halcomb & Davidson, 2006; Seidman, 2013).

3.7 Methodological approach

The research adopts both the explanatory and the descriptive research approaches under the qualitative and quantitative methodology. A mixture of these methods offers the investigator the ability to ascertain the phenomenon under study (Saunders, 2012). Burns and Grove (2003, p.19) describe a qualitative approach as 'a systematic subjective approach used to describe life experiences and situations to give them meaning'. A qualitative approach lends itself to a 'holistic description of events, procedures, and philosophies occurring in a natural setting needed to make accurate situational decisions' (Stainback & Stainback, 1988, p.88). It is because of this reason that the qualitative approach has been employed in this research; the

data gathered on case studies and used in this study is descriptive and is from the owner's experience. The subjective view on a participant is based on 'world view' with the aim to present a holistic picture of an individual's perceptions and interpretations on a concept (Yin, 2016).

Although both the mixed approaches were used, the quantitative was the dominant methodology. Choosing the qualitative research over the quantitative research offers the researcher the ability to deploy the perceptions of participants in relation to known theories (Conti, 2006). Also, by depending on multiple sources, the researcher obtains contextual information to make the results more significant and less subject to diverse interpretations (Fraenkel & Wallen, 2008). Furthermore, particularly, that researches accomplished through the quantitative paradigm do not give thought to subjective and social phenomena, compared to qualitative researches, where participants are studied in their usual environments, the mixed approached was considered (Conti, 2006). Qualitative approaches have the added advantage because 'the researcher can study dynamic processes (i.e. documenting sequential patterns and change)' (Burke et al., 2014, p.20). Furthermore, within the qualitative research paradigm, prevailing knowledge or perception has significance within a specified circumstance or context (Hadjidemetriou & Williams, 2002).

Although qualitative research permits in-depth examination of the participants, the validity of the research can be rejected (ibid.). According to Silverman (2006), the problem of reliability arises through definitions and the use of longer narratives. This problem arises perhaps mainly because of the way the 'researcher goes about categorizing the events or activities described ... many studies provide the readers with little more than brief, persuasive, data extracts' (Silverman, 2006, pp.44–47). The mixed approached was favoured because their benefits complements each other to produce a more object conclusion even though, the use of quantitative approach often involves heavy calculations which may be expensive and time consuming.

3.8 Data collection

Since no single data source has the ability to provide the researcher with the entire accurate and whole picture of the exact occurrence of events, multiple viewpoints were sought to represent the variety of data collection methods to present the diverse features of the same research problem or question (Waters-Adams, 2006). To respond to the research question about the impact of ICT in mathematics education, data collected entailed lessons showing instructions on how to employ the ICT tool to teach the various concepts of circle theorems and linear functions. A detailed investigation of the literature around the role of digital technology in education was also included in the review. Recorded videos about the use of GeoGebra application in the lesson were observed and analysed.

To aid the analysis, the lessons reflections were examined and scrutinised. The reflections on the lessons allows the researcher to step back and look at what has been achieved and what could have been changed to improve their performance. According to Mizuno et al., (2016), a reflection offers practitioners a view of their lessons for critics and analysis. Various methods are employed for lesson reflection and the one used in this study is the student observations strategy. This involves getting feedback from the students regarding what has been delivered to them. Alternatively, giving the students a practical exercise that they are required to complete can tell the teacher about what they learned and what they did not understand (Mizuno et al., 2016).

During the data collection process, Mizuno et al., (2016) mentioned that understanding the language and the body expression of the research participants helps to understand their perceptions as well as their values regarding a particular matter of investigation. It ultimately helps the researcher to analyse the data accurately.

3.9 Questionnaires

Questionnaires were used to collect data from 40 maths teachers to analyse and present their views. The uses of questionnaires were used to supplement the information obtained from own practice through the case studies. Apart from being an effective way of obtaining data, using questionnaires has the benefit of obtaining a large of data within a limited time constraint. The results of questionnaires when analysed allows more scientific comparisons than results from other researches (Popper, 2004). However, most phenomenologists claim that the use of questionnaires in researches are unrealistic since by using it , a researcher requires for only limited information without any elaboration. Also, it is not obvious to know whether the respondents answered the questions truthfully or not when questionnaires are used in researches (Ackroyd, & Hughes, 1981).

3.10 Participants

3.10.1 Students

The participants were students from my own class. The selection of students was purely due to convenience taking into consideration gender and academic achievements. Ten students in the class were approached for the feedback on the lesson. The views of the sample students were analysed and deductions made representing the entire population. In defining the implication of the perception of the sample on the entire population, Polit et al., (2001) indicated that samples are used in researches because they are seen as 'a proportion of a population'. A smaller sample is justified because in researches involving qualitative methodology, it is typical 'to study a few individuals or a few cases' to make generalisation (Creswell, 2011, p.209). Also, Ritchie et al., (2003) recommend the need for a small sample size within qualitative research because phenomena only need to appear once to be part of the analysis. The advantage with the small sample is that it saves time and is convenient (Creswell, 1998).

3.10.2 Teachers

To ensure that the information from participants satisfied the purpose of the research, ten teachers were selected from different social and economic backgrounds for the interviews. Whilst a total of 50 questionnaires were administered to selected schools, only 40 were received and analysed. The teachers were selected from both comprehensive state and private schools with varied teaching experiences. The information collected from teachers taking into consideration age, sex, socio-economic status, experiences of service mirrors the larger population that the study intend to represent (Center, 2015).

The table below states the teachers' positions, and other vital information used to aid the analysis (Thorpe & Tran, 2015).

Teachers	Length of servi	ice (years)	Age			
Code	Years in service	In school	(Years old)	Gender	Appendix	
T1	10	5	30-35	Female	15	

Table 1 Mathematics teachers: interviews

T2	15	10	30-45	Female	16
T3	8	3	30-35	Female	17
T4	10	4	30-45	Male	18
T5	20	9	30-35	Male	19
T6	15	4	35-45	Female	
Τ7	8	8	30-45	Male	NOT
T8	12	7	35-40	Female	INCLUDED
T9	14	5	30-35	Male	
T10	10	8	30-45	Male	

Table 2 Mathematics teachers: Questionnaires

Gender	Frequency	Percentage	The data collected reflected
Male	24	60	mixed gender composition
Female	16	40	of 60% male teachers and
Total	40	100	40% female
Age of teachers			
Under 30 years	6	15	The results demonstrated
30-40	24	60	that 85% of the teachers
40 and above	20	25	were above 30 years
Total	40	100	
Teaching Experience			
Under 5 years	6	15	The results indicated that
5 – 10 years	24	60	60% of the teachers had
10 and above	10	25	between 5 to 10 years of
Total	40	100	teaching experience.
Qualification			
Diploma	10	25	The research revealed that

Degree	24	60	75% of the teachers had a
Masters	6	15	bachelor's degree and
Total	40	100	above, very qualified to
			teach the subject.
ICT Experience			
Below average	12	30	The results demonstrate that
Average	24	60	30% of the interviewed
Above Average	4	10	teachers had below average
Total	40	100	ICT use.
ICT in teaching	Frequency	Percentage	
Mathematics			
Never	2	5	The results indicated that
Sometimes	32	80	95% of teachers use ICT in
Always	6	15	maths education.
Total	40	100	

3.11 Materials/research tools

The materials used are found in Appendix 5. Three types of interviews were considered before deciding which method would best suit the study on understanding the impact of students' learning with an integrated ICT tool. The three distinct types of interviews considered were the structured, semi-structured and the unstructured. The semi-structured interview was preferred over the other types of interview due to its ability to represent a large amount of information (Kvale, 1996). Structured interviews are used where the research is exploratory and it is difficult to take account of a catalogue of potential pre-codes research (Mauthers et al., 2002).

3.12 Recording interview data

All data obtained throughout the interview were carefully recorded on audiotape together with documented field notes to enrich the taped discussions and to aid analysis. Note-taking is important to capture the interview and it acts to supplement the data stored on the audiotape (Holloway & Wheeler, 2002).

The subsequent issues were reflected when audio-recording to safeguard a fruitful interview: I obtained consent to use the tape recorder at the beginning of the interview. Again, all the participants agreed to its use at the start of the interview. By using the tape recorder, the researcher could maintain eye contact to preserve the applicants' words throughout the data collection (Holloway & Wheeler, 2002). The tape recorder and the electric power point in the room were verified before the interview to guarantee a good working condition. An additional back-up with batteries was available in the unlikely event of an electric power cut. Finally, all audio tapes used were properly addressed with dates, name, gender and the number of the participants interviewed to ensure that the right interview was analysed at each stage of the data analysis.

3.13 Recording lesson episodes' data

Recording one's own practice experience ensures that the research is as objective as possible. Although studies on own practice examine life in its usual setting, caution must be maintained not to extinguish the interactional honesty of the setting (Bateson, 1979).

To be able to keep track of the information needed for the research, it was vital that all data are collected as one goes along (White, 1998). Philip White, commenting on the importance of recording every detail when researching yourself, reiterated that collecting all possible data on own practice has the added benefit of the ability to examine the information with the available field notes. By immediately recording the data as it happens, this grants the researcher the creative, complex, and profound understanding of practice to maintain a deeper or more qualitative level of information than when collected at a later date on previous conventional sources (White, 1998). In analysing the data, it is a good practice to always keep in mind, as one proceeds with the analysis of data, that such research is a dynamic, and cyclically recursive, process of inquiry (Ladkin, 2004).

3.14 Case studies

The uses of case studies are widely noted in social sciences researches. The design of a good case study aids the researcher to collect, analyse and present data in a fairer manor (Yin, 2009). The use of case studies in both quantitative and qualitative gives the researcher the advantage to understand changes due to complexities related to personal procedures that have appeared through broader communal context (Cronin, 2014). However, the disadvantage here

is that, the data obtain from case studies can be complex hence presenting researchers with the difficult to analyse and present their findings especially due to multiple cases inherited in the study (Soy, 1997). Most researchers use case studies to improve and introduce new theory in order approve or refute an existing theory, to explain a situation, to explore or to describe a phenomenon or an object. Case studies are also used for empirical and descriptive research (Solberg et al., 2009). In descriptive researches case studies are used for observed situations whilst in empirical researchers they are used for enact among parties to increase knowledge of the actions. This research combines the benefits on both the descriptive and the empirical approaches.

3.15 Data analysis methods

Rubin and Babbie (2012) cite two major data analysis methods, quantitative and qualitative data analysis. This research uses both the qualitative and quantitative method analysis. In a case of qualitative data analysis, the researcher generally obtains the response with the help of semi-structured interviews and analyses the response along with the secondary data available from other journal articles (Blumberg et al., 2014). Analysing own research involves the ability to be objective with your own findings as much as possible (Hubbard & Power, 1999).

Again, by using the qualitative data analysis technique, the responses of my actions and lesson episodes were matched with the information stated in different scholarly journals and articles (Blumberg et al., 2014). Since the study is focused on analysing human beliefs and perceptions, the choice of quantitative data analysis technique was inappropriate to suit the purpose of the case studies, since the data collected was purely non-numeric (ibid.). Blumberg et al., (2014) state that, generally, quantitative techniques are chosen in case of determining the percentage rates of satisfaction and usage. Using the qualitative analysis techniques allowed the researcher to analyse the opinions and beliefs of the respondents (ibid.); hence, the choice of qualitative analysis technique over quantitative technique for the case studies.

Analysing one's own research data required me to think carefully with the aid of the TPACK framework. This process involved visualisation and scrutinising my own practice without pre-judging own efforts. This procedure enables a researcher to act as detective, observe, contemplate, inspect and look for configurations and patterns alive in their own data (Pine,

2009). Also, the objective of the study is to critically analyse the data in the light of the Knowledge Quartet to provide explanations as close to the available data and experience achieved when the researcher avoids the predetermined dichotomies, held views are pronounced as axioms by specifying rigid formalisms about how to conjecture and research (Pine, 2009).

The analysis used the thematic content analysis approach to analyse the data. The use of thematic content analysis ensures that qualitative data for the case studies are analysed completely using patterns, and broad themes, instead of counting and analysing statistically. The decisions around the selected size of circles as well as the level of unit of overlay were grounded on specific trends recognised in the data. The four intersecting knowledge regions and the three main knowledge units (content, pedagogy, and technology) of the TPACK framework model were used as major coding categories.

3.15.1 Triangulate data

Triangulation consisted of the use of numerous data sources. To maintain validity of results, a researcher combines triangulation with reflexivity (Banford, 1996). The act of reflection consists of critical awareness, conscious by individuals about their specific prejudices, preferences, and expectations both before the study starts and as it develops (ibid.). The research question was studied from different perspectives before arriving at the conclusions. The different data sources identified for the triangulation included lesson episodes, the reflections, lesson plans, interviews with students and samples of students' work. According to Ferrance (2000), studies that involve own practice usually consider an extensive variety of sources to guarantee the researcher the needed validity.

To obtain the right information, questions were continuously asked to ensure that the data collected was indeed suitable to answer the research question (White, 1998). By examining the data collected against any previous data held, this offered the researcher the bases to decide which information should be focused on for the study (Pine, 2009).

3.15.2 Coding and analysis teachers responses

Coding is a systematic way to combine and organise data based on relevant themes, topics, ideas, terms, concepts, keywords, terms, and phrases (Albugami, 2015). Coding is a very useful process that helps to retrieve the most salient data from a large data set, and helps the

researcher to compare and analyse the data properly (Albugami, 2015). Codes from the categories enable the researcher to manage data by labelling, storing and retrieving it according to the codes (Miles & Huberman, 1994). As the key elements of the data are identified, it was possible to categorise the data for coding. Designing a methodical method to analyse the data develops through the realisation of the requirements of the research question.

Coding becomes possible through categorisation and clarification of the data to determine how to organise the data results, in an order of significance and frequency (Pine, 2009). By using the existing predefined coding (Appendix 3), colours were used to highlight the frequency of the information associated with the predefined codes of TPACK and the Knowledge Quartet. Using colour coding enhances the researcher's ability to discover similar trends for the analysis.

3.16 Research ethics

Throughout the research, the involvement of human respondents in the project necessitated the need to abide by the ethical guidelines as prescribed by BERA (2011). Prior permission from the respondents was taken from both teachers and students and the intention of the data collection with the use and the aim of the research was fully explained before conducting the research (Bernard & Bernard, 2012). As per the Data Protection Act 1999, further care was taken to maintain the privacy of the personal information collected from the participants, which included name, date of birth, gender and other details. This information was stored on a secured data stick with password protection.

Moreover, care was taken to avoid any kind of coercion or pressure over students so that unbiased views regarding the questionnaire could be obtained (Blumberg et al., 2014). During the data collection process, prior permission from the students and teachers was obtained before any questions were asked (BERA, 2011). Also, the need for the participant to withdraw at any point of the interview if not comfortable with it was explained (Blumberg et al., 2014). Finally, together with the questionnaires, all participants were granted anonymity and confidentiality by the use of ethical codes (BERA, 2011).

3.17 Limitations of the research

While the current study has presented findings that have an impact on the usage of technology dynamics to teach linear functions and mathematics in general, this implication cannot be generalised. The scope of my study poses a limitation on the findings in that it solely depended on two lessons, interviews and questionnaires on 40 participants from 5 schools. As such, it may not be applicable to many other classes and schools that uses technology dynamics. Additionally, the suggested implications might not apply if the same research is conducted but with a different class and a different technology. Also, research on my own practice may not produce the objectivity needed for this study. The following is a list of challenges that were experienced during the pursuance of this study:

- Difficulties of objectively scrutinising own work and separating myself from the research.
- Inadequate documentation and transcript of the semi-structured interview results due to time pressure and school's end of year activities.
- Difficult of obtaining all the questionnaires as expected.

3.18 Chapter summary

In considering the methodology approach for the research design, it was imperative to use both the qualitative and quantitative methodology approaches within the interpretivist paradigm. The data obtained from the questionnaires, the semi-structured interviews and the case studies provided the data for the analysis. All materials used with their ethical consideration were conducive for the chosen research design.

The analysis of the findings emerging from the application of the design questions are discussed in the next section.

CHAPTER 4: DATA ANALYSIS

4.1 Introduction

This chapter of the study addresses the analyses of the results collected from the semistructured interviews and questionnaires. Both quantitative and qualitative analytical tools were used to analyse the responses from the teachers. All data collected from the participants during the interviews was analysed as soon as the researcher was conversant with the data (Braun and Clarke, 2006). The audio from the semi-structured interviews was transcribed and read many times in order to develop a coding scheme (Bogdan & Biklen, 1992). The initial codes were then used to generate themes to aid the content analysis of the data (Albugami & Ahmed, 2015), and subsequently answer the main research question: 'what is the impact on ICT integrating in mathematics education?'

The following research questions guides this study:

- What are the benefits of ICT in Mathematics education?
- What are the barriers to ICT in mathematics education?
- What factors will facilitate ICT in Mathematics education?

4.2 Rationale of analysis for the interviews

The data analysis was done using illuminative analysis, within the interpretivist paradigm. By adapting the constant comparative method, the data from each interview was compared with the elements, phrases, sentences, and paragraphs of the other interviews (Thomas, 2009). The analysis process consisted of familiarising myself with the data. A general pattern emerged from listening to the tapes, transcribing interviews from tape to paper, and reading over the written transcripts several times (Rubin & Rubin, 2005).

Following the formal process of analysing the qualitative data, the data obtained was coded and main ideas in the data identified. The process continued with gathering identical information together in categories to ensure that similar themes and ideas were connected (Rubin & Rubin, 2005). Organising the similar themes and ideas in a table also provided a useful snap short of the information to be analysed, as 'valid analysis is immensely aided by data displays that are focused enough to permit viewing of a full data set in one location and are systematically arranged to answer the research question at hand' (Miles & Huberman, 1994:p.432).

4.3 Possible outcomes

By analysing the findings from the case studies, interview from teachers in conjunction with the literature review, I expect to find that the use of ICT has a strong impact on mathematics education. It is also expected that the teachers' perceptions of the barriers to ICT integration are related to their experience of using ICT. Finally, I expect to find that teachers' will report that the school curriculum is a barrier to the use of ICT in mathematics education. To improve the impact of ICT integration in mathematics, I expect teachers to recommend that schools must encourage ongoing professional development to boost their ICT confidence. The research is somewhat limited by some teachers' difficulty in providing answers that honestly reflect their perceptions on the subject, and the inevitable bias of participants towards the use of ICT.

4.4 Dissemination

Ethical considerations are considered in the dissemination of the findings (ESRC, 2010). According to BERA (2011) researchers must refrain from publishing unethical materials and participants should be made aware of what is reported. The dissemination tools considered for this study includes peer review papers, research reports, and presentation of report at both national and international reviews. A report of the findings will be presented first to participants and will be encouraged to comment on the study. Furthermore, the report will be sent to interested schools for deliberations and recommendations on the views raised on the perception of ICT pedagogy.

Samples of the records of the semi-structured interview schedules with transcripts are seen in Appendices 15 to 19.

4.5 Case study 1: Circle theorem

The target audiences are students in year 9 (14-15 years old). I will discuss the rational for choosing the TPACK and KQ framework for the plaining, delivery and the findings from the taught lesson. The reflections and the interviews with students on the impact of the use of

ICT form the bases of the analysis. I planned to use both open and closed-ended questions, as advocated by Malibar and Pountney (2002), whilst aiding whole-group discussions through collective learning (Ruthven & Hennessy, 2003; Godwin & Sutherland, 2007).

For the purpose of the analysis, the students in the class were put into groups. Group A consisted of students S1, S2, and S3, who shared a computer. Group B included students S4, S5, S6, who shared a computer as depicted in the table 3 below.

Table 3: Computer sharing

S1	S2	S 3	S4	S5	S6		
Group A			Group B				

Appendix 1: Demonstrate the application of using Technological, Pedagogical Content Knowledge to design the Lesson with examples of activities that were considered.

Appendix 2: Activity Sequence, including the objectives of the lesson with the task to be completed.

Appendix3: A detailed analysis of the discussion on each circle theorem concept.

: The use of GeoGebra to demonstrate how to resolve the misconception on the topic.

The following were the series of planned task to aid pupils understanding of each key concept.

- Task 1: The angle at the centre of a circle is twice the angle at the circumference of the circle from Task 2: Angles inscribed in a semicircle are right angles the same arc.
- Task 3: Angles in the same segment are equal.
- Task 4: Cyclic Quadrilateral Theorem
- Task 5: Alternate Segment Theorem
- Task 6: Perpendicular & chord
- Task 7: Angle between tangent and radius

4.6 Case study 2: Linear functions

The aim of this study is to establish how ICT is used in teaching linear functions. Using GeoGebra, several episodes can be analysed to determine the effectiveness of the learning.

4.6.1 Lesson Episode 1: Drawing Graphs

The aim was to use Geogebra to draw graphs by exploring how to adjust the y- axis, x - axis and to edit in the graphic view. This was a one-hour introduction lesson in GeoGebra. The activities of the lesson were followed by the completion of a worksheet whereby each student used the software in generating certain example of a graph family in the form: y = c, x = c, y = x + c, y = mx, y = x + 2, y = x - 3, and then record them on the worksheet (Appendix 4 and 6).

In order to answer the research question, what is the impact of teaching mathematics using ICT? Over the major part of the lesson, the format of the activity alternated between short quizzes and the whole classwork discussions. The use of the software enhanced the accessibility of the tasks and made it less tiresome to the students to complete the classwork. Also the use of software function assisted the teaching in the interpretation of the variable in the linear graphs through colour coding. Henceforth the use of different colours of the graphs helped the students to see which colour goes with which function to aid the interpretation of drawn graphs. In the lesson, I posed a speculative question concerning the sloping of the lines in different ways, which then led to similar trailing episodes inserted in conventional investigation alongside introductory review.

4. 6. 2 Lesson Episode 2: Slope and y-intercepts

In this lesson, I introduced how to draw a linear graph by introducing students to how to construct two distinct points followed by joining the two points together to produce a straight line. The relation between the line drawn was established and changed into a function or equation in the form of $\mathbf{y} = \mathbf{mx} + \mathbf{c}$. To transform the line into a linear equation, I guided the learners to create an object to show the equation of the line already drawn to appear on the screen by inputting this equation of $\mathbf{y} = \mathbf{x} + \mathbf{1}$, in the input box (Appendix 6)

The next step was to interpret the characters in the equation. Using a cursor in the grid, I was able to help the learners to identify the required interpretation on the equation of a line and also identify the constant of the function as the point where the line intercepts with the y axis.

While at this point, I was able to make changes to the y-intercept by moving the line or the points to achieve intercepts such as 0.25 and 2.

In the next step, I explained the significance of the slope in the linear function and how it is determined using the GeoGebra software. By making shifts using the drawn line, it was possible to explain and demonstrate that the slope was determined by the movement to the right from the y grid at the intercept and a movement to the position of the value of the x in the grid. By illustrating that the movement of the steps to the right can be used to describe the ratio in the gradient, it was possible for the learners to appreciate the meaning of gradient as expressed in ratios. After guiding the students to determine different slopes including $\frac{1}{2}$, 3, 5, and so on, I then guided the students to work on their own examples (Appendix 4 and 6).

In order to answer the research question, what is the impact of teaching mathematics using ICT? I physically moved around the class to find out what the students were able to complete in the lesson. Also, key questions were used during the lesson to be able to determine the barriers in their learning. By using questions such as 'What is going to be the next stage?' I was able to keep most students focused on the task. Similarly, the brief pause of lesson review and questioning provided time for students to think through on what the next steps would be before demonstrating the next activities. Students reported that the use of questions facilitated their understanding of the topic better.

4.7 Teachers Interview questions

In order to answer the research questions exhaustively, codes were further grouped into the following categories with their respective interview questions:

4.7.1 Part 1/Code 1: Benefits of ICT pedagogy

Interview question 1: What do you believe to be the benefits of using ICT in mathematics education?

In the answer to the above question, T1 (Appendix 15) and T7 stated that they believe that the integration of ICT into their pedagogy helps them to reinforce learning within the classroom. Both also mentioned that if ICT were included in the pedagogy, then there would be higher student interest and more active participation in learning. T3 indicated that 'interactivity and instant feedback to class work' was possible through the use of ICT (Appendix 17). T4

mentioned that due to the interactivity provided by the integration of ICT, it had become easier to supervise students in the classroom (Appendix 18). The use of ICT has the ability to keep most pupils on task, which has improved pupils' behaviour for learning and enabled better class room management (T4, Appendix 18).

According to T4 and T9, the integration of ICT tools assisted the use of modern technologies. T1 indicated that modern technology is 'incredibly useful and speeds up the research process' when students are looking for new information (Appendix 15). For example, most teachers mentioned that with the help of projectors and interactive white boards, important learning videos were embedded into the pedagogy. T3 and T6 further stated that integration of ICT tools helped to improve their pedagogy by improving their teaching skills (Appendix 17). Thus, the use of ICT technologies helped in developing suitable learning opportunities for the students (T3, Appendix 17).

When demonstrating complex diagrams and key concepts in mathematics T3 (Appendix 16), indicates that ICT tools can promote visual understanding of key concepts. Added to the enhancement of learning is the effective use of ICT tools to increase attainment in processing information, concept mapping and modelling. Online dictionaries and 'edublogs' also help to increase the understanding of key words in the English vocabulary of learners when utilised by teachers (T7 and T10).

When answering the question on the benefits of integrating ICT pedagogy, most teachers related their answers to students' attainment levels. The use of ICT helps to meet the requirements of different learning styles, which makes it possible to achieve higher academic attainments (T2, Appendix 16). Most teachers acknowledged that the integration of ICT has worked positively to close the learning gaps and raise the attainment of pupils in their departments. Further, all teachers agreed that the use of ICT enhances learning through independent work. T5 commented that 'learning achievement goes up when teachers allowed differentiated independent learning' (T5, Appendix 19). Through independent learning, guarded questions are offered to students to improve their learning.

Most teachers also believed that the use of ICT can accelerate learning by ensuring adequate work to support both the weaker and more able students in class. According to T5, the use of ICT has the ability to 'offer a greater variety' of work to learners (Appendix 19). ICT has the added advantage of extending the 'breadths and depths of interaction' of all the learners (T5,

Appendix 19). T3 believed that ICT helps students to complete differential tasks within the school and helps teachers to provide quick feedback on completed work to improve students' attainment. However, T6 believed that there is little evidence of the impact of ICT on pupil attainment, and that ICT is not necessary for higher learning gains in studying mathematics. Refuting this claim, T5 agreed that there is a significant impact of ICT on pupils' attainment (T5, Appendix 19) when used consistently to complete both schoolwork and homework. Most teachers commented that negotiation of learning outcomes is better with both students and teachers when ICT strategies are effectively incorporated into pedagogy.

4.7.2 Part 2/Code 2: Barriers to ICT pedagogy

Interview question 2: What are the barriers to integrating ICT in mathematics education?

In answering this particular question, the teachers mentioned several obstacles or barriers that are creating problems in integrating ICT pedagogy into the teaching of mathematics in secondary schools. Particularly for T3 and T1, the main barrier to the integration of ICT in the pedagogy of the departments is hardware problems. Most teachers said that the lack of proper maintenance of ICT tools such as computers and interactive white boards discourages them from using ICT (T3, Appendix 17; T1, Appendix 15). They further mentioned that in most secondary schools, there are insufficient numbers of computers to enable student access, with limited computer suites and long waiting times for booking. Schools often need to work with out-of-date ICT tools that are slow and inefficient. Very slow broadband with poor networking facilities also makes the internet connection in schools very slow, which can make using online resources frustrating.

The absence of 'curriculum, resources and software problems' was reported to be a major barrier to using ICT integration (T4, Appendix 18; T5, Appendix 19). The inability to effectively deploy ICT in their lessons was attributed to poor ICT training on using software applications and time wasted looking for the appropriate resources. This has greatly affected their confidence and discouraged them from using ICT. The absence of experience in using ICT due to lack of confidence and skills has made most teachers resist ICT integration. This attitude of teachers has led to unsuccessful ICT integration in some schools. In addition, due to the negative perceptions and conservative attitudes of some teachers in adopting modern teaching methods, the lack of interest in using ICT was reported especially among teachers who were in their 40s (T2, Appendix 16; T4, Appendix 18; T6; T8).

T1 and T5 mentioned that lack of adequate skills and times are other obstacles to ICT integration (T1, Appendix 15; T5, Appendix 19). This was agreed by T3, who stated that the lack of 'teachers' expertise' remains the greatest barrier in his school (T3, Appendix 17). In addition, the inability to have the time to train due to heavy school workloads worsens the situation (T5, Appendix 19). Restrictions of the national curriculum were also mentioned as barriers to integrating ICT. T5 stated that the time and space available on the school's timetable for the coverage of the entire subject curriculum and the lack of interest of the school teachers make it difficult to integrate ICT into his secondary school's pedagogy (T5, Appendix 19).

Prior ICT experience was reported as a significant obstacle to integrating ICT into teachers' pedagogy; for example, T1 refused to use ICT due to the lack of 'past experience' in using computers (T1, Appendix 15). Supporting this claim, T4 mentioned that although several attempts had been made to include ICT in his pedagogy, he continues to be reluctant and remains uninterested due to personal ICT experiences in the past resulting from poor ICT exposure in teacher training institutions (T4, Appendix 18). Meanwhile, T2 could not integrate the ICT properly into her pedagogy due to the belief that ICT applications were difficult to learn (T8).

In this section of the interview, most of the teachers mentioned that the lack of sufficient knowledge regarding the use of ICT tools has created problems in integrating ICT pedagogy in their departments. Compounding this problem is the lack of knowledge on using up-to-date software (T2, Appendix 16). Most teachers also indicated that lack of available support staff to provide the needed knowledge prevents them from using ICT (T2, Appendix 16). According to T3, (Appendix 17), not knowing whom to approach for assistance is the greatest barrier to using ICT. Most teachers lamented the absence of whole-school initiatives to provide technical knowledge of ICT tools and applications when they are in difficulty, which further discourages them from using it (T5, Appendix 19). Nevertheless, T5 reiterated that 'having champions within the department matters most and teachers are not expected to know everything as a leader' (T5, Appendix 19). He believed that 'a greater opportunity to be

creative' is vital in leading a department to integrate any educational initiative (T5, Appendix 19).

4.7.3 Part 3/Code 3: Facilitators of ICT pedagogy

Interview question 3: What are the factors that will facilitate ICT in Mathematics education?

In response to this particular question, teachers commented on the different training and development strategies that enable the use of ICT. Also mentioned were the existing tools available to promote ICT. All of the research participants stated that they use several tools and applications to facilitate ICT in their pedagogy. Three teachers in particular identified their ICT facilitators as 'visual and dynamic aids' (T1, Appendix 15; T4, Appendix 18; T5 Appendix 19). The tools discussed included 'videos and films, interactive boards, YouTubes, tablets' (T1, Appendix 15 and T4, Appendix 18), 'educational websites and online applications including calculators' (T2, Appendix 16), and 'WhatsApp, Twitter, [and] mobile phones' (T3, Appendix 17).

According to L1, the integration of ICT tools allows learners to develop effective analytical skills and problem-solving skills. In addition, the development of ICT tools has helped students in the development of the conceptual understanding of mathematics (T6 and T7). T6 and T7 agreed with other teachers on using ICT tools such as tablets, whiteboards, and internet applications to improve learning. The Mathematics teachers mentioned that they were using the interactive whiteboard and many different online applications to facilitate ICT in their pedagogy (T2, Appendix 16; T4, Appendix 18).

The interviews demonstrated that in most cases, the lack of effective ICT training makes the teachers technologically handicapped, and thus the usefulness of ICT tools is not fully recognised in mathematics education. Notably, the teachers found the provision of training to facilitate the use of ICT pedagogy (T3, Appendix, 17). Both T2 and T7 were against any facilitation of ICT received externally (Appendix 16) and indicated that the ongoing inservice training equips teachers with the right skills to implement ICT pedagogy. This is because T2 believed that the lack of continuous training in the past by external training providers led to less efficient ICT integration (T2, Appendix 16). T6 and T8 pointed out that if the schools' administration involved more experienced and knowledgeable local teachers, then they could deal with the current situation of integration in a better way. Recruiting a

technology co-ordinator was therefore thought to be very important for the integration of ICT in the secondary schools' pedagogy (T4, Appendix 18).

In relation to training opportunities, the teachers discussed several professional on-going developments within their schools. T2 and T3 mentioned that both the departmental training sessions and the training sessions provided by the school impacts on their ICT pedagogy (T2, Appendix 16; T3, Appendix 17). Both T2 and T3 commented that their school's professional development activities include working in teams to support all (Appendix 16; Appendix 17). However, schools may be reluctant to provide continuous professional development due to other school priorities (T1, Appendix 15). The need to provide training sessions through INSET/staff training and sharing good practice was seen as vital to support teachers to integrate ICT (T3, Appendix 17).

4.8 Teachers: Questionnaires

The data was collected from 40 out of 50 teachers using interviews and Questionnaires from 10 schools in south London. The responded represented 80% of the target group. The following represent the analysis of the data with the findings presented according to the objectives in the following sections.

The following data was obtain using SPSS tool to analyse the questionnaires

Benefits of integrating ICT pedagogy;

- Sustained student interest: 55% Agreed
- Aid lesson planning with wide range of resources: 75% Agreed
- Improved the attainment of children: 60% Agreed
- Enhance interactivity and exploration: 75% Agreed

Barriers to integrating ICT pedagogy;

- Lack of curriculum resources: 66% Agreed
- Software and Hardware problems: 32% Agreed

- Lack of experience with ICT tools: 35% Agreed
- Negative attitudes: 40% Agreed

Facilitators of integrating ICT pedagogy;

- The Availability of ICT tools: 65% Agreed
- Experience and personal knowledge of using ICT: 85% Agreed
- School leadership: 75% Agreed
- Opportunities for training: 26% Agreed
- The following table shows the full summary results.

Table 4: Summary data analysis using SPSS

Research							Strongl	y Agre	ed (SA	A) Agı	reed (A)
questions	Interview questions						Disagreed (D) Strongly				
							Disagreed(SD)				
Benefits of	Sustained student interest.						Improved the attainment of				
integrating ICT	SA A D SD Total						children.				
pedagogy	30%	55%	10%	5%	100%		SA A D SD Total				Total
	Aid lesson planning with wide range of						25%	60%	10%	5%	100%
	resources.						Enhance interactivity and				
	SA	А	D	SD	Total		exploration.				
	20%	75%	5%	0	100%		SA	А	D	SD	Total
							20%	75%	5%	0	100%
Barriers to	Lack of	curricu	lum res	ources			Lack of	f exper	ience v	with IO	CT tools
integrating ICT	SA	А	D	SD	Total		SA	А	D	SD	Total
pedagogy	20%	66%	10%	4%	100%		50%	35%	10%	5%	100%
	Software and Hardware problems						Negativ	ve attitu	ides.	1	
	SA	А	D	SD	Total		SA	Α	D	SD	Total
	68%	32%	0	0	100%		60%	40%	0	0	100%

Facilitators of	The Availability of ICT tools						School leadership				
integrating ICT	SA	А	D	SD	Total		SA	А	D	SD	Total
pedagogy	30%	65%	5%	0	100%		15%	75%	10	0	100%
	Experience and personal knowledge of						Opportunities for training				
	using ICT						SA	А	D	SD	Total
	SA A D SD Total						70%	26%	4%	0	100%
	10%	85%	5%	0	100%		L	1		1	·1

4.9 Chapter summary

The responses of teachers to the interview questions and questionnaire revealed a common trend. The emerging themes on ICT-integrated pedagogy demonstrated that teachers held contrasting views on the benefits and obstacles in ICT integration. In terms of the perceptions of teachers' role in ICT, most teachers lamented the lack of adequate ongoing training to promote ICT integration. Additionally, the lack of technical support, lack of accessibility of ICT tools, and lack of effective training makes it difficult for teachers to deploy any effective strategies for using ICT to change pedagogical practices.

The next chapter discusses the findings from the data collected from the participants and how these relate to the findings of the literature review. All associated theories between the findings and the research are discussed.

CHAPTER 5: EVALUATIONS AND FINDINGS

5.1 Introduction

The aim of the research was to find out the impact of ICT on mathematics education? To be able conduct the research detailed, I used own teaching experience through case studies and the responses from both students and teachers to answer the following research questions;

- What are the benefits of ICT in Mathematics education?
- What are the barriers to ICT in mathematics education?
- What are the factors that will facilitate ICT in Mathematics education?

The following sections will discuss the findings from the case studies and the responses from both teachers on the impact of ICT in mathematics education. First, the reasons for selecting GeoGebra as a tool for the ICT lesson will be discussed followed by the case studies and the interviews from teachers.

5.2 Rationale for the choice of GeoGebra

GeoGebra was use as ICT tool due to its ability to enhanced understanding through the multiple representations of views and its user friendly nature. In Singapore, Venkataraman (2012) conducted a study in secondary mathematics examining inventive activities used to develop geometrical thinking ability in GeoGebra. The study indicated that students who studied through the software recorded significant progress in understanding mathematics, a basis for additional deductive reasoning in mathematics (Ritz, 2009). The research concluded that the ability to use the dragging feature in GeoGebra was the vital component that impacts the generalisation achieved by students using the program to learn, since abstract concepts are made more meaningful through visualisation (Venkataraman, 2012). The GeoGebra software provides learners with the interactive, reasoning and visualisation (Hohenwarter, 2007) necessary to understand key concept. The tool offers the teachers the benefit of inviting learners to contribute to a compelling investigation through affirming the fundamental computational and mathematical essentials (Clements, 2004).

Using GeoGebra to teach mathematics is more effective than using other software because it has the ability to provide learners with deeper understanding of the key concepts (Chimuka,

2017). Again the use of GeoGebra allows students to identify and explore different theorems through the process of 'trial and improvement' (Ruthven and Hennessy, 2003). The pedagogy in mathematics using GeoGebra has the additional ability to empower students to begin with the process of discovering new problem-solving skills, which aids the understanding of a mathematical conjecture (Ritz, 2009). According to Chimuka (2017), using GeoGebra enables students to perform experiments and make discoveries that enhance students reasoning skills in a way that would be impossible through 'chalk and talk' instruction.

5.3 Lesson design

A lesson design simply refers to the strategy employed by the instructor to structure, develop and implement a lesson plan. The effectiveness of a lesson design is characterised by the following features. First, the learning aims and objectives should be known at the start of the lesson. The activities were centred on the learning outcomes. In the lesson design, clear steps were set on the consolidation of pupils understanding in the previous lesson before the introduction of new concepts. The design of the lesson took into consideration the provision for students to be allowed to work in teams or pairs. The importance of this is to enhance their understanding by providing them with the chance to support their peers to understand and develop knowledge in ways in which the teacher cannot assist.

Contrary to these good designs, poorly designed lessons tend to be fully instructor-directed and teacher focused. They fail to avail of the opportunities for the exploring of knowledge prior to the provision of the instructions. A well-designed lesson usually encourages the instructor to begin the lesson by determining how much knowledge of the lesson's topic the students have. This may be through group discussions or a report out of the class action. Also, it allows the teacher to pause in breaks to ask questions and allow the learners to express how much of the lesson they have understood.

5.4 Frame work for ICT lesson

Using the TPACK and the Knowledge Quartet framework for the lesson design necessitated an exploration of the prior knowledge of the students and, as such, encouraged the students' input. The lesson planning is viewed through the Technology, Pedagogy, and Content Knowledge (TPACK) framework). By using the TPACK framework, a teacher is able to draw attention to the diverse kinds of information needed for effective integration of a digital technology (DT) (Shulman, 1986). The TPACK framework incorporates technology and teaching in a desired method by providing the interconnection between three categories of knowledge delivery that combine for fruitful digital incorporation (Koehler & Mishra, 2009; Cox & Graham, 2009).

The TPACK framework was considered against other modules such as Maslow's Hierarchy of Needs or Bloom's Taxonomy modular because it affirms technology, content and pedagogy to be the key advocators of education whilst the former modular explains and shapes the dynamics of learning processes (Archambault & Barnett, 2010). Notwithstanding the number of challenging assumptions of using the TPACK framework, for example, the assumption on secluded components being understood as separate overlying circles representing empirical reality is absurd. Also, the supremacy of content knowledge being considered as one of the central educational apprehensions and the unrealistic assumption of the model without drivers and goals makes TPACK student-centred framework, a worry for many educators (Clark-Wilson, et al., 2014).

I evaluated my expertise in line with the TPACK and PCK frameworks. As part of my planning for TCK, integrating GeoGebra in my lesson required a detailed understanding of how the various circle theorems were going to be introduced. For example, the need for calculating angles and descriptions of the various theorems were identified as part of my TCK expectations. The use of TPK helped me to be mindful of my role as a teacher and transmitter of knowledge. Using this role allows teachers to correct errors or misconceptions in the learning process (Hadjidemetriou & Williams, 2002). My planning for PK and PCK enabled students to carefully be guided to use an accurate collective dialogue.

5.5 Lesson plans

A lesson plan is a description, a thorough learning trajectory of a sequence or order of instruction. The learning trajectories entail the learning goals and expectations, the lesson developmental path and the set of activities matching the thinking levels in the developmental path, which enable the learners to develop higher-order thinking. A lesson plan acts as the blueprint and guides for a lesson including the teaching activities, classroom arrangement, student engagement and the systematic approach of teaching the concepts.

5.5.1 Experience of teacher with ICT

To be able to plan the lesson effectively, Rohaan et al. (2010) argue that the teacher's selfefficacy with digital technology should be examined against their own practice. A welldeveloped lesson plan incorporates the best practices and correlates with the teacher's philosophy of education. Rohaan et al. (2010) indicate that when planning lessons, teachers should be mindful of their self-efficacy in connection with their capabilities with ICT. By taking into consideration of own expertise in using ICT, acknowledging the knowledge gaps and taking steps to resolve them boost the confidence with the application in the lesson.

5.5.2 Objectives of the lesson

The aim was to design a lesson and teach circle theorems using GeoGebra to facilitate students' mathematical reasoning. The topic was taken due to the difficulty of teachers to demonstrate the hidden views of the shape when using pen and paper. My aim was to develop activities that would be effective for all pupils, with the goal to inspire students to learn. Also vital was the aim to use digital pedagogies aligned with the learning goals that students already demonstrated throughout their own learning: interaction, engagement, and peer-to-peer discussions with focused assessment. In peer-to-peer learning, for instance, the aim was for me to promote students' reflections and logical reasoning with comments on peer discussions and student ownership of the learning through student-centred activities.

5.4.3 Consulted wide range of resources

To be able to teach the lesson successfully, I consulted variety of resources on the topic to understand how other teachers have taught similar lessons, since teachers' approaches to the incorporation of ICT have shaped the foundation of how they have advanced their pedagogy (Saudelli & Ciampa, 2016). To overcome my anticipated difficulties in incorporating ICT in the lesson, Saudelli and Ciampa (2016) cites that teacher's beliefs and confidence are influenced by their attitude and the chosen resources towards technology integration. In addition, I was mindful that connectivity among people, according to Siemens (2006), has the ability to transform the way that people access, interact, and process knowledge. Therefore, thought was given to how to plan effectively using numerous activities that maintain full engagement with the learning. By deploying technology as a servant, teachers are able to advance the reasoning and conjecturing in mathematics education (Goos, 2014).

5.4.4 Experience of students with technology

According to Halldórsson et al., (2009) and Murray (2016), knowing pupils' backgrounds knowledge in the topic in the lesson planning, including skills and experiences, ensures a better outcome of the lesson. The Knowledge of pupils' background prior understanding in the topic helped me to plan an inclusive lesson in view of students' different past experiences with ICT teaching (Godwin & Sutherland, 2007). When teaching difficult concepts, Ruthven et al., (2008) advocate the using of technology as a vehicle to break down key contents through a structured approach. To support students who had little experience with the tool, I planned to use scaffolding task activities. The use of scaffolding actives ensures step-by-step delivery of the key concepts to promote the development of the required technological skills (Ruthven, 2014 & Murray, 1996).

5.5 Case study 1: lesson on Circle theorem

5.5.1 Review previous lesson

A good place to start when teaching a lesson in circle theorems was to deliver a recap of angle facts relating to circles. I aimed to ask students to do some exploration into angles in circles using pen and paper. The aim was to help the students to apply their acquired knowledge and skills in previous lessons in various situations (Malibar and Pountney, 2002). For instance, the concept of calculating angles in the previous pen-and-paper lesson encouraged a feeling of conjecturing and improving on the answers to the task. After that students were given more practice, starting with simple cases and then moving on to more demanding questions involving more than the use of one theorem.

5.5.2 Misconceptions were corrected

During the lesson, time was allocated to discuss pupils' completed worksheet with feedback. This role enabled me to gain full control of the learning process, as I had the opportunity to help students to overcome their difficulties and rectify misconceptions resulting possibly from the integration of the technology. According to Becta ICT Research (2003) since computer assisted learning is individualistic it is easy for a student to face difficulties due to knowledge gaps. Also, a student is at a loss of the expectations during an ICT lesson if instructions were not carefully followed and understood. This frustration is compounded when the ICT lessons are held at a different environment from the usual classroom since the
alternation of rooms creates discontinuity of knowledge. This is because often teachers uses the ICT lessons as a continuation of the previous traditional lesson and the lessons became more beneficial to students who have already understood the concepts from the previous lesson.

5.5.3 Access to online resources

Each activity was hyperlinked to the online resources (www.mymaths.co.uk) for students to gain more practice. The use of thorough supervision of students' work in progress was required to minimise disengagement (Jones et al., 2009) and enabled me to provide directions to resolve misconceptions. To be able to accommodate for the different learning needs in the class, differentiated task were planed and online resources were used to extend pupils learning. Also, where pupils needed tailored support, to promote classroom engagement, ready-made resources were provided (Appendix 3).

5.5.4 Collective learning

I planned whole-class discussions as part of the lesson. According to Godwin and Sutherland (2007), embarking on whole-class discussions is essential in mathematics learning, as they enable collective student knowledge building. Although the whole class discussions gave all students the opportunity to receive collective feedback on their progress in the lesson, I physically moved around the class to provide individual tailored feedback and supported each student in difficulty, recognising the different needs of each student (Saudelli & Ciampa, 2016).

5.5.5 Facilitated assessment

To prompt students to make simplifications and interpretation on calculated answers, worksheets were designed to request the reason for any decision taken and to discourage generalisations (Mavrikis et al., 2013). Also, I endeavoured to caution pupils to critically scrutinise their findings and not recklessly accepted their answers from the use of ICT as their default final answers (Lu, 2008). The use of 'Why' and 'what' questions such as, "What happens to the size of angle after varying the size of the circle?" were structured into the lesson to motivate learners to remain focus on task (Appendix3).

5.5.6 Increased learning confidence

My supervisory role ensured that children were confident to think through every stage carefully through justification of proof. This student-focused, inquest-based approach to teaching with ICT (Dochy et al., 2003; Harada et al., 2008) promotes a great degree of student partnership and application of reasoning and thoughtful skills when ICT are used as mechanism for teaching (Harada et al., 2008). Once students were familiar with the vocabulary, they were able to use the right terminology and language to justify their reasoning behind the calculated results. The use of PCK demonstrated that unfamiliar vocabulary use of ICT in lessons was an obstacle to the understanding of mathematical concepts (Hadjidemetriou and Williams, 2002; Jones et al., 2009).

5.5.7 Exploration

Also, at the start of the lesson, time was allocated for exploratory-based learning (Malibar & Pountney, 2002) to offer students the chance to freely experiment and justify their taught process. This made it possible me to partner with the use of technology to provide learners with innovative motivation beyond their learning. This approach enabled learners to develop a critical thinking on key concepts (Elliot, 1998; Zbiek et al., 2007; Goos, 2014). The technological tool aided the exploration of a set of circle theorems through dragging of the points on the circumference of the circle to discover the connection between the changes in the size of the angles. For example, the theorem that the 'angle at the circumference is half the angle at the centre' is difficult to understand when using pen and paper, since most students find it difficult to realise that there could be four other cases (Appendix 3) to consider.

The use of the technology aided the design of the lessons in using strategies that enabled me to explore the key concepts of the topic in a step by step manner. Exploring one concept at a time when learning with ICT helps to structure the knowledge (Ruthven et al., 2008), which is vital to accelerate understanding of difficult concepts in mathematics education. The lessons were structured to allow the exploring of one concept at a time. Such an approach allowed me to use the technology as a servant (Goos, 2014).

5.6 Lesson report

5.6.1 Task 1: The angle at the centre of a circle is twice the angle at the circumference of

the circle from the same arc.

The test of this theorem consisted of several parts. Initially, all students (Groups A and B) arrived at a construction that demonstrated the inscribed angles and the angles at the centre intercepting at the same arc. Students from Group A quickly noticed that the degree of the angle at the centre appeared twice the degree of the angle at the circumference. Meanwhile, students from Group B varied the calculations, even though it was obvious by using the input function to compute the ratio of the degree of the angle that appeared at the centre to the degree of the angle at the circumference (entering β / α , which showed f = 2 in Algebra view). My role guided the students on the best choice of opportunities to arrive at the answer when faced with numerous ICT options. In the opinion of Monaghan (2004), teachers' role as facilitator of ICT in mathematical knowledge when guided prompts were used assisted students with their discoveries. Although this views as indicated by Sherman (2014), rather creates source of inconsistency in the general responsibilities of the teacher. This is because the teacher becomes the centre of mathematical authority and reasoning to a catalyst of architect of students' mathematical knowledge and activity. Upon performing the drag test (Finzer & Bennett, 1995), S4 stressed the several ways to visualise the inscribed angle theorem (figure 1b) while confirming what he was familiar with in the previous lesson when mymaths.co.uk was used.

5.6.2 Task 2: Angles inscribed in a semicircle are right angles

According to Saudelli and Ciampa (2016), understanding the TPACK framework equips teachers to find ways that promote learners to experiment an institutional perspective in a mathematising process. Most students started with just the drawing of the circle, preceded by exploring the drawing of a line passing through the centre of the circle. This was followed by the construction of two chords to produce the inscribed angle. As seen in figure 2a, the angle produced between the chords was 90°. The understanding of the concept was affirmed when S2 dragged the corner of the inscribed angle along to the other semicircle. In figure 2b (Appendix 3), the resultant answer was 270°, as a result of the dragging, showing that the angle of interest was measured as 90°. According to Finzer and Bennett (1995), the use of

dragging to explore the changes in the size of angles aids students understanding in circle theorem.

As specified by Blum and Niss (1991), understanding ICT barriers that could affect the different stages of the modelling processes is vital to classroom practice. The barriers identified included the slow rate at which the software was functioning perhaps due to the over reliance of the internet facilities. As a consequence, this effect influences the level of the mathematising process and often hinders learners' perceptions of tasks and the development of modelling competencies (Blum & Niss, 1991). Once the barriers were resolved, S4 and S4 constructed the inscribed angles and extended the triangle outside the circle to investigate the link between the angle externally and the angle internally. S4, S3 mentioned, "If the central angle is a straight angle then that means that the angle at the centre and the two inscribed angles added up must add up to 180 degrees" (figure 2c Appendix 3).

5.6.3 Task 3: Angles in the same segment are equal

As indicated by Bosch et al. (2006), the act of solving mathematical problems consists of both the knowing how to do it and knowing why to do it. Doerr and Pratt (2008) differentiate between exploratory models (expert, teacher) and expressive models (learner, students). Most students were successful in constructing two inscribed angles that met at the same arc. This was interesting to notice, since I was expecting students to construct another inscribed angle by deploying the drag function. However, the Group B students further demonstrated the construction, as illustrated in figure 3a, through the drag test (Finzer & Bennett, 1995) using the already constructed inscribed angles.

Zbiek et al., (2007) indicated that students engage in exploratory activity when given a preprocedure to carry out. When I probed S5 and S6 why the need for dragging the apex of the inscribed angle? Both S5 and S6 demonstrated expressive activities to affirm on why they decided on preferred approaches to justify their choices. They both understood the procedures to use through exploration using the technology without my direct interference in the required steps to take. Furthermore, for example, S5 responded "We're testing to see if it's actually true that any angle at the circumference are equal, so not just those specific angles, but any angle at the circumference will be equal". S2 also indicated this theorem's association with the Inscribed Angle (Task 1, Appendix 3) Theorem and suggested measuring the central angle. The students in Group B recognised the connection between this theorem and the Inscribed Angle Theorem by entering β / α , to obtain f = 2 as seen in the Algebra Side bar and typing $\beta + \alpha = 360$ degrees, as seen in figure 3c Appendix 3).

5.6.4 Task 4: Cyclic Quadrilateral Theorem

The awareness of the knowledge of dragging shapes in dynamic geometry (Finzer & Bennett, 1995) allowed me to guide students cautiously on how to carefully drag a shape to preserve its properties. For example, with the aid of dragging the shapes, the class was able to visualise the nature of the change in angles and through investigation confirm that opposite angles of a cyclic quadrilateral add up to 180^o. The Cyclic Quadrilateral Theorem raised an important discussion on construction for the Group A students who used the input bar and confirmed that the opposite angles of the cyclic quadrilateral were supplementary.

Most students started by constructing a circle and finding their quadrilaterals by indicating four random points on their drawn circle and joining them through line segments. S2 clarified the results by measuring the angles; S1 confirmed it again by entering the total of the opposite angles ($\alpha + \beta = 180^{\circ}$ in the Algebra view of the input bar. The peer-to-peer discussion that emerged made the students understand and affirmed that the construction of the cyclic quadrilateral was correct (figure 4a, Appendix 3). S2 noticed that the total of the opposite sides would be wrong if the points were not indicated on the circle when dragged (Finzer and Bennett, 1995) out of the circle (figure 4c). S3 quickly managed to fix it to ensure that the total of the opposite angles added up to 180 degrees. S4 was able to affirm through the dragging test (Borwein, 2005; Ruthven, 2009) that the opposite angles were always supplementary. The ensuing discussion on the activity triggered a lot of enthusiasm among the students.

5.6.5 Conclusion

Pedagogies with ICT have the benefit of high achievements in mathematics education through motivation, concept formation, high-order thinking, visualisation, analysis, abstraction, comprehension and retention, and rigour. Students' feedback and interviews revealed that lessons delivered with digital tools have the ability to transform learning.

When seen through the lens of the TPACK, the lesson demonstrated the need to carefully think through the various kinds of knowledge in the planning of ICT-integrated learning for a successful lesson. I believe the incorporation of GeoGebra, an ICT tool helped to create a

new environment not usually experienced in lessons. Most students demonstrated high enthusiasm about using the software to advance their understanding of key mathematical concepts. As such, using ICT continues to justify the understanding of key mathematics concepts, providing the needed intuition, insight, and exploration of similar but new patterns and relationships (Borwein, 2005; Ruthven, 2009).

Although the study established that there was a significant gain in learning when ICT was incorporated in the learning of mathematics, existing research on using integrated technologies for teaching mathematics has indicated the need to scrutinise the professional learning that complements the use of new technologies into teaching practice. Further acknowledged in the literature is the need for pedagogies that promote integrated technology mathematics education that requires an alteration in teachers' professional attitude taking into account the barriers of learning (Gueudet & Trouche, 2009: Ruthven, 2009).

5.7 Case study 2: Lesson in Linear functions

5.7.1 Introduction

The reason for the lesson again was to investigate on the impact of ICT pedagogy in linear functions. The relationship between dependent and independent variables in linear functions and the impact they have on each other is difficult to observe by students when learning with pen and paper hence the ICT lesson. The lessons were categorised into two lesson episodes, to answer the research questions whether, the use of an ICT has any impact on mathematics education?

To achieve this aim the lesson plan ensured the design of learning that introduced the most basic concepts to the complex. I also examined the findings of the general design of these lessons. The main aims of the lessons are as follows:

- Introduction lesson to linear functions using GeoGebra
- To sketch a linear graph when given the gradient and point of interception
- To construct a table of values for given linear functions and draw a linear graph

5.7.2 Task 1: Introduction lesson to linear functions using GeoGebra

The main objective of the lesson was to introduce students on how to use GeoGebra for graphing. The lesson was an introduction lesson, which introduces both the software GeoGebra and the topic of graphing. The introductory remarks clearly express why the employment of the chosen graphing software is important instead of using paper. Remarkably, in the first minute, I expressed where a student can get the software to be used at a personal level. At the start of the lesson, students were asked to open the software after explaining and showing the class the location of the Geogebra.org software on the school's system. I further demonstrated to the class how to use the functions on the software and what they are for. I further explained how to access the graph functions in particular and then went ahead to explain how a learner can access the different features to manipulate the drawn graph, such as the use of cursor to access different points of the graph and to show a magnifier to increase the visibility. To understand the right level to pitch the lesson, I used a range of strategies, including open-ended questions to check with pupils' prior understanding in functions. After a group practice on the initial application on the use of the software, an independent practice session was given to the learners so that they could execute the skills learnt on their own. The task provided took account of what students already knew about linear functions. It also involved the assessment of the ability of the students to understand and use the basic functions of the GeoGebra software to draw graphs.

5.7.3 Task 2: To sketch a linear graph when given the gradient and point of interception

The objective of the practice exercise was to understand how to graph a line function using GeoGebra. Most students were able to use coordinates to sketch a linear graph with no difficulties. However, few students had difficulties when they were presented with only the gradient and asked to draw a linear function. After dialoguing with the students and inviting feedback from completed students, all the students were now in a position to complete the assigned task. Asking students to use GeoGebra to draw a function given the point of intersection was again not answered correctly by most students because most students failed to visualise that although the gradient of functions can be different, the point of intersection and the gradient are used to find the equation of the function. However, with the gradient and the y intercept, most students were able to overcome their difficulties and were able to draw the line with no difficulties.

5.7.4 Task 3: To construct a table of values for given linear functions and draw a linear graph

Throughout the lesson, the use of the ICT supported pupils in understanding the mathematical experimentation in line graphs in the form y = x + c. The use of the software provides encouragement to students learning through trail and improvement, which played a critical role in the understanding of the topic. Also, most students were able to engage with the mathematical speculations and experiment to enhance on their knowledge in the subject.

Immediately past this point, the students were introduced to the following linear equations:

$$Y=2x-7 \text{ and } Y=-x+5.$$

Past this point, the class was introduced to how to use the graphical feature to plot these two equations. Prior to introducing other actions and equations, I assisted the students to erase a graph when a mistake was made. Using prompts, I used examples to explain to the students how the gradient of the linear functions can be determined. I then gave the students the option to create problems and try to solve them with the software.

5.7.5 Lesson report

Most students were able to construct a table of values given the linear functions. Few students were able to link the skills gained in completing this task to a similar task in Excel. The transfer of knowledge in completing this task was not obvious because of the lack of understanding of using GeoGebra to draw a table. Most students were able to draw the linear graph given the function.

Revisiting a lesson with the aim of improving future lessons is crucial for an instructor. More so, when teaching resources are involved, reflection is vital for ensuring that the teacher employs the resources in a more effective manner. The use of technology in linear functions includes instructing and picking up, managing new types of dynamic portrayal and correspondence. However, it is likewise certain that the requirement for suitable expert advancement to help educators in planning innovation upheld exercises remains fundamental. Exclusively giving innovation is lacking to the effective joining of new unique devices to instructing.

As Olsson (2017) reports, giving access to learning software do not really prompt innovation. However, there is confirmation that proper expert development of ICT and collegial help can help instructors' readiness to incorporate innovation into their educating and can boost their ability to create fruitful innovation (Öçal, 2017). Some portion of this may involve helping educators in understanding the affordances, limitations, and the general instructive nature of new authentic assets in connection to the particular points in school mathematics education (Mata-Pereira & Ponte, 2017).

5.7.6 Pupil's response

The analysis on the semi-structured interviews with students, on the using of dynamic software (GeoGebra) to teach linear functions, revealed that learners were profoundly motivated to learn. Students (S1 and S4, Appendix 7) reiterated that using the GeoGebra tool for learning linear functions made their learning fun and exciting. Students (S2, S3 and S5, Appendix 7) supported the initial views on motivation and commented that using GeoGebra provided them with the necessary encouragement to keep on working through the process of trial and improvement even when their answers were wrong. Student S3 said, 'I would have quit trying for the right answer if it was investigation using pencil and paper due to cancelling my work and starting on a new page'. The ability when using GeoGebra to provide learners with a clean page to work on when mistakes are done is an incentive to encourage the completion of more work.

Additionally, as indicated by the interviews, most students reported that using GeoGebra for learning was reasonable and simple as far as drawing geometrical shapes. Most students commented that I was able to support their learning through the timely feedback presentation on the completed work. Also, most students were quick to comment on the sequence of the lessons with the differentiated work. According to students (S2 and S5, Appendix 7), the lesson was set at the right pace, which enabled them to remain engaged in the learning. The visualisation and the capability to use the multiple representation views function in GeoGebra were commented on by most students as the key feature that enabled understanding to take place.

5.7.7 Impact on learning functions with ICT on mathematics

The students reported that GeoGebra software had a positive impact on their mathematical performance. In spite of the fact that the post-execution test scores demonstrate abatement

contrasted with the pre-accomplishment test, the study demonstrates that there are noteworthy contrasts between the post-execution tests on utilising GeoGebra. The study demonstrates that the utilisation of GeoGebra in the learning can give a significant effect in enhancing the learner's ability. Similar to a study done by Masri et al., (2017), the consequences of this investigation additionally demonstrate that the utilisation of GeoGebra in the learning and teaching has the ability to improve learners understanding and mathematical reasoning capacity. In the same study, it was reported that students who used GeoGebra scored higher compared to those who learned with customary strategies. In view of these findings, GeoGebra is extremely useful in the conventional classroom and more powerful than customary instructing. The increase in learners' accomplishments' tests scores is likely because of variables of their fascination towards the dynamic technology. The advancement of innovation instruments expands students' enthusiasm to make sense of the new thing. The learners tend to investigate the technology to apply in learning mathematics.

The contrast in the performance of learners using GeoGebra and those not using it demonstrates that utilising GeoGebra has a positive consequence on the student's accomplishment in mathematics education. A review by Leung (2017) likewise demonstrates that learners using GeoGebra outperformed those who did not use it. In light of their observations, this proposes that GeoGebra is exceptionally useful in the customary classroom educating scenario and more powerful than conventional instructing. The augmentation in the students' accomplishments' tests scores are likely because of variables of their fascination towards the innovation.

Also, similar to the findings of Jacinto and Carreira (2017), the learners' recognition was distinguished through an arrangement of a survey comprising nine things. The outcomes picked up from the survey demonstrate positive outcomes. The investigation found that the things in the poll that had the most reduced mean expressed that students can think innovatively and basically with a mean of 3.93, while the most elevated mean is 4.62, which is received for the first time demonstrates that the general pool of students concurred with positive explanations when ICT was used especially, GeoGebra.

The outcome of the study presumes that the utilising of GeoGebra in learning linear functions contributes to an advantage to the learners' certainty and their inspiration in learning mathematics. Hegedus et al., (2017) expressed that the utilisation of ICT in instructing and learning is not just to enhance understudy execution, but also inspiration, and stated that

learners who reacted emphatically demonstrate the most elevated rate contrasted with different reactions. According to Hegedus et al., (2017), learners who used ICT for learning demonstrates enthusiasm for learning mathematics. In a similar research by Granberg and Olsson (2015), indicates that when ICT is used students gain comparable understanding through exploratory learning. It demonstrates that learning with dynamic technology could likewise trigger on-educational curiosity for learning mathematics.

To teach a lesson in linear function, the teacher's planning aimed at finding the resources to equip students with the skills of being able to identify the links between algebra symbols, tabular and graphical representation. In the planning to equip students, resources were designed to make students understand the different meaning of algebraic symbols. Undoubtedly, teaching linear functions is a complex procedure in which teachers plan to use diverse activities to help students to figure out the links that connect verbal, tabular, graphical and symbolic exemplifications. Using digital technology to help teacher establish this relationship is remarkable in mathematics education, even though the process of incorporating technology remains a complex and sensitive issue (Cuban, et al., 2001). The use of GeoGebra assists teachers to provide the clear impact of linear functions on graphs, visually for learners to think mathematically and appreciate the association between algebra and symbols.

However, it remains a controversial issue as to what role the integration of technology should contribute to students' mathematical reasoning and to what extent technology can enhance a teacher's mathematical pedagogy. Convincing enough, the use of the software made it easy to illustrate to students through dragging. Although it is possible for teachers to deliver lessons to make students understand algebraic functions without the use of technology, the presence of numerous challenges impeding students understanding is well known. The absence of technology makes it difficult to demonstrate the key concept of expressing relations, generalisations and formulae (Bell, 1996). Being aware of the student difficulties help in the planning of lessons that rely on technology to act as a medium to support students learning.

As commented on by many reviews, the learning of mathematics is centred on algebra and symbols rather than visuals. Hence, the use of technology has the ability to aid the visualisation that adds to the understanding of the subject. However, added to this benefit is the ability for students to understand the unique meaning attached to the symbols in functions when technology is used. In a digital environment, to generate a linear function, students are

able to enter a formula and instantly produce a graph and a table. All three representations can be seen as a goal enhancing learners' mathematical experience. Through dragging, algebraic values in functions values in functions are changed, which enables teachers to be able to show learners the role of each symbol (Noss, 1997).

5.7.8 Conclusion

In most cases, the use of ICT allows students to own their learning without being greatly dependent on teachers, giving a feeling of accomplishment and excitement in having control and ownership of their own projects. The use ICT has the ability for teachers to enhance on learners understanding not usually possible through the traditional way of teaching. The use of ICT allowed students to critically solve the mathematical problems though exploration and discovery learning. With the appropriate design tools and tasks, the students felt the urge in seeking for deeper creativity to enhance their understanding in mathematics.

5.8. Research question 1: What are the benefits of ICT in Mathematics education?

This session examines the data generated through both the questionnaires and the semistructured interviews on the views of teachers on the benefits of using of ICT in mathematics education.

5.8.1 Sustained student interest

The data analysis of the questionnaires indicated that although 85% of the teachers either strongly agreed or agreed, 15% either disagreed or strongly disagreed with the use of ICT to sustain student's interest in lessons. Also, the results of the interviews support the literature review finding that the benefits of ICT include sustained student interest (Al-Ansari, 2006; T1, Appendix 15). Ager (2000) similarly describes the benefit of ICT in motivating children through integrative applications. This agrees with the perceptions of Young (2014) and Phillips and Scrimshaw (1997), who indicate that using ICT tools promotes visual learning and enhanced excitement in learning. The belief that ICT can promote the advancement of talented pupils (OECD, 2011) was held by most teachers, who agreed that the use of ICT helps widen learner's accessibility to more educational resources to promote student creativity towards learning (Brown, 2011). The use of ICT has an added benefit of promoting excellent behaviour for learning since most students through the excitement of working with computer will remain fully engaged with their classwork(T4, Appendix 18)..

5.8.2 Aid lesson planning with wide range of resources

The data analysis of the questionnaires indicated that although 95% of the teachers either strongly agreed or agreed, 5% either disagreed or strongly disagreed with the use of ICT to aid lesson planning with wide range of resources. In addition, it was widely accepted by the teachers that using ICT allows them to plan lessons more easily and effectively (T4, Appendix 18). The teachers further agreed that the ability to use ICT to aid visual learners (T3, Appendix 17) ensures that teachers are able to plan their lessons to accommodate different learning needs of learners (T4, Appendix 18).

5.8.3 Improved the attainment of children

The data analysis of the questionnaires indicated that although 85% of the teachers either strongly agreed or agreed, 15% either disagreed or strongly disagreed with the use of ICT to improve the attainment of children Furthermore, in the interviews, the teachers indicated that integrated ICT lessons improved the attainment of children (T5, Appendix 19), confirming the views held by Ertmer et al., (2012). However, Wittwer and Senkbeil (2008) cited in Danhui and Luman (2016) argue that ICT has no direct impact on performance. This claim is questionable in light of the teachers' comments. Apart from the higher attainment perceived by teachers, a large body of empirical evidence accepts that teaching is improved when ICT is integrated into lessons (Abbott, 2001). The tension raised in the 'Think Again' Report indicates that the 'increase use of ICT ' will increase the gap between high and low school achievements' in shape contrast to what is already known and speculated by the teachers.

5.8.4 ICT enhance interactivity and exploration

The data analysis of the questionnaires indicated that although 95% of the teachers either strongly agreed or agreed, 5% either disagreed or strongly disagreed with the use of ICT enhances learning through interactivity and exploration practices. An additional benefit of integrating ICT is to enhance interactivity in lessons (Phillips & Scrimshaw, 1997). This view was reinforced by most of the teachers (T2, Appendix 16; T5, Appendix 19). According to T3, lessons taught using ICT promote 'interactivity and instant feedback to class work' (T3, Appendix 17). In addition, most teachers agreed with Gardner's (1993) views on using ICT to promote multiple intelligences.

5.9 Research question 2: What are the barriers to ICT in mathematics educations?

This session examines the data generated through both the questionnaires and the semistructured interviews on the views of teachers on the barriers of using ICT in mathematics education. The barriers to ICT integration include the lack of curriculum resources, hardware problems, and teachers' negative attitudes due to lack of experience and support.

5.9.1 Lack of curriculum resources

The data analysis of the questionnaires indicated that although 86% of the teachers either strongly agreed or agreed, 14% either disagreed or strongly disagreed that the lack of curriculum resources was a barriers to integrating ICT pedagogy in mathematics educations. In addition, T3 (Appendix 17) indicated that the absence of ICT curriculum resources is an obstacle to using ICT, as pointed out by Abbott (2001) in the literature review. The need for teachers to find suitable resources toiled to the aims of lessons makes it impossible to frequently integrate ICT pedagogy. School teachers 'not having the skills to develop the right resources and the lack of skills to use them were identified as an obstacle to ICT integration (T5, Appendix 19).

5.9.2 Software and hardware problems

The data analysis of the questionnaires indicated that 100% of the teachers either strongly agreed or agreed that the lack of software and hardware were a barrier to integrating ICT pedagogy in mathematics educations. Commenting on this source of frustration, Johnston-Wilder (2005) points out that although training workshops increase teachers' confidence in ICT integration, most teachers who are themselves class teachers complained about the lack of implementation due to software and hardware problems. T2 (Appendix 16), supported by Graham (1993), indicated that the cost and reliable ICT tools are further problems, including issues related to both out of date software and obsolete hardware problems, poor internet connections and poor interactive white boards continue to discourage teachers from using ICT pedagogy. Added to this, teachers cited that schools must improve upon the ICT infrastructure through the acquisition of up to date hardware and software to resolve these problems.

5.9.3 Lack of experience with ICT tools

The data analysis of the questionnaires indicated that although 85% of the teachers either strongly agreed or agreed, 15% either disagreed or strongly disagreed that the lack of experience with the use of ICT tools was a barrier to integrating ICT pedagogy in mathematics educations. A further barrier is the lack of teachers' own expertise on the use of technology (T3, Appendix 17). The lack of the right up to date knowledge to use ICT tools is a frustration perceived by most teachers in the application of ICT-integrated lessons. The inability to find the required ongoing specialist support on using ICT application tools has made most teachers to develop a poor attitude towards the use of ICT in lessons.

Similarly, the review indicated that the lack of teachers, experience and interest are determinant factors to ICT integration. (T4, Appendix 18). It was widely reported that teachers are likely to use ICT when they had prior skills in using ICT devices (T5, Appendix 19). This agrees with the research conducted by Pelgrum (1993), which revealed that school teachers who are in favour of ICT integration out of school were enthuse to incorporate ICT in their pedagogy.

5.9.4 Negative attitudes

The data analysis of the questionnaires indicated that 100% of the teachers either strongly agreed or agreed that the teacher's negative attitudes were a barrier to integrating ICT pedagogy in mathematics educations. The implication for teachers is that they lack the attitude and the desire to promote ICT-integration (Mingaine, 2013). Most teachers agreed with the view that, time should be set aside for training staff with a clear policy on ICT integration.

5.10 Research question 3: What are the facilitators to ICT into mathematics education?

This session examines the data generated through both the questionnaires and the semistructured interviews on the views of teachers on the facilitators of using of ICT in mathematics education.

5.10.1 The availability of ICT tools

The data analysis of the questionnaires indicated that although 95% of the teachers either strongly agreed or agreed, 5% either disagreed or strongly disagreed that the up to date

provision of ICT tools will facilitate the integrating of ICT pedagogy in mathematics educations. In answering the above question, it was interesting to discover that most teachers frequently mentioned in the interview that the provision of visual and dynamic aids, including videos and films, interactive whiteboards, YouTube videos, and tablets as the main facilitators of integrating ICT into mathematics pedagogy (T1, Appendix 15).

5.10.2 Experience and knowledge of using ICT

The data analysis of the questionnaires indicated that although 95% of the teachers either strongly agreed or agreed, 5% either disagreed or strongly disagreed that the experience and persona knowledge of using ICT will facilitate the use of ICT pedagogy in mathematics educations. **Also,** the interview questions confirmed the perception that teachers own experience and personal knowledge of using ICT were the main facilitators of using ICT (T2, Appendix 16; T4, Appendix 18), a finding consistent with the literature review (Ellis & Loveless, 2001).

5.10.3 School leadership

The data analysis of the questionnaires indicated that although 90% of the teachers either strongly agreed or agreed, 10% either disagreed or strongly disagreed that the school leadership on ICT will facilitate the integrating of ICT pedagogy in mathematics educations. The teachers' responses from the interview also confirmed that the attitudes of school leaders play a major role in determining the extent of ICT involvement in their lessons (Fullan et al., 1988). The participants acknowledged the need for school leaders to have a clear ICT policy initiates in lessons and designated staff to induct teachers who lack the required ICT expertise (T5, Appendix 16). Many teachers commented on the existing school leaders' styles. This corresponds with the views of O'Reilly and Reed (2010), who claim that the application of ICT policies depends on the style of school leaders. Again Fay was of the view that a leader that adapts the transformational style in the implementation of any school policy will achieve good results (Fay, 1987). In addition, a leader who operates under the collegial approach will be able to influence teachers to adapt a school policy with confidence (Bush, 2011). Generally, most teachers stated the need for school leaders to offer teachers plenty of chances to experiment with new technologies and provide ongoing support to reflect expertise (Fullan, 2006) before any classroom implementation.

5.10.4 Opportunities for training

The data analysis of the questionnaires indicated that although 94% of the teachers either strongly agreed or agreed, 4% either disagreed or strongly disagreed that the opportunities for training will facilitate the integrating of ICT pedagogy in mathematics educations. School leaders must provide opportunities, incentives and workshops to motive the integration of ICT. The training can be provided either on-the-job or off-the-job. The benefit of training for teachers is that it can then change the attitudes of teachers (Dawson & Rakes, 2003). However, surprisingly, most teachers reported that the lack of appropriate professional development has resulted in a lack of confidence in the ICT skills provided whilst training as a teacher. This corresponds with the literature review finding that most teaching agencies offered typically too short courses with no follow-up.

5.11 Chapter Summary

The above evaluation on relationship between this research, questionnaires, case studies and the semi structured interview indicates that the findings of this study are fully supported by the findings of previous studies. Having no contradictions to previous research demonstrates that the findings of the study are not biased, but genuine. Therefore, from the discussion and evaluation of the findings, it can be stated that the impact of ICT in mathematics education is dependents on a number of factors, including the benefits, barriers, support, and facilitators of ICT. Other factors include the style of teachers and the available professional support to equip teachers with the required confidence in the development of ICT pedagogy. Teachers need a clear vision with ongoing training to improve their attitudes towards integrating ICT into mathematics pedagogy.

CHAPTER 6: CONCLUSION

6.1 Introduction

The aim of this study was to explore the question: 'What are the impacts on ICT in mathematics education? In order to understand these impacts, the study conducted a review of the existing literature in light of the available information obtained in interviews through primary research. To respond to the research question, the study examined two case studies on own practice, questionnaires and interviews of teachers. The following sub questions guided the entire research;

- What are the benefits of ICT in Mathematics education?
- What are the barriers to ICT in mathematics education?
- What factors will facilitate ICT in Mathematics education?

The deductive research approach and the descriptive design of the research enabled detailed content analysis and investigation of the research questions. The research used both the primary data as well as the secondary data set through a mixed methodology approach. In the literature review, the study identified that previous research has indicated several benefits that school teachers gained if ICT was included within their mathematics pedagogy. Some of the benefits included easy problem solving and interactive lessons. At the same time, the literature review indicated some barriers that may exist while including ICT in mathematics pedagogy. Some of the barriers included lack of ICT knowledge among teachers, lack of ICT hardware, and associated software problems.

6.2 Effect digital tool

An ICT dynamic software tool such as GeoGebra has features that allow quick and easy creation of mathematical objects presented as visual representations. GeoGebra software supports the collaboration and the creativity of mathematical thinking and reasoning. It provides the students with a problem-solving method and a shared working space and feedback of the respective actions on the space.

Students reported that they found the use of ICT tool helpful in testing ideas and concepts by formulating input and submitting according to a question hypothesis. In the lessons examined in this study, it is clear that as the learners interacted with GeoGebra, they benefited with the

understanding of the topics better. There is, however, a factor that affects the effectiveness of employing ICT in lessons. It is, yet, vital to realise that when students utilised ICT to work on their tasks they participated in creative reasoning and therefore were able to demonstrate a higher understanding in mathematics

6.3 TPACK Framework

While many researchers find in the TPACK model a worldwide model for prospectus design, I found it awkward, confusing and misleading to sort out into clear groups of TK, PK, CK, TPK, PCK, TCK and TPACK, to enable me advance with the lesson design. Mention can be made of the empirical study comprising 596 online teachers in the United States whose views were studied on the practicality of using TPACK (Archambault & Barnett, 2010). According to this report, just as I found it in my lesson design, it was difficult to verify knowledge into pedagogy, content and technology as three separate domains. The only obvious classification of knowledge was the domain that segregates itself from the others, technology.

I agree with the view that the isolation of knowledge into the TPACK framework is not obvious and not a theory that is possible to empirically verify. This is because throughout the review it was identified that TPACK has no reliant or autonomous defined variables since all available constituents are classified as identical with no clear conditional or causal inferred associations. The non-existence of any null-hypotheses to examine with no known evidence-based results to predict makes the concept even more confusing to justify.

TPACK was unsuccessful when used in many applied researches in the initial empirical research when knowledge was to be distinguished into the seven equally detached domains. Even most worryingly, however, is the inability of the model to propose either an aimdirected structure that is linked with the authorisation of learners in the usage of technology or the relationship around the impact of digital liberalisation routes on society. It is worth mentioning that TPACK needs to be established so that the optimistic impact credited to the intrinsic effects of the method not founded on the critical thoughts are certain to attain when teachers deliberate on the incorporation of dynamic technologies into their pedagogies.

Using the strengths of the Knowledge Quartet to complement TPACK was eye-opening. This framework assumes the view that ownership of knowledge enlightens pedagogical strategies and choices in a fundamental way with the key components being knowledge and

understanding of mathematics. The Knowledge Quartet offered a framework to review and analyse my lessons from a mathematics point of view. Using the framework assisted me to concentrate on my mathematics contents instead of the technology, which was accommodated through the TPACK framework.

6.4 Lesson learned

The lessons learnt through the use of the framework have paved the way for me to understand the needs of my learners better when ICT is used and has equipped me to plan sequences of lessons aimed at engaging my students. Using the Knowledge Quartet granted me the needed objective view on the plight of students' mathematical understanding when ICT software is integrated in the learning process. In this manner, my technique towards the pedagogy of the subject becomes modified after each practice, hence epitomising the ideas of Scott et al., (2003), who pointed out that such a practice is able to promote a conjectural loop existing between philosophy and practice. The insights gained on my own practice were valuable and I hope will be used as a catalyst to improve my mathematics subject knowledge.

6.5 Summary of findings

While discussing the relevance of ICT in mathematics pedagogy from the professional point of view, it was vital to use the responses of teachers and students to clearly understand that insights of ICT integration were important for teacher implementation, as opinions influence attitudes, and attitudes affect behavioural intentions (Polizzi, 2011). An ICT integrating learning in mathematics education is 'successful if it is accompanied by social and institutional change in educational settings' (Punie et al. 2006:37). Even though to support ICT practice in mathematics, institutional policies should be redefined to support ICT pedagogy, policy makers must also 'acknowledge that the impact of ICT is complex and uncertain' (wood, 2003; 2).

The study has revealed the planning of ICT lessons in mathematics must appreciate the importance of prior knowledge on ICT before any implementation. The difficulties held by most teachers could be resolved through ongoing developments to address these challenges. Notwithstanding, the number of conclusions that can be drawn concerning teachers' attitude towards the impact of ICT integration into mathematics pedagogy, my initial stance was that the confidence of teachers influences the level of ICT usage in the lesson. Admittedly, after

carefully analysing the facts, I am of the view that other factors including the initial impact on ICT use is contradictory in light of the empirical evidence backed by the primary data. The review revealed that teachers' perceptions on the impact are independent of their attitudes towards ICT integration. This refutes the opinions of most teachers interviewed, and I believe that there is a relationship between teachers' attitudes and the implementation of ICT use in lessons. Institutionalised resistance also affects 'subliminally and therefore without acknowledged intentionality' happening consciousness impacts on ICT use in schools (Somekh, 2008:116). However, given the appropriate provision of relevant technologies, attitudes and behaviour can change over time (UNESCO, 2003).

6.6 Limitation and difficulties

No doubt combining a mixed methodology approached in this study was very cumbersome to do. I must knowledge that all of the information gained from such a wide range of literature including questionnaires and interview responses cannot possibly be presented in a single research paper due to time constraints and resources. Many problems were encountered in embarking on this study. The greatest was the difficulty of securing appointments to conduct interviews with teachers. I found it very frustrating with the numerous interview appointment cancellations that I had to go through before being able to conduct this research. Added to this was the huge amount of time required to transcribe the interviews from the tapes. Also I am uncertain that I was able to capture every detail as communicated by the participants.

Notwithstanding, designing the lesson was compounded with many technological problems. Among the problems were the lack of up to date computers and the latest version of the GeoGebra software. This problem created difficulties in the delivery of the lesson through loss of teaching time. Added to this was the unavailability of qualified staff to support students when faced with software difficulties. I believe that when these and many barriers of ICT lessons already mentioned are resolved teachers will be motivated to integrate technology in the teaching of mathematics.

6.7 Recommendations for further research

Finally, I have found this experience to be worthwhile, and the research experience has made me appreciate empirical inquiry. It has also taught me valuable critical thinking skills and enabled me to meet the skills required by the job market. As a teacher, I felt honoured to research and contribute to a topic that is known to have caused controversies within mathematical educational. This satisfaction motivates me to continue conducting research into interesting topics in whole school education such as "ICT across the whole school". Through this, I will gain the needed confident to lead the whole school initiative on ICT learning.

Bibliography

Abbott, C. (2001). ICT: Changing Education. London: RoutledgeFalmer.

- Abboud-Blanchard, M. (2014). Teachers and technologies: Shared constraints, common responses. In A. Clark-Wilson, O. Robutti, & N. Sinclair (Eds.), *The mathematics teacher in the digital era* (pp. 297–317). Dordrecht: Springer.
- Ager, R. (2000). The Art of Information and Communications Technology for Teachers. London: David Fulton.
- Aikenhead, G.S. (1996). Science education: Border crossing into the subculture of science. *Studies in Science Education*, 27, 1–52.
- Ainscow, M. (2015). Towards Self-Improving School Systems: Lessons from a City Challenge. London: Routledge.
- Al-Ansari, H. (2006). Internet use by the faculty members of Kuwait University. The Electronic Library Vol.24, No. (6), Pp; 791-803.
- Albugami, S.S. and Ahmed, V (2015). 'Success factors for ICT implementation in Saudi secondary schools : from the perspective of ICT directors, head teachers, teachers and students', International Journal of Education and Development using Information and Communication Technology, **11** (1), pp. 36-54.
- Allen, R. and West, A. (2011). Why do faith secondary schools have advantaged intakes? the relative importance of neighborhood characteristics, social background and religious identification amongst parents. British Educational Research Journal. 37(4) pp.691-712.
- Altschuler, B. M. and Altschuler, E. L., (2016). *Proof of the Irrationality of the square root of two in Babylonian geometry problem tablets. arXiv preprint arXiv:1603.06656.*
- Anthony, G. and Walshaw, M. (2007). Effective Pedagogy in Pāngarau/Mathematics: Best Evidence Synthesis Iteration (BES) Available at: <u>http://www.educationcounts.govt.nz/__data/assets/pdf_file/0007/7693/BES_Maths0</u> 7_Complete.pdf (Accessed: 20/12/2019)
- Archambault, L., & Crippen, K. (2009). Examining TPACK among K-12 Online Distance Educators in the United States. *Contemporary Issues in Technology and Teacher Education*, 9, 71-88. <u>http://www.citejournal.org/vol9/iss1/general/article2.cfm</u>

- Archambault, L.H. and Barnett, J.H. (2010). Revisiting pedagogical content knowledge: Exploring the TPACK framework. *Computers & Education*, *55*, pp.1656–1662.
- Artigue, M. (2002). Learning Mathematics in a CAS Environment: The Genesis of a Reflection about Instrumentation and the Dialectics between Technical and Conceptual Work. International Journal of Computers for Mathematical Learning, 7, 245-274.
- Askew, M. (2010). Issues in teaching Numeracy in Schools. Maidenhead: Open University Press
- Askew, M. and Wiliam, D. (1995). *Recent Research in Mathematics Education 5-16. Ofsted Reviews Of Research.* London: HMSO
- Askew, M., Brown, M., Rhodes, V., William, D. and Johnson, D. (1997). *Effective Teachers* of Numeracy. London: King's College, University of London.
- Askew, M., Hodgen, J., Hossain, S. and Bretscher, N. (2010). Values and Variables: Mathematics education in high performing countries. London: Nuffield Foundation;
- Aspers, P. (2004). Markets in Fashion, A Phenomenological Approach, London: Routledge. (First edition 2001, City University Press).
- Austin, J. H., (2009). Selfless insight: Zen and the meditative transformation of consciousness. Cambridge MA: The MIT Press.
- Baki, A., Kosa, T. and Guven, B. (2011). A comparative study of the effects of using dynamic geometry software and physical manipulatives on the spatial visualisation skills of pre-service mathematics teachers. *British Journal of Educational Technology*, 42(2), pp.291-310.
- Ball, D.L. (1991). Research on teaching mathematics: Making subject matter knowledge part of the equation. In: J. Brophy, ed. 1991. Advances in research on teaching: Vol 2. Teacher's knowledge of subject matter as it relates to their teaching practice. Greenwich, CT: JAI Press.
- Ball, S. J. (1990). Politics and policy-making in education. London: Routledge .
- Bandura, A. (1995). *Self-efficacy in changing societies*. Cambridge: Cambridge University Press.
- Banford, H., Berkman, M., Chin, C., Cziko, C., Fecho, B., Jumpp, D., (1996). *Cityscapes: eight views from the urban classroom*. Berkeley, CA: National Writing Project.

- Barak, M. (2007). Transition from traditional to ICT-enhanced learning environments in undergraduate chemistry courses. *Computers & Education*, 48(1), pp.30-43.
- Barker, B., Bird, R. and Mercer, J. (2010). Human Resource Management in Education: Contexts, Themes and Impact. London: Routledge.
- Bateson, G. (1979). Mind and Nature: a necessary unity. New York: E. P. Dutton, Steps to an Ecology of Mind: Collected Essays in Anthropology, Psychiatry, Evolution, and Epistemology. San Francisco: Chandler Publishing Company, 1972.
- Becker, G. S. (1978). *The Economic approach to human behaviour*. The University of Chicago Press.
- Becker, H. J. (2000). Findings from the teaching, learning, and computing survey: Is Larry Cuban right? [PDF file]. Centre for Research on Information Technology and Organizations (accessed: 02/04/2017, from <u>http://www.crito.uci.edu/tlc</u>.
- Becker, H.J., Ravitz, J.L. and Wong, Y.T. (1999). Teacher and teacher-directed student use of computers and software. Technical Report #3: Teaching, learning and computing, 1998 national survey. Irvine: University of California at Irvine.
- Becta (2004). http://dera.ioe.ac.uk/1603/1/becta_2004_barrierstouptake_litrev.pdf (Accessed: 10/03/2019).

Becta

(2006).

http://dera.ioe.ac.uk/1584/7/becta_2006_icttestbed_annualreport_report_Redacted.p df (Accessed: 10/03/2019).

- Becta ICT Research (2003). What the Research Says about Using ICT in Maths. British Educational Communications and Technology Agency. http://webarchive.nationalarchives.gov.uk/20130401151715/ https://www.education.gov.uk/publications/eOrderingDownload/15014MIG2799.pdf
- Bell, A. (1996). Problem solving approaches to algebra: *Two aspects. Perspectives for research and teaching* (pp.167–187). Netherlands Kluwer Publishers.
- Belli, G. (2008). No experimental Quantitative Research. Analysis and Interpretation in Nonexperimental Studies

Bernard, H.R. (2011). Research Methods in Anthropology 5th edition, AltaMira Press

- Bernard, H.R. and Bernard, H.R. (2012). Social research methods: *Qualitative and quantitative approaches*. Sage.
- Bester, G. and Brand, L. (2013). The effect of technology on learner attention and achievement in the classroom. *South African Journal of Education*, *33*(2) pp.1–35.

- Betz, N. E. and Fassinger, R.E. (2011). Research methodologies. In E.M. Altmaier & J.C. Hansen (Eds.), The Oxford handbook of counseling psychology (p. 237-269). New York: Oxford University Press.
- Bhattacharya, I. and Sharma, K. (2007). India in the knowledge economy an electronic paradigm, International Journal of Educational Management Vol. 21 No. 6, Pp. 543-568.
- Bindé, J. (2005). Towards knowledge societies UNESCO world report. Paris: UNESCO.
- Birgin, O., Gurbuz, R. and Catlioglu, H. (2012). Determining eighth grade students' understandings and difficulties of linear functions. *Energy Education Science and Technology Part B*, 4(3), pp.1345–1354.
- Blum, W. and Niss, M. (1991). Applied mathematical problem solving, modelling, applications, and links to other subjects state, trends and issues in mathematics instruction. *Educational Studies in Mathematics*, 22(1), pp.37–68.
- Blumberg, B.F., Cooper, D.R. and Schindler, P.S. (2014) Business research methods.McGraw-hill education.
- Bochner, S. (2014). *Role of mathematics in the rise of science*. Princeton, NJ: Princeton University Press.
- Bogdan,R.C. and Biklen, S.K.(1992). Qualitative research for Education: An Introduction to Theory and Methods. 2nd ed. Boston. MA: Allyn &Bacon.
- Borwein, J. M. (2005). The experimental mathematician: The pleasure of discovery and the role of proof. *International Journal of Computers for Mathematical Learning*, 10, pp.75–108.
- Bosch, M., Chevallard, Y. and Gascón, J. (2006). Science or magic? The use of models and theories in didactics of mathematics. In: M. Bosch, ed., *Proceedings of the Fourth Congress of the European Society for Research in Mathematics Education* (pp.1254–1263). Barcelona: Fundemi IQS – Universitat.
- Boutry, P., Braun, G. and Narboux, J., (2018). Formalization of the arithmetization of Euclidean plane geometry and applications. *Journal of Symbolic Computation*.
- Bozinovski, S. and Božinovski, A., (2017). Artificial Intelligence and infinity: Infinite series generated by Turing machines, *SoutheastCon*, 2017 (pp. 1-2). IEEE.
- Bozkurt, G., and Ruthven, K. (2017). Classroom-based professional expertise: a mathematics teacher's practice with technology. *Educational Studies in Mathematics*, *94*(3), pp.309–328.

- Bramall, S. and White, J. (2000). *Why Learn Maths?* London: Institute of Education, University of London.
- Bray, A. and Tangney, B. (2017). Technology usage in mathematics education research–A systematic review of recent trends. *Computers & Education*, *114*, pp.255–273.
- Brennan, K., and Resnick, M. (2012). <u>New frameworks for studying and assessing the</u> <u>development of computational thinking</u>. *Proceedings of the American Educational Research Association (AERA) annual conference.*
- Brinkmann, S., Kvale, S. (2015). *Interviews: Learning the craft of qualitative research interviewing*. 3rd edn. Thousand Oaks, CA:Sage and Kvale.
- British Educational Communications and Technology Agency (Becta) (2001). Primary Schools of the Future achieving today. Coventry: Becta.
- British Educational Research Association (BERA) (2011). Ethical guidelines for educational research. Available at: <u>https://www.bera.ac.uk/researchers-</u> <u>resources/publications/ethical-guidelines-for-educational-research-2011</u> (Accessed: 17/8/ 2019).
- Brophy, S. (2015). *What is pedagogical innovation?* [online] transformative LEARNING. Available at: http://stevebrophy.com.au/2015/06/18/what-is-pedagogical-innovation/ (Accessed 7/7/2019).
- Brown, M. (2010). 'Swings of the Pendulum', in I. Thompson (ed.) *Issues in Teaching Numeracy in Primary Schools.* 2nd edition, Maidenhead: OUP
- Brown, M. (2011) Going Back Or Going Forward? Tensions in the Formulation of a New National Curriculum in Mathematics.
- Brown, M., Millett, A., Bibby, T. and Johnson, D. C. (2000). 'Turning our attention from the what to the how: the National Numeracy Strategy', *British Educational Research Journal*, 26(4), pp. 457 – 471.
- Brown, S. and McIntyre, D. (1993). *Making sense of teaching*. Buckingham: Open University Press.
- Brown, S. I. (1995). Philosophy of mathematics education: POM(E), PO(ME) OR
 POE(M)?, Philosophy of mathematics education newsletter, No. 8 (May 1995): 16–
 18. Retrieved from
 http://socialsciences.exeter.ac.uk/education/research/centres/stem/publications/pmej/

- Brown, T., McNamara, O., Jones, L. and Hanley, U. (1999). Primary student teachers' understanding of mathematics and its teaching. *British Education Research Journal*, 25(3), pp.299–322.
- Bryman, A. (2004). Qualitative research on leadership: A critical but appreciative review. The Leadership Quarterly 15(6), 729-769.
- Buabeng-Andoh, C. (2012). Factors influencing subject leaders' adoption and integration of information and communication technology into teaching: A review of the literature. International Journal of Education and Development using Information and Communication Technology, 8(1), p.136.
- Burke, B.J, Anderson, J.J, Baptista, A.M. (2014). Evidence for multiple navigational sensory capabilities by Chinook salmon. Aquatic Biology.
- Burns, N. and Grove, S. K. (2003). Understanding nursing research. L4 ed. Philadelphia: Saunders Company.
- Burrill, G., Allison, J., Breaux, G., Kastberg, S., Leatham, K., and Sanchez, W. (eds.)
 (2002). Handheld graphing technology in secondary mathematics: research findings and implications for classroom practice. Dallas, TX: Texas Instruments.
- Bush, T. (2003) Theories of Educational Leadership and Management. (3rd ed.) London: Sage.
- Bush, T. (2008). Leadership and Management Development in Education (Education leadership for social change). London: Sage Publications Ltd, United Kingdom.
- Bush, T. (2011). Theories of Educational Leadership and Management (4 ed.) London: SAGE.
- Bush, T. and Crawford, M. (2012). Mapping the field over 40 years: A historical review
- Bush, T. and Middlewood, D. (2005). Leading and Managing People in Education (L1ed.) London: SAGE.
- Cabero, L. (2001). Oversold and underused: computers in the classroom. Cambridge, MA: Harvard University Press.
- CadwalladerOlsker, T. (2011). "What Do We Mean by Mathematical Proof?," Journal of Humanistic Mathematics, Volume 1 Issue 1 (January 2011), pages 33-60. DOI: 10.5642/jhummath.201101.04 . Available at: http://scholarship.claremont.edu/jhm/vol1/iss1/4, Accessed, 15/07/2018
- Caldwell, B. and Spinks, J. (2013) The Self-Transforming School. London: Routledge

- Capel, S., Breckon, P. and O'Neill, J. (2006). *A practical guide to teaching physical education in the secondary school*. London: Routledge.
- Cauthen, L. (2013). What's Wrong with SAMR in Education? [online]
 Thelearningcounsel.com. Available at:
 http://thelearningcounsel.com/archives/Wrong-With-SAMR-Education (Accessed 7/7/2019).
- Centre, B. (2015). Research Methodology: Tools and Techniques, Bridge Centre Buzzau: European Union Printed in Romania
- Chambers, P., (1999). Teaching Pythagoras' Theorem. *Mathematics in School*, 28(4), pp. 22-24. [online] Available at: <u>http://www.jstor.org/stable/30212035</u>. Accessed: 20/06/2019
- Chimuka, A. (2017). The effect of integration of GeoGebra software in teaching of circle geometry on grade 11 students' achievements, University of South Africa.
- Chitty C. (2014). *The Evolving Curriculum in Education Policy in Britain Hampshire* Palgrave: Macmillan
- Chitty C. (2014). Education Policy in Britain (3rd ed) Hampshire: Palgrave Macmillan
- Christodoulou, D. (2014). Seven myths about education. Routledge: London.
- Clark-Wilson, A., Robutti, O., and Sinclair, N. (Eds.). (2014). *The Mathematics Teacher in the Digital Era: an International Perspective on Technology Focused Professional Development*. Dordrecht: Springer.
- Clausen-May, T., Jones, K., McLean, A. and Rollands, S. (2000). Perspectives on the design of the geometry curriculum. *Proceedings of the British Society for Research into Learning Mathematics*, 20 (1&2), pp.34-41.
- Clements, D. H. and Conference Working Group (2004). Part one: major themes and recommendations. In: D. H. Clements, J. Sarama and A.-M. DiBiase, eds., *Engaging young children in mathematics: standards for early childhood mathematics education* (pp. 1-72). Mahwah: Erlbaum.
- Cochran, K.F., de Ruiter, J. A. and King, R. A. (1993). Pedagogical content knowledge: An Integrative model for teacher preparation Journal of Teacher Education 44 (4): 263 – 272
- Cohen, L., Manion, L. and Morrison, K. (2007). Research Methods in Education. Oxon: Routledge. Mixed Methods Research: A Research Paradigm Whose Time Has Come Author(s):

- Collarbone, P. and West-Burnham, J. (2008). Understanding Systems Leadership: Securing Excellence and Equity in Education. [London]; New York, NY: Network Continuum/Continuum International Pub. Group.
- Conti, G. J. (2006). Research in the Tribal Community. Two Research Paradigms. http://www.fdl.cc.mn.us/tcj/summer97/GC.html (Accessed 02/03/2019).
- Coolican, H. (1999). Research Methods and Statistics in Psychology. London Holder and Stoughton
- Corrigan, D, Dillon, J & Gunstone, R (eds) (2007), <u>*The Re-emergence of Values in Science</u></u> <u><i>Education*</u>. Sense Pubications, Rotterdam.</u>
- Cox, M. J, Rhodes, V. and Hall, J. (1988). The use of Computer Assisted Learning in primary schools: some factors affecting the uptake. Computers and Education Vol 12(1). pp. 173-178.
- Cox, M. J., and Webb, M. (Eds.). (2004). A review of the research literature relating to ICT and pedagogy. Coventry: Becta/London: DfES.
- Cox, M.J. (1993). Information Technology Resourcing and Use in Watson, D.M. (Ed.) Impact - An evaluation of the Impact of the Information Technology on Children's Achievements in Primary and Secondary Schools, King's College London. DfES: ICT in Schools: <u>www.dfee.gov.uk/ictinschools/</u>
- Cox, M.J., Preston, C. and Cox, K. (1999). What Motivates Teachers to use ICT?. Paper presented at the British Educational Research Association Conference. Brighton.
- Cox, S., and Graham, C. (2009). Diagramming TPACK in Practice: Using and Elaborate Model of the TPACK framework to Analyse and Depict Teacher Knowledge. TechTrends, 53(5), 60- 69.
- Creswell, J. W. (1998). Qualitative Inquiry and Research Design. Choosing Among Five Traditions, Sage Publications, London.
- Creswell, J. W. (2011). Educational Research: Planning, Conducting, and Evaluating Quantitative and Qualitative Research. (Fourth.). Pearson.
- Cronin, C. (2014). Using case study research as a rigorous form of inquiry. Nurse Researcher, 21(5), 19-27.
- Crook, C. (1994). Computers and the collaborative experience of learning London,: Routledge

- Cuban, L., Kirkpatrick, H. and Peck, C. (2001). High access and low use of technologies in high school classrooms: Explaining an apparent paradox. *American Educational Research Journal*, 38(4), pp.813–834.
- Cyr, J. (2016). The pitfalls and promise of focus groups as a data collection method. Sociological Methods & Research, 45(2), pp.231-259.
- Daft, R. L. (1999). Leadership, The Dryden Press, Fort Worth, TX.
- Danhui , Z. and Luman, L. (2016). How does ICT use Influence Students' Achievements in Math and Science Over Time? Evidence from PISA 2000 to 2012.
- Davies, B. and Ellison, L. (2003). The New Strategic Direction and Development of the School. London: RoutledgeFalmer.
- Davis, F. D. (1989) Perceived Usefulness, Perceived Ease of Use, and User Acceptance of Information Technology. MIS Quarterly 13(3): 318–339.
- Davis, F. D., Bagozzi, R.P., and Warshaw, P. R. (1989). User Acceptance of Computer Technology: a Comparison of Two Theoretical Models." Management Science 35(8): 982–1003.
- Davis, P. (1993). Visual Theorems. *Educational Studies in Mathematics*, 24(4), 333-344. Retrieved from http://www.jstor.org/stable/3482648
- Dawson, C. and Rakes, G. C. (2003) The Influence of Principals: Technology Training on the Integration of Technology into Schools." Journal of Research on Technology in Education 36(1): 29–49.
- Day, C., Sammons, P., Hopkins, D., Harris, A., Leithwood, K., Gu, Q. and Brown, E.(2010). 10 Strong Claims about Successful School Leadership, College for Leadership of Schools and Children's Services
- De Lemos, J., (1995). Reader reflections: The Pythagorean theorem. *Mathematics Teacher*, 88(1), pp. 79-80.
- Deetz, S. (1996). Describing differences in approaches to organization science: Rethinking Burrell and Morgan and their legacy, Organization Science, (7:2), 1996, pp. 191– 207.
- Denscombe, M. (2014). *The good research guide: for small-scale social research projects*. 5th edn. Maidenhead: Open University Press/McGraw-Hill.
- Denzin, N. and Lincoln, Y. (2008). The landscape of qualitative research. 3rd ed. Los Angeles: Sage Publications.

- DfE (Department for Education (2012a). *National Curriculum for England* (online) Available at www.education.gov.uk/ schools/teachingandlearning/curriculum (accessed 01.01.18.)
- DfE (Department for Education) (2012b). National Curriculum for England (2012). (online) Available at www.education.gov.uk/ schools/teachingandlearning/curriculum (accessed 01.01.18.)
- DfEE (1998). Survey of Information and Communication Technology in Schools 1998: Norwich: The Stationary Office.
- Diefenbach, T. and Sillince, J.A. (2011). Formal and informal hierarchy in different types of organization. Organization Studies, 32(11) pp.1515-1537.
- Dochy, F., Segers, M., Van den Bossche, P. and Gijbels, D. (2003). Effects of problembased learning: a meta-analysis. *Learning and Instruction*, 13, pp.533–568.
- Dockendorff, M. and Solar, H. (2018). ICT integration in mathematics initial teacher training and its impact on visualisation: the case of GeoGebra. *International Journal* of Mathematical Education in Science and Technology, 49(1), pp.66–84.
- Doerr, H. M. and Pratt, D. (2008). The learning of mathematics and mathematical modelling. In: M. K. Heid and G. W. Blume, eds., *Research on technology in the teaching and learning of mathematics*: Vol. 1. Research syntheses (pp.259–285). Charlotte, NC: Information Age.
- Donevska-Todorova, A. (2018). Fostering Students' Competencies in Linear Algebra with
 Digital Resources. In: *Challenges and Strategies in Teaching Linear Algebra* (pp. 261–276). Springer, Cham.
- Doyle, W., (1983). Academic work. Review of Educational Research, 53, pp. 159-199.
- Drijvers, P. (2013) 'Digital technology in mathematics education: why it works (or doesn't)', *PNA*, 8(1), pp. 1 20.
- Dunaway, M., K. (2011). "Connectivism: Learning theory and pedagogical practice for networked information landscapes", Reference Services Review, Vol. 39 Issue: 4, pp.675-685, https://doi.org/10.1108/00907321111186686.
- Dupagne, M., and Krendl, K. A. (1992). Teachers' attitudes toward computers: a review of the literature. Journal of Research on Computing in Education, 24(3), 420-429
- Duval, R., (2002). Proof understanding in mathematics: What ways for students? In: Proceedings of 2002 International Conference on Mathematics: Understanding Proving and Proving to Understand, pp. 61-77.

- Eckert, E., (1992). Primitive Pythagorean triples, *The College Mathematics Journal, Mathematical Association of America*, [e-journal] 23(5), pp. 413–417. Doi: 10.2307/2686417, JSTOR 2686417.
- Edwards, R., and Nicoll, K. (2006). Expertise, competence and reflection in the rhetoric of professional development. BERJ, 32(1), pp. 114-131.
- Eisenhart, M., Borko, H., Underhill, R., Brown, C., et al. (1993). Conceptual knowledge falls through the cracks: Complexities of learning to teach mathematics for understanding. *Journal for Research in Mathematics Education*, 24(1), pp.8–40.
- Ekanayake, S.Y., and Wishart, J., (2015). Integrating mobile phones into teaching and learning: A case study of teacher training through professional development workshops. British Journal of Educational Technology, 46(1), pp.173-189.
- Elliot, S. (1998). Visualisation and using technology in a level mathematics. In: T. Rowland and C. Morgan, *Proceedings of the BSRLM*, 18(2).
- Ellis, V. and Loveless, A. (2001). ICT, Pedagogy, and the Curriculum: Subject to Change. London: RoutledgeFalmer.
- Erickson, F. (2012). Qualitative Research Methods for Science Education. In: B. Fraser, K.Tobin & C. McRobbie (Eds.), Second International Handbook of Science Education.Springer International Handbooks of Education, vol 24. Dordrecht: Springer.
- Ernest, P. (1991). The philosophy of mathematics education. Basingstoke: Falmer Press.
- Ertmer, A. P. (2005). Educational Technology Research and Development, Volume 53, Issue 4, pp 25–39.
- Ertmer, P.A., Ottenbreit-Leftwich, A.T., Sadik, O., Sendurur, E. and Sendurur, P., (2012).Teacher beliefs and technology integration practices: A critical relationship. Computers & Education, 59(2), pp.423-435.
- Eshetu, G. (2015). Factors affecting instructional leaders perception towards educational media utilization in classroom teaching. Hamburg: Anchor Academic Publishing.
- ESRC (2010). Framework for Research Ethics Swindon: ESRC <<u>http://www.esrcsocietytoday.ac.uk/ESRCInfoCentre/Images/Framework%20for%2</u> <u>OResearch%20Ethics%202010_tcm6-35811.pdf</u>> (Accessed 16/08/ 2019)
- Evans, R. (1996). The Human Side of School Change: Reform, Resistance, and the Real-Life Problems of Innovation. San Francisco: Jossey-Bass.
- Ferrance, E. (2000). Action research. Providence, RI: LAB at Brown University.

- Finzer, B. and Bennett, D. (1995). From drawing to construction with the Geometer's Sketchpad. *Mathematics Teacher*, 88, pp.428–431.
- Flanagan, J.C. (1954). 'The critical incident technique', *Psychological Bulletin*, 51(4), pp. 327–58.
- Forsythe, S. and Jones, K. (2009). Tasks that support the development of geometric reasoning at KS3. *Proceedings of the British Society for Research into Learning Mathematics*, 29(3), pp.103–108.
- Fraenkel, J.R. and Wallen, N.E. (2008). *How to design and evaluate research in education*. (7th ed). Boston: McGraw-Hill.
- Fu, J.S. (2013). ICT in education: A critical literature review and its implications. International Journal of Education and Development using Information and Communication Technology, 9(1), p.112.
- Fujita, T. and Jones, K. (2009). The Place of Experimental tasks in Geometry teaching: learning from the textbook designs of the early 20th century. *Research in Mathematics Education*, 5:1, 47-62, DOI: <u>10.1080/14794800008520114</u>
- Fullan, M. (1993). Change Forces: Probing the Depth of Educational Reform, Falmer Press, London.
- Fullan, M. (2001). The New Meaning of Educational Change. New York: Teachers College Press.
- Fullan, M. (2006) Change theory A force for school improvement. Centre for Strategic Education, Victoria.
- Fullan, M. G., Matthew, M., and Stephen, E. A. (1988). Strategies for Implementing Microcomputers in Schools: The Ontario Case. Toronto, Ontario: Ministry of Education. Ontario Ministry of Education.
- Gal, H. and Linchevski, L., (2010). To see or not to see: Analyzing difficulties in geometry from the perspective of visual perception, *Educational Studies in Mathematics*, 74, pp. 163-183.
- Gardner, H. (1993). Frames of Mind: *The Theory of Multiple Intelligence*. London: Fontana Press.
- Garet, M. S., Porter, A. C., Desimone, L., Birman, B. F., and Yoon, K. S. (2001). What makes professional development effective? Results from a national sample of teachers.American Educational Research Journal, 38, 915–945.

- Gephart, R. (1999). Paradigms and Research Methods: http://division.aomonline.org/rm/1999.RMDForum_Paradigms-and-(Accessed 10 /6/ 2019).
- Gerlach, J. M. (1994). "Is this collaboration?" In Bosworth, K. and Hamilton, S. J. (Eds.), Collaborative Learning: Underlying Processes and Effective Techniques, New Directions for Teaching and Learning No. 59.
- Gerstein, J. (2014). SAMR as a Framework for Moving Towards Education 3.0. [online] User Generated Education. Available at: https://usergeneratededucation.wordpress.com/2014/02/23/samr-as-a-frameworkfor-moving-towards-education-3-0/ (Accessed 6/8/ 2019).
- Giaquinto, M., (2007). Visual thinking in mathematics: An epistemological study. Oxford, UK: Oxford University Press.
- Gillespie, H. (2006). Unlocking Learning and Teaching with ICT: Identifying and Overcoming Barriers. London: David Fulton.
- Gioia D. A., Thomas J. B. (1996). Identity, image, and issue interpretation: Sensemaking during strategic change in academia. Administrative Science Quarterly, 41, 370–403
- Glatter, R. (2006) Leadership and organisation in education; Time for a re-orientation?, School Leadership and Management, 26, 1, pp,69-83.
- Godwin, S. and Sutherland, R. (2004). Whole- class technology for learning mathematics: *the case of functions and graphs, Education, Communication & Information,* 4:1, 131-152, DOI: <u>10.1080/1463631042000210953</u>
- Godwin, S. and Sutherland, R., (2007). Whole- class technology for learning mathematics: the case of functions and graphs", Education, Communication & Information. Education, Communication & Information, 4(1), pp. 131-152.
- Goos, M. (2010). Using technology to support effective mathematics teaching and learning:What counts? Paper presented at *Teaching Mathematics? Make it count: What research tells us about effective teaching and learning of mathematics.*
- Goos, M., (2014). Technology Intergration in Secondary School Mathematics:The Development of Teacher's Professional Identities. In: A. Clark-Wilson, O. Robutti & N. Sinclair, eds. The Mathematics Teacher in the Digital Era. s.l.:Springer, pp. 139-161.

- Goos, M., Geiger, V. & Dole, S. (2014). Transforming professional practice in numeracy teaching. In Y. Li, E. Silver & S. Li (Eds.). Transforming mathematics instruction: *Multiple approaches and practices* (pp. 81–102). New York: Springer.
- Gordon, M., (2016). Developing Students' Mathematical Intelligence. In: *Enabling Students in Mathematics*. Springer, Cham. pp. 5-11.
- Gove , M. (2010). Foreword, in T. Oates (2010) Could do better: using international comparisons to refine the National Curriculum in England. Cambridge: Cambridge Assessment.
- Graham, C.R., Burgoyne, N., Cantrell, P., Smith, L., St. Clair, L. and Harris, R. (2009). TPACK development in science teaching: *Measuring the TPACK confidence of inservice science teachers. TechTrends*, p.70.
- Graham, D. (1993). A lesson for Us All: The Making of the National Curriculum. London Routledge
- Granberg, C. and Olsson, J. (2015). ICT-supported problem solving and collaborative creative reasoning: Exploring linear functions using dynamic mathematics software. *The Journal of Mathematical Behaviour*, 37, pp.48–62.
- Greenes, C. E., (2003). Navigating through problem solving and reasoning in prekindergarten-kindergarten. Reston, VA: National Council of Teachers of Mathematics.
- Gueudet, G. and Trouche, L. (2009). Towards new documentation systems for mathematics teachers? *Educational Studies in Mathematics*, 71(3), pp.199–218.
- Hadjidemetriou, C. and Williams, J., (2002). Children's Graphical Conceptions. Research in Mathematics Education, 4(1), pp. 69-87.
- Hadjidemetriou, C., & Williams, J. (2002). Teachers' Pedagogical Content Knowledge:
 Graphs, from a Cognitivist to a Situated Perspective. In A. D. Cockburn, & E. Nardi (Eds.), Proc. 26th Conf. of the Int. *Group for the Psychology of Mathematics Education* (Vol. 3, pp. 57-64). Norwich, UK: PME.
- Halcomb, E.J., Davidson, P.M. (2006). 'Is verbatim transcription of interview data always necessary?' *Applied Nursing Research*, 19, pp. 38–42.
- Halldórsson, A. M., McKelvie, P. & Björnsson, J. K., (2009). Are Icelandic boys really better on computerized tests than conventional ones? Interaction between gender, test modality and test performance. In: F. Scheuermann & J. Björnsson, eds. The
Transition to Computer-Based Assessment: New Approaches to Skills Assessment and Implications for Large-scale Testing. s.l.:European Communites, pp. 178-193.

- Hallet (2014). (electronic resource) Effective and caring leadership in the early years IramSiraj-Blatchford, Elaine.
- Hammersley-Fletcher, L. and Brundrett, M. (2005). Leaders on leadership: the impressions of primary school head teachers and subject leaders. School Leadership and Management, Vol.25, No.1, pp.59-75.
- Hanna, G. and Jahnke, H. N. (1996). Proof and Proving. In A. J. Bishop (Ed), International Handbook on Mathematics Education. Dordrecht: Kluwer.
- Harada, V. H., Kirio, C. and Yamamoto, S. (2008). Project based learning: rigor and relevance in high schools. *Library Media Connection*, 26(6), pp.14–16, 18, 20.
- Harman, G. (1984). Conceptual and theoretical issues. In J.R. Hough (ed.) Educational Policy: An International Survey. London: Croom Helm.
- Harris, K. (1995). 'Education for citizenship', in W. Kohli (ed.), Critical Conversations in Philosophy of Education. New York: Routledge.
- Harris, L.R. and Brown, G.T.L. (2010). Practical Assessment, Research & Evaluation, Mixing interview & questionnaire methods , Vol 15, No 1, From: <u>http://pareonline.net/pdf/v15n1.pdf</u> [Assessed: 17/08/2019]
- Harris, S. (2002). Innovative pedagogical practices using ICT in schools in England. Journal of Computer Assisted Learning, No. 18, Pp;449-458.
- Hatcher, R. (2006). 'Social class and schooling: Differentiation or democracy?' in Cole, M.
 (Ed.) Education, Equality and Human Rights: Issues of Gender, 'Race', Sexuality, Disability and Social Class. London: Routledge.
- Hegedus, S., Laborde, C., Brady, C., Dalton, S., Siller, H.S., Tabach, M. and Moreno-Armella, L. (2017). Uses of technology in upper secondary mathematics education. In Uses of Technology in Upper Secondary Mathematics Education (pp.1–36). Cham: Springer.
- Heid, M. K. and Blume, G. W. (Eds.). (2008). Research on technology in the teaching and learning of mathematics: Research syntheses (Vol. 1). Charlotte, NC: Information Age Publishing.
- Herbst, P. and Balacheff, N., (2009). Proving and knowing in public: The nature of proof in a classroom. In: M. Blanton, D. Stylianou, and E. Knuth (2008), eds. *Teaching and*

learning proof across the grades: A K-16 perspective. New York, NY: Routledge. pp. 40-63.

- Herr, K. and Anderson, G. (2015). *The action research dissertation: a guide for students and faculty*. 2nd edn. CA: Thousand Oaks: SAGE.
- Hersh, R. (1993). Proving is convincing and explaining. *Educ Stud Math* **24**, 389–399. <u>https://doi.org/10.1007/BF01273372</u>.
- Herzberg, F. (1968). One More Time: How do you Motivate Employees? Harvard Business School
- Hew, K. F., and Brush, T. (2007). Integrating Technology into K-12 Teaching and Learning: Current Knowledge Gaps and Recommendations for Future Research. Education Technology Research and Development, 55, 223-252.
- Hewitt, P., (2017). Newton's (Often Misunderstood) Third Law of Motion. *The Science Teacher; Washington* 84(2), pp. 12-14.
- Higgins, S., Hall, E., Wall, K., Woolner, P. and McCaughey, C. (2005). The Impact of School Environments: A literature review, Produced for the Design Council; The Centre for Learning and Teaching; University of Newcastle.
- Hobgood, B. and Ormsby, L. (2015). NC Learn: Inclusion in the 21st-century classroom: Differentiating with technology. [online] Available at: <u>http://www.learnnc.org/lp/editions/every-learner/6776</u> [Accessed: 27th July, 2018].
- Hoffman, S. (2017) Organization, Administration, Management, and Planning. Academic Librarianship Today, p.29.
- Hohenwarter, M., & Preiner, J. (2007). Dynamic mathematics with GeoGebra. Journal of Online Mathematics and its Applications. ID 1448, vol. 7, March 2007
- Holloway, I., and Wheeler, S. (2002) Qualitative Research in Nursing. Oxford, UK: Blackwell Science Ltd.
- Hoyles, C., and Lagrange, J.-B. (2010). *The 17th ICMI Study: Mathematics Education and Technology – Rethinking the Terrain*. New York, NY: Springer.
- Hubbard, R.S. and Power, B.M. (1999). *Living the questions: A guide for teacherresearchers*. Portland, Maine: Stenhouse.
- Husbands, C. (2015). Knowledge and the Curriculum. London, Policy Exchange.
- Hutapea, L. M., Suryadi, D. and Nuerlaelah, E., (2015). Analysis of students' epistemological obstacles on the subject of Pythagorean Theorem. *Jurnal*

Pengajaran Matematika dan Ilmu Pengetahuan Alam. 20. 1. 10.18269/jpmipa.v20i1.555.

- Hutchins, R. M. (1968). The Learning Society. New York: Frederick A. Praeger, Pub.
- Institute of Physics, (1990). *Physics in a mathematical mood* [online] Available through: https://www.stem.org.uk/resources/elibrary/resource/ 27003/physics-mathematicalmood – Particularly sections 1,2,5 and 12. Access: 20/07/2018
- Jacinto, H. and Carreira, S. (2017). Mathematical problem solving with technology: the techno-mathematical fluency of a student-with-GeoGebra. *International Journal of Science and Mathematics Education*, *15*(6), pp.1115–1136.
- John, P. D. (2006). Lesson planning and the student teacher: *re-thinking the dominant model. Journal of Curriculum Studies*, *38*(4), 483-498.
- Johnston-Wilder, S. and Pimm, D. (2005). Teaching Secondary Mathematics with ICT. Maidenhead: Open University Press.
- Jones, K. (2002). Issues in the Teaching and Learning of Geometry. In: Linda Haggarty, ed., Aspects of Teaching Secondary Mathematics. London: Routledge Falmer, pp.121-139.
- Jones, K. (2011). The value of learning geometry with ICT: Lessons from innovative educational research. In: A. Oldknow and C. Knights, eds., *Mathematics*. Education with Digital Technology, London: Continuum.
- Jones, K., Kavicza, Z., Hohenwarter, M., Lu, A., Dawes, M., Parish, A. and Borcherds, M., (2009). "Establishing a professional development network to support teachers using dynamic mathematics software GeoGebra", *Proceedings of the British Society for Research into Learning Mathematics*, 29(1), pp.97-102.
- Joy II, E. H. and Garcia, F. E. (2000). Measuring learning effectiveness: A new look at nosignificant-difference findings, Journal of Asynchronous Learning Networks 4(1).
- K. Popper, *The Logic of Scientific Discovery* (1959), reprinted (2004) by Routledge, Taylor & Francis
- Kaplan, B. and Maxwell, J.A. (1994). "Qualitative Research Methods for Evaluating Computer Information Systems," in Evaluating Health Care Information Systems: Methods and *Applications, J.G. Anderson, C.E. Aydin and S.J. Jay (eds.),* Sage, Thousand Oaks, CA, 1994, pp. 45-68.
- Kaplan, B. and Maxwell, J.A.(1994). "Qualitative Research Methods for Evaluating Computer Information Systems," in Evaluating Health Care Information Systems:

Methods and Applications, J.G. Anderson, C.E. Aydin and S.J. Jay (eds.), Sage, Thousand Oaks, CA, 1994, pp. 45-68.

- Keengwe, J. and Onchwari, G. (2008). Computer technology integration and student learning: Barriers and promise, Journal of Science Education and Technology, 17, 560–565.
- Kelly A. (2004). The Curriculum: Theory and Practice (5th edition) London: Sage
- Kember, D. (2008). Digital world, digital pedagogies: Reframing the learning landscape.Paper presented at the Australian Computers in Education Conference (ACEC), Canberra, Australia.
- Kennewell, S. (2000). The Role of Computer Modelling in the Teaching and Learning of Algebra to Pupils Aged 11-14. Unpublished Ph.D. thesis. Swansea: University of Wales Swansea.
- Kennewell, S., Parkinson, J. and Tanner, H. (2000). Developing the ICT-Capable School. London; New York: RoutledgeFalmer.
- Key Curriculum Press. (2003). The geometer's sketchpad [Computer software]. Retrieved December 23, 2017, from <u>www.keypress.com/sketchpad</u> Contemporary Issues in Technology and Teacher Education, 3(4)
- Knisely, K. (2007). No experimental Quantitative Research in Analysis and Interpretation in No experimental Studies, University of Texas at Brownville
- Koehler, M. and Mishra, P. (2005). What happens when teachers design educational technology? The development of technological pedagogical content knowledge. *Journal of Educational Computing Research*, 32(2), pp.131–152.
- Koehler, M. and Mishra, P. (2008). Introducing TPCK. In: AACTE Committee on Innovation and Technology, ed. 2008. *Handbook of technological pedagogical content knowledge (TPCK)*. New York, NY: Routledge.
- Koehler, M. J. and Mishra, P. (2009). What is technological pedagogical content knowledge? *Contemporary Issues in Technology and Teacher Education*, 9(1), pp.60-70.
- Koh, J.H.L. and Chai, C.S. (2014). Teacher clusters and their perceptions of technological pedagogical content knowledge (TPACK) development through ICT lesson design. Computers & Education, 70, pp.222-232.
- Kolb, D.A. (2015). *Experiential learning: experience as the source of learning and development*. 2nd edn. New Jersey: Pearson Education Inc.

- Kolpas, S. J., (1992). The Pythagorean Theorem: Eight Classic Proofs. Palo Alto, CA: Dale Seymour Publications.
- Kolpas, S. J., (2016). Mathematical Treasure: James A. Garfield's Proof of the Pythagorean Theorem, *Convergence* (February 2016)
- Kotter, J. P. (1990). A force for change: How leadership differs from management. New York, NY: Free Press.
- Krauss, S., Brunner, M., Kunter, M., Baumert, J., Blum, W., Neubrand, M. and Jordan, A. (2008). Pedagogical content knowledge and content knowledge of secondary mathematics teachers. *Journal of Educational Psychology*, *100*(3), pp.716–725. doi: 10.1037/0022-0663.100.3.716
- Kreijns, K., Van Acker, F., Vermeulen, M. and Van Buuren, H., (2013). What stimulates subject leaders to integrate ICT in their pedagogical practices? The use of digital learning materials in education. Computers in human behavior, 29(1), pp.217-225.
- Kuchler, J. M. (1998). The effectiveness of using computers to teach secondary school (grades 6-12) mathematics: a meta-analysis (Doctoral dissertation, University of Massachusetts, Lowell) ProQuest Dissertations and Theses.
- Kvale, S. (1996). Interviews an introduction to qualitative research interviewing. Thousand Oaks London: Sage.
- Ladkin, D. (2004). The phenomenological roots of action research. Action Research, 3(1), 109-127
- Latuconsina, N. K., Sriyanti, A. and Halimah, A., (2018). The Effect of the Realistic Mathematical Approach Towards the Result of Learning Mathematics Reviewed from the Ability of Numerical Students. In *Journal of Physics: Conference Series 1028*(1), p. 012170.
- Leask, M. (2001) Issues in Teaching using ICT. London: RoutledgeFalmer.
- Leask, M. and Pachler, N., (2013). Learning to Teach Using ICT in the Secondary School: A companion to school experience. Routledge.
- Leask, M., Younie, S. and Turner, T. (eds) (2013). Learning to Teach in the Secondary School : A Companion to School 1009 Experience, Taylor and Francis, Florence. Available from: ProQuest Ebook Central. [14 January 2018]. Created from ucl on 2018-01-14 00:35:11.P1040)
- Leron, U., (1983). Structuring mathematical proofs. *The American Mathematical Monthly*, *90*(3), pp. 174-184.

- Lester, S. (1999) An introduction to phenomenological research. Stan Lester: Developments,Taunton. http://www.sld.demon.co.uk/resmethy.pdf (Assessed 06/07/2019)
- Leung, A. (2017). Variation in Tool-Based Mathematics Pedagogy. In: *Teaching and Learning Mathematics through Variation*, Sense Publishers, Rotterdam (pp. 69–84).
- Leung, L. (2015). Validity, reliability, and generalizability in qualitative research. J Family Med Prim Care. Available from: http://www.jfmpc.com/text.asp?2015/4/3/324/161306 (Assessed: 10/08/2019).
- Liljedahl, P., (2005). Mathematical discovery and affect: The effect of AHA! Experiences on undergraduate mathematics students. *International Journal of Mathematical Education in Science and Technology*, *36*(2-3), pp. 219-236.
- Liljedahl, P., (2007). Persona-based journaling: striving for authenticity in representing the problem-solving process. *International Journal of Science and Mathematics Education*, 5(4), pp. 661-680.
- Lim, C. P., and Khine, M. S. (2006). Managing subject leaders' barriers to ICT integration in Singapore schools. Journal of Technology and Teacher Education, 14(1), 97–125.
- Lin, J.M.C., Wang, P.Y. and Lin, I., (2012). Pedagogy technology: A two- dimensional model for subject leaders ICT integration. British Journal of Educational Technology, 43(1), pp.97-108.
- Linnenbrink, E. A. and Pintrich, P. R., (2002). Achievement goal theory and affect: an asymmetrical bidirectional model, *Educational Psychologist* 37, pp. 69-78.
- Livingstone, S., (2012) Critical reflections on the benefits of ICT in education. Oxford review of education, 38(1), pp.9-24.
- Loomis, E. S., (1968). *The Pythagorean proposition*. Washington, DC: National Council of Teachers of Mathematics.
- Loveless, A., DeVoogd, G.L. and Bohlin, R.M. (2001) Something old, something new... Is pedagogy affected by ICT, in: V. Ellis (Ed) ICT, Pedagogy and the Curriculum (London and New York, RoutledgeFalmer).
- Lu, Y.-W. A., (2008). The use of Mathematical Software in Secondary Mathematics Teaching: a New Challenge for Young Teachers. 11th International Congress on Mathematical Education (ICME).
- Lumby, J. (2012). Leading organizational culture: Issues of power and equity. Educational Management Administration & Leadership, 40(5), pp.576-591.

- Lupton, T. (1991). "Organisational Change: "Top- down" or "Bottom- up" Management?", Personnel Review, Vol. 20 Issue: 3, pp.4-10, https://doi.org/10.1108/EUM000000000788. (Assessed 10/03/2020)
- Ma, Y., Zhang, S., Lin, C. and Li, L. (2017). A Data Collection Method Based on the Region Division in Opportunistic Networks. Applied Computational Electromagnetics Society Journal, 32(1).
- MacGilchrist, B. (2003). *Has school improvement passed its sell-by date?* London: Institute of Education.
- Malibar, I. and Pountney, D.C. (2002). Using technology to integrate constructivism and visualisation in mathematics education. Paper presented at the 2nd International Conference on the Teaching of Mathematics. Hersonissos, Crete, Greece, July 1-6. (Accessed: 20/5/2019).
- Mama, M. and Hennessy, S. (2013). Developing a typology of teacher beliefs and practices concerning classroom use of ICT. Computers and Education, 68, pp.380-387.
- Mann, W. and Tall, D. (1992). Computers in the Mathematics Curriculum. Leicester: The Mathematical Association.
- Mareschal, D., Butterworth, B., Tolmie, A. eds., (2013). *Educational Neuroscience*. Oxford: Wiley-Blackwell.
- Martin-Alcazar, F., Romero-Fernandez, P. M., and Sanchez-Gardey, G. (2005). Strategic human resource management: integrating the universalistic, contingent, configurational and contextual perspectives. International Journal of Human Resource Management, 16(5), 633-659.
- Marzano, R. J. (2003). What works in schools: Translating research into action. Alexandria, VA: ASCD.
- Masri, R.M., Ting, S. H., Zamzamir, Z. and Ma'amor, R.L.Z.R. (2017). The effects of using GeoGebra teaching strategy in Malaysian secondary schools: A case study from Sibu, Sarawak. *Geografia-Malaysian Journal of Society and Space*, 12(7).
- Mata-Pereira, J. and da Ponte, J.P. (2017). Enhancing students' mathematical reasoning in the classroom: teacher actions facilitating generalisation and justification. *Educational Studies in Mathematics*, 96(2), pp.169–186.
- Mauthner, M., Birch, M., Jessop, J., and Miller, T. (Eds.). (2002). *Ethics in qualitative research*. London: Sage.

- Mavrikis, M., Noss, R., Hoyles, C. and Geraniou, E. (2012). Sowing the seeds of algebraic generalisation: designing epistemic affordances for an intelligent microworld Journal of Computer Assisted Learning, 29, 68-84.
- Mavrikis, M., Noss, R., Hoyles, C. and Geraniou, E. (2013). Sowing the seeds of algebraic generalisation: designing epistemic affordances for an intelligent microworld. In R., Noss and A., diSessa, (eds.) Special issue on knowledge transformation, design and technology, *Journal of Computer Assisted Learning*, 29(1), pp.68-84.
- McEwan, H. and Bull, B. (1991). The pedagogic nature of subject matter knowledge. *American Educational Research Journal*, 28(2), pp.316–334.
- McFarlane, A. and Sakellariou, S. (2002). *The role of ICT in science education*. Cambridge Journal of Education, 32, pp.219-232.
- Merkley, D. J., Bozik, M. and Oakland, K, (1997). Investigating Support for Teachers Using Distance Learning in Education: a Case Study. Deonews 7(11). http://www.ed.psu.edu/acsde/deos/ deosnews/deosnews7_11.asp.
- Meyer, R. A. (1980). Mathematical problem-solving performance and intellectual abilities of fourth-grade children. In: J. G. Harvey & T. A. Romberg (Eds.), *Problem solving studies in mathematics*, pp. 179-1980, Madison; Wisconsin Research and Development centre monograph series.
- Miles, M.B and Huberman, A. M. (1994). Qualitative Data Analysis: An Expanded Source, 2nd ed. Thousand Oaks, CA:SAGE.
- Mingaine, L. (2013). 'Leadership challenges in the implementation of ICT in public secondary schools, Kenya', *Journal of Education and Learning*, 2(1), pp. 61-71: Canadian Center of Science and Education.
- Mishra, P. and Koehler, M. (2006). Technological pedagogical content knowledge: a framework for integrating technology in teacher knowledge. *Teachers College Record*, *108*(6), pp.1017–1054.
- Mishra, P. and Koehler, M. J. (2006). Technological pedagogical content knowledge: a new framework for teacher knowledge. *Teachers College Record.* 108(6), pp.1017-1054.
- Mishra, P. and Koehler, M.J. (2009). What is Technological Pedagogical Content Knowledge. Contemporary Issues in Technology and Teacher Education, 90(1), pp.60–70.

- Mizuno, N., Kaisha, K., and Denki, Y. (2016). Trace data collection system, controller, motor control device, operation terminal, and trace data collection method. U.S. Patent Application 15/140,473.
- Monaghan, J. (2004). Teachers' activities in technology-based mathematics lessons. International Journal of Computers for Mathematical Learning, 9(3), pp.327–357.
- Morgeson, F. P., DeRue, D. S., Karam, E. P. (2010). Leadership in Teams: A Functional Approach to Understanding Leadership Structures and Processes: Journal of Management, Vol. 36 No. 1, January 2010 5-39 DOI: 10.1177/0149206309347376 : Southern Management Association.
 <u>https://msu.edu/~morgeson/morgeson_derue_karam_2010.pdf</u> (Accessed; 15/08/2019)
- Mullins, L.J. (2007). Management and Organisational Behaviour. (8th ed.) Harlow: Financial Times Prentice Hall.
- Mullis, I. V. S., Martin, M. O., Kennedy, A. M., Trong, K. L. & Sainsbury, M. (2009). PIRLS 2011 assessment framework. Chestnut Hill, MA: TIMSS & PIRLS International Study Center, Lynch School of Education, Boston College.
- Mumtaz, S. (2000). Factors affecting subject leaders' use of information and communications technology: a review of the literature. Technology, Pedagogy and Education, 9(3), 319-342.
- Murray, J. C. (1996). A generalised Van Hiele. Cape Town, South Africa: University of Stellenbosch.
- National Council for Accreditation of Teacher Education. (2001). Professional standards for the accreditation of schools, colleges, and departments of education. [online] Available from: <u>http://www.ncate.org/2000/2000stds.pdf</u> (Accessed: 27/6/ 2019).
- National Council of Teachers of Mathematics. (1991). Professional standards for teaching mathematics. Reston, VA: Author.
- National Council of Teachers of Mathematics. (2000). Professional standards for teaching mathematics. Reston, VA: Author.
- National Science Teachers Association. (1999). *NSTA standards for science teacher preparation*. [online]: <u>http://www.iuk.edu/faculty/sgilbert/nsta98.htm</u> (Accessed: 28/6/ 2019).
- Navas, A. (2016). A remark concerning bi-Lipschitz equivalence of Delone sets Notes available from the author (not intended for publication)

- Nelsen, R. B., (2000). *Proofs without words II: More exercises in visual thinking*. Washington, DC ; (Great Britain) : Mathematical Association of America.
- Nemirovsky, R. and Borba, M., (2004). PME special issue: Bodily activity and imagination in mathematics learning. *Educational Studies in Mathematics*, (57), pp. 303-305.
- Neugebaur, O., (1969). The exact sciences in antiquity. 2nd ed. New York: Dover Publications.
- Newby, P. (2010). Research Methods for Education. Harlow: Longman.
- Newhouse, P. and Rennie, L. (2001). A longitudinal study of the use of student-owned portable computers in a secondary school, Computers & Education 36(3), pp. 223–243.
- NFER (National Foundation for Educational Research). (2008). England's achievement in TIMSS 2007: National report for England. NFER: Berkshire.
- Niess, M. L., Ronau, R. N., Shafer, K. G., Driskell, S. O., Harper S. R., Johnston, C., Browning, C., Özgün-Koca, S. A. and Kersaint, G. (2009). Mathematics teacher TPACK standards and development model. *Contemporary Issues in Technology and Teacher Education*, 9(1), pp.4-24.
- Norris, D., McQueen, J. M., and Cutler, A.(2003). "Perceptual learning in speech," Cogn. Psychol. 47, 204–238.
- Noss, R. (1997). On static and dynamic algebra systems. In: J. Kaput, ed. 1997. *Early Algebra*. New York: SUNY Press. pp. 203-233
- Noss, R., and Hoyles, C. (1996). Windows on Mathematical Meanings: Learning Cultures and Computers. Dordrecht: Kluwer Academic Publishers. Studies in Mathematics, 96(2), pp.169–186.
- O'Reilly, D., and Reed, M. (2010). Leaderism: An evolution of managerialism in UK public service reform. Public Administration, 88, 960–978.
- Öçal, M.F. (2017). Asymptote Misconception on Graphing Functions: Does Graphing Software Resolve It? *Malaysian Online Journal of Educational Technology*, *5*(1), pp.21–33.
- OECD (Organisation for Economic Co-operation and Development). (2000). Viewing the United Kingdom school system through the prism of PISA. OECD Publishing, London

- OECD (Organisation for Economic Co-operation and Development). (2003).. Viewing the United Kingdom school system through the prism of PISA. OECD Publishing, London
- OECD (2011). How do education systems and schools select and group students?, in PISA 2009 at a Glance, OECD Publishing, Paris.DOI: http://dx.doi.org/10.1787/9789264095250-35-en.
- OECD. (2001). The Well-being of Nations: The Role of Human and Social Capital:Organisation for Economic Co-operation and Development. OECD Publishing, London
- OECD. (2007). Human capital: how what you know shapes your life: OECD. OECD Publishing, London
- Office for Standards in Education (OfStEd] (2004)). *ICT in schools the impact of government initiatives: Secondary mathematics*. London: OfStEd.
- Ohman, A. (2005). Qualitative methodology for rehabilitation research. Journal of Rehabilitation Medicine, 37(5), 273-280.
- Oldknow, A. and Taylor, R. (2003). *Teaching Mathematics; using Information and Communications Technology*. Continuum: London
- Olive, J. and Makar, K., with V. Hoyos, L. K. Kor, O. Kosheleva and R. Straesser (2010).
 Mathematical knowledge and practices resulting from access to digital technologies.
 In C. Hoyles & J. Lagrange (Eds.), *Mathematics education and technology Rethinking the terrain. The 17th ICMI Study* (pp.133–177). New York: Springer
- Olsson, J. (2017). *GeoGebra, enhancing creative mathematical reasoning*. PhD. Umea Universitet.
- Olsson, J. (2019). Relations Between Task Design and Students' Utilization of *GeoGebra*. *Digit Exp Math Educ* **5**, 223–251 (2019).
- Omer, M.A.M. and Wilson, D. (2016). October. New rules for deriving formal models from text. In Students on Applied Engineering (ISCAE), International Conference for (pp. 328-333). IEEE.
- Orlando, J. (2011). ICT, constructivist teaching and 2L1 century learning.Curriculum and Leadership Journal 9(11). http://www.curriculum.edu.au/leader/ict,_constructivist_teaching_and_2L1_century _lear,33278.html?issueID=12401 (Assessed: 22/04/19).

- Osborn, M. (2001). Constants and Contexts in Pupil Experience of Learning and Schooling: Comparing learners in England, France and Denmark. *Comparative Education*. 37(3), pp. 267-278.
- Patel, M., Doku, V., and Tennakoon, L. (2017). Challenges in recruitment of research participants. Advances in Psychiatric Treatment. 9. (2003) (accessed: 22/06/ 2019)
- Pelgrum, W. J., and Law, N. (2003). ICT in Education around the World: Trends, Problems and Prospects UNESCO-International Institute for Educational Planning. Available: <u>www.worldcatlibraries.org/wcpa/ow/02d077080fcf3210a19afeb4da09e526.html</u>. (Accessed: 11/07/2019)
- Pelgrum, Willem J. (1993). Attitudes of School Principals and Teachers towards Computers: Does it Matter What They Think? Studies in Educational Evaluation 19(2): 199–212.
- Pettigrew, A.M. (2014). The politics of organizational decision-making. Routledge.
- Phillips, T. and Scrimshaw, P. (1997). Talk around adventure games, in: R. Wegerif and P. Scrimshaw (Eds) Computers and talk in the primary classroom (Clevedon, Multilingual Matters), 65–80.
- Piaget, J., (1936). Origins of intelligence in the child. London: Routledge and Kegan Paul.
- Pierce, R. and Stacey, K. (2010). Mapping pedagogical opportunities provided by mathematics analysis software. *International Journal of Computers for Mathematical Learning*, 15(1), p.20.
- Pine, G.J. (2009). *Teacher Action Research: Building Knowledge Democracies*. Los Angeles: SAGE Publications.
- Pinello, D, (2016). Socio economic data collection: The case of Lebanon.
- Plomp, T.; Pelgrum, W. J. & Law, N. (2007). 'SITES2006—International comparative survey of pedagogical practices and ICT in education', Education and Information Technologies Vol.12, No. (2), Pp; 83-92.
- Polit, D.F., Beck, C.T. and Hungler, B.P. (2001). Essentials of Nursing Research: Methods, Appraisal and Utilization. L5 Ed., Philadelphia: Lippincott Williams & Wilkins.
- Polizzi, G. (2011). The National Association for Media Literacy Education's Journal of Media Literacy Education 3:2, 113 - 122 Measuring School Principals' Support for ICT Integration in Palermo, Italy available on www.jml.org.
- Pollard, A. (2011). Department of Quantitative Social Science, Institute of Education, University of London, UK.

- Pring, R., Hayward, G., Hodgson, A., Johnson, J., Keep, E., Oancea, A., Rees, G., Spours, K. & Wilde, S. (2009). *Education for All: the future of education and training for 14-19 year olds*. Abingdon: Routledge.
- Punch, K.F. (2005). Introduction to Social Research–Quantitative & Qualitative Approaches. London: Sage
- Punie, Y. and Cabrera, M. with Bogdanowiez, M.,Zinnbauer, D. and Navajas, E. (2006). The Future of ICT and Learning in the Knowledge Society, Report on a Joint DG JRCDG EAC Workshop held in Seville, 2021 October 2005, Brussel: European Commission, Directorate General, Joint Research Centre.
- Raelin, J. A. (2016). Leadership-as-Practice: Theory and Application: Northeastern U. D'Amore-McKim School of Business Research Paper No. 2776794. Available at SSRN: <u>https://ssrn.com/abstract=2776794</u>
 <u>https://www.researchgate.net/publication/302029718_Introduction_to_Leadership-as-Practice.</u> (Assessed: 20/07/2019)
- Ramsey, W., Stich, S. P. and Rumelhart, D. W. (1991). Philosophy and Connectionist Theory. In: *Developments in Connectionist Theory*, Rumelhart and Gluck, eds.
- Read, (2004). Interpreting the Mathematics Curriculum: Developing reasoning through algebra and geometry. National Curriculum Ma, QCA.
- Reeves, T.C., and Hedberg, J.G. (2003) Interactive learning systems evaluation. Englewood Cliffs, N.J.: Educational Technology Publications.
- Reisby, K. and Schnack, K. (Eds.) (1995). What can Curriculum Studies and Pedagogy learn from Sociology Today? Copenhagen: Royal Danish School of Educational Studies
- Reiss, M.J. & White, J. (2014). An aims-based curriculum illustrated by the teaching of science in schools. The Curriculum Journal, 25, p.76-89.
- Richards, J.C., and Farrell, T.S.C. (2005) *Professional development for language teachers: strategies for teacher learning*. New York: Cambridge University Press.
- Richards, J.C., and Lockhart, C. (1996). Reflective teaching in second language classrooms. 6th edn. New York: Cambridge University Press.
- Ritchie, J., Lewis, J., Nicholls, C. and Ormston, R. (2013). *Qualitative research practice*.2nd ed. London: Sage Publications.
- Ritz, J. M. (2009). A new generation of goals for technology education. Journal of Technology Education, 20(2), pp.50–64

- Robertson, M. and Al-Zahrani, A. (2012). Self-efficacy and ICT integration into initial teacher education in Saudi Arabia: Matching policy with practice. Australasian Journal of Educational Technology, 28(7).
- Roblyer, M.D. and Doering, A. (2010). *Integrating Educational Technology into Teaching*.5th ed. Boston, MA: Pearson Education, Inc.
- Robson, C. (2011). Real World Research. 3rd edn. Chichester: Wiley
- Rodd, M., (2010). Geometrical visualisation epistemic and emotional. For the Learning of Mathematics 30(3), pp. 29-35.
- Rogers, L. (2004). Integrating ICT into science education and the future. In R. Barton (Ed.), *Teaching secondary science with ICT* (pp. 139-154). Maidenhead, England: Open University Press.
- Rohaan, E., Taconis, R. and Jochems, W. (2010). Analysing teacher knowledge for technology education in primary schools. *International Journal of Technology & Design Education*, 22, pp.271–280.
- Rohann, E., Taconis, R., & Jochems, W. (2010). Reviewing the relations between teachers' knowledge and pupils' attitude in the field of primary technology education.
 International Journal of Technology and Design Education, 20(1), 15–26.
- Ross, A. (2000). Curriculum: Construction and Critique. London: Falmer Press.
- Rost, J.C. (1993). Leadership for the 2L1 Century. New York, NY: Praeger.
- Rota, G. C., (1997). The phenomenology of mathematical proof. *Synthese*, 111, pp. 183-196.
- Rousseau, J.J. (1968). *The Social Contract.* Trans. Maurice Cranston. Harmondsworth, England: Penguin Books.
- Rowland, T. and Ruthven, K. (2011). *Mathematical Knowledge in Teaching*. London: Springer.
- Rowland, T. (2014). 'The knowledge quartet: the genesis and application of a framework for analysing mathematics teaching and deepening teachers' mathematics knowledge', *SISYPHUS Journal of Education*, 1(3), pp. 15–43. Available at: <u>http://revistas.rcaap.pt/sisyphus/issue/view/293</u> (Accessed: 23/4/201).
- Rowland, T., Huckstep, P. and Thwaites, A. (2005). 'Elementary teachers' mathematics subject knowledge: the knowledge quartet and the case of Naomi'. *Journal of Mathematics Teacher Education* 8(3) pp. 255-281.

- Rowland, T., Turner, F., Thwaites, A. and Huckstep, P. (2009). Developing Primary Mathematics Teaching: Reflecting on practice with the Knowledge Quartet. London: Sage.
- Royal Society and Joint Mathematical Council (2001). Teaching and Learning Geometry 11-19, London, Royal Society/Joint Mathematical Council.
- Royal Society, (2011). *Neuroscience: Implications for education and lifelong learning* [online] Available at: http://royalsociety.org/policy/projects/brain-waves/educationlifelong-learning/ (Accessed 26/6 2019).
- Rubin, A. and Babbie, E.R. (2012) Brooks/Cole Empowerment Series: Essential research methods for social work. Cengage Learning.
- Rubin, H. J., and. Rubin, I. S. (2005): Qualitative Interviewing The Art of Hearing Data.2. Edition, Sage Publications, Thousand Oaks, London, New York.
- Ruthven, K. (1998). The Use of Mental, Written and Calculator Strategies of Numerical Computation by Upper Primary Pupils within a 'Calculator- aware' Number Curriculum.
- Ruthven, K. (2007). Teachers, technologies and the structures of schooling. In: D. Pitta-Pantazi and G. Philipou, eds. 2007. *Proceedings of the Fifth Congress of the European Society for Research in Mathematics Education CERME 5*. Cyprus: Larnaca. pp.52–67.
- Ruthven, K. (2009). Towards a naturalistic conceptualisation of technology integration in classroom practice: The example of school mathematics. *Education and Didactique*, 3(1), pp.131–159.
- Ruthven, K. (2011). Using international study series and meta-analytic research syntheses to scope pedagogical development aimed at improving student attitude and achievement in school mathematics and science, *International Journal of Science and Mathematics Education* 9 (2), pp. 419-458
- Ruthven, K. (2014). Frameworks for analysing the expertise that underpins successful integration of digital technologies into everyday teaching practice. In: A. Clark-Wilson, O. Robutti and N. Sinclair, eds., *The mathematics teacher in the digital era* (pp.373–393). Dordrecht: Springer.
- Ruthven, K. and Hennessy, S. (2003). A teacher's perspective on successful ICT use in secondary Mathematics teaching, Micromath, 19(2), pp.20-24.

- Ruthven, K., Deaney, R. and Hennessey, S. (2008). Practitioner Use of Graphing Software to Teach about Algebraic Forms, in Jouberts, M. *Proceedings of the BSRLM*, 28(3).
- Ruthven, K., Deaney, R. and Hennessy, S. (2009). Using graphing software to teach about algebraic forms: a study of technology-supported practice in secondary school mathematics. *Educational Studies in Mathematics*, *71*(3), pp.279–297.
- Ruthven, K., Hennessy, S. and Brindley, S. (2004). Teacher representations of the successful use of computerbased tools and resources in secondary-school English, Mathematics and Science. *Teaching and Teacher Education*, 20(3), pp.259–275.
- Ruthven, K., Hennessy, S., and Deaney, R., (2008). Constructions of dynamic geometry: A study of the interpretative flexibility of educational software in classroom practice. *Computers and Education 51*(1), pp. 297-317.

S. Ackroyd and J. A. Hughes, Data Collection in Context (1981). Longman

- Sahlberg, Pasi (2016). How GERM is infecting schools around the world?. [online] Available at: http://pasisahlberg.com/text-test/ (Accessed 10/12/ 2017).
- Salame, M., del Rosario, P. C., Uzho, A. and Paola, S., (2016). Use of bodily-kinesthetic intelligence theory to improve the creative learning of the English language in students of 1st EGB at Unidad Educativa Bilingue Instituto Particular Abdón Calderón. (Bachelor's thesis, Guayaquil: ULVR, 2016.).
- Salsabila, F., Johar, R. and Wahyuni, S. P., (2017). Students' Ability in Proving Pythagorean Theorem through Discovery Learning Model Using Geogebra Software. *Proceedings of AICS-Social Sciences*, 7, pp. 717-726.
- Sandoltz, J H, Ringstaff, C and Dwyer, D C. (1997). Teaching with technology: Creating studentcentered classrooms and Sandoltz, J.H. Learning to teach with technology: A comparison of teacher development programs, as cited in N. White, C. Ringstaff, and L.
- Sanyal, B. C. (2001). New functions of higher education and ICT to achieve education for all, Paper prepared for the Expert Round table on University and Technology-for Literacy and Education Partnership in Developing Countries, International Institute for Educational Planning, UNESCO, September 10 to 12, Paris.
- Saudelli M. G. and Ciampa. K. (2016). Exploring the role of TPACK and teacher selfefficacy: an ethnographic case study of three iPad language arts classes. *Technology, Pedagogy and Education*, 25(2), pp.227-247.

- Saudelli, M. G., & Ciampa, K. (2016). Exploring the role of TPACK and teacher selfefficacy: An ethnographic case study of three iPad language arts classes. *Technology, Pedagogy and Education*, 25(2), 227-247. doi:10.1080/1475939X.2014.979865
- Saunders, H. (1988). When Are We Ever Gonna Have to Use This? Dale Seymour Publications. Papick, Ira J. (2007) Algebra Connections Mathematics for Middle School Teachers. Pearson Prentice
- Saunders, M. N. K. (2012). Choosing research participants. In G. Symon, and C. Cassell (Eds.), The practice of qualitative organizational research: Core methods and current challenges (pp. 37- 55). London, England: Sage.
- Savage, M., and Williams, J., (1990). *Mechanics in action: Modelling and practical investigations*. Cambridge: Cambridge University Press.
- Schein, E.H. (2010). Organizational Culture and Leadership. (4th ed.) San Francisco, CA: Jossey-Bass.
- Schiller, J. (2002). Interventions by school leaders in effective implementation of information and communications technology: perceptions of Australian principals, Journal of Information Technology for Teacher Education, 11:3, 289-301, DOI: 10.1080/14759390200200138.
- Schleicher, A. (2012). Preparing Teachers and Developing School Leaders for the 21st Century: Lessons from around the World, OECD Publishing.
- Schleicher, A. (2015) Preparing Teachers and Developing School Leaders for the 2L1 Century: Lessons from around the World, OECD Publishing, Paris, http://dx.doi.org/10.1787/9789264174559-en.
- Schmidt, D. A., Baran, E., Thompson, A. D., Mishra, P., Koehler, M. J., & Shin, T. S. (2009). Examining preservice teachers' development of technological pedagogical content knowledge in an introductory instructional technology course. In I. Gibson, R. Weber, K. McFerrin, R. Carlsen, & D. A. Willis (Eds.), Society for Information Technology and Teacher Education International Conference book, 2009 (pp. 4145–4151). Chesapeake, VA: Association for the Advancement of Computing in Education (AACE).
- Scott, W., Reid, A. & Jones, N. (2003). Growing Schools- Innovation Fund Projects: an external evaluation. Council for Environmental Education and University of Bath.

- Segall, A. (2004). Revisiting pedagogical content knowledge: the pedagogy of content/the content of pedagogy. *Teaching and Teacher Education*, 20(5), pp.489–504.
- Seidman, I. (2013). *Interviewing as qualitative research: a guide for researchers in education and the social sciences*. New York: Teachers College Press.
- Sen, B.A. and N. Ford, (2009). 'Developing reflective practice in LIS education: the SEAchange model of reflection', *Education for Information*, 27(4), pp. 181–95.
- Serhan, D. (2007). School Principals Attitudes Towards the Use of Technology: United Arab Emirates Technology Workshop. The Turkish Online Journal of Educational Technology 6(2): 42–46.
- Sexton, M. (2003). A supple approach to exposing and challenging assumptions and PhD path dependencies in research, Key note speech of the L3 International Postgraduate Research Conference, Lisbon.
- Sharma, R. (2003). Barriers in Using Technology for Education in Developing Countries', IEEE0-7803-7724-9103.Singapore schools, Computers & Education Vol .41, No.(1),Pp; 49--63.
- Sherman, M. (2014). The role of technology in supporting students' mathematical thinking: Extending the metaphors of amplifier and reorganizer. *Contemporary Issues in Technology and Teacher Education*, 14(3), pp.220-246.
- Shulman, L. (1986). Paradigms and research programs in the study of teaching: A contemporary perspective. In: M. C. Wittrock, ed. *Handbook of research on teaching*. New York: MacMillan, pp.3-36.
- Shulman, L. S. (1987). Knowledge and teaching: Foundations of the new reform. *Harvard Educational Review*, 57(1), 1–22.
- Shulman, L.S. (1986). Those who understand: Knowledge growth in teaching. *Educational Researcher*, *15*(2), pp.4–14.
- Siemens G (2006). Connectivism. Learning theory or pastime for the self-amused? Retrieved from http://www.elearnspace.org/Articles/connectivism_self-amused.htm
- Siemens, G. (2004). Connectivism: A learning theory for the digital age. *International Journal of Instructional Technology and Distance Learning*. Retrieved on November 12, 2019, from http://www.itdl.org/Journal/Jan_05/article01.htm.
- Silverman, D. (2006) Interpreting Qualitative Data (L3 ed). London: Sage Publications Ltd. <u>http://www.uk.sagepub.com/upm-data/11254_Silverman_02.pdf</u>.

- Skelton, A (2011). Value conflicts in higher education teaching, *Teaching in Higher Education*, 17(3) pp. 257-26.
- Slavin, E. R. (2013). Theory and Practice in Educational Psychology. (Translation Editor: Galip Yüksel). Ankara: Nobel Publications.
- Solberg Søilen, K. & Huber, S. (2004). Berglunds Mekaniska AB. Jönköping: Jönköping International Business School.
- Somekh B .(2008). Factors affecting teachers' pedagogical adoption of ICT: Education and Social Research Institute, Manchester Metropolitan University, Manchester, UK.
- Soni, A.S., Weinert, S., Rasch-Menges, J., Walling, P.D., Bousamra, S.A. and Greenburg, A.M., Roche Diabetes Care, Inc., (2017). Management method and system for implementation, execution, data collection, and data analysis of a structured collection procedure which runs on a collection device.U.S. Patent 9,659,037.
- Sorensen, P., Twidle, J., Childs, A. and Godwin, J. (2007). The use of the internet in science teaching: A longitudinal study of developments in use by student-teachers in England. International *Journal of Science Education*, 29, pp.1605-1627.
- Sousa, D.A., (2014). How the brain learns mathematics. Thousand Oaks: Corwin Press.
- Southworth, G. (1993) School leadership and school development: Reflections from research School Organisation, 13(1), pp. 73–87. doi: 10.1080/0260136930130107.
- Soy, S (1997). The case study as a research method. Unpublished paper, University of Texas at Austin.
- Stainback, S.B. and Stainback, W.C. (1988) Understanding and conducting qualitative research. Dubuque, IA: Kendall/Hunt.
- Stenhouse, L. (1975). An Introduction to Curriculum Research and Development. Oxford: Heinemann.
- Stevenson, D. (1997) Information and Communications Technology in UK School: An independent Inquiry. London: The Independent ICT in Schools Commission. .
- Stigler, J., & Hiebert, J. (2004). The teaching gap: Best ideas from the world's teachers for improving education in the classroom. New York, NY: Free Press.
- Strathern, P., (1997). Pythagoras and his theorem. London: Arrow Books.
- Stupel, M. and Ben-Chaim, D. (2017). Using multiple solutions to mathematical problems to develop pedagogical and mathematical thinking: A case study in a teacher education program. *Investigations in Mathematics Learning*, 9(2), pp.86–108.

- Subramaniam, K. B., (2018). Pythagoras' theorem: A proof by contradiction. *The Mathematical Gazette*, *102*(553), pp. 128-128.
- Tabachnick, B. R., and Zeichner, K. (1985). The development of teacher perspectives: Final report. Madison: University of Wisconsin, Wisconsin Center for Education Research.
- Tanujaya, B., Prahmana, R.C.I. and Mumu, J. (2017). Mathematics instruction, problems, challenges, and opportunities: A case study in Manokwari regency, Indonesia. World Transactions on Engineering and Technology Education, 15, pp.287.
- Teaching and learning geometry 11-19 Report of a Royal Society / Joint Mathematical Council working group (T and L G).
- Teknomo, K. (2016). Microscopic pedestrian flow characteristics: Development of an image processing data collection and simulation model. arXiv preprint arXiv:1610.00029.
- Teknomo, K. (2016). Microscopic pedestrian flow characteristics: Development of an image processing data collection and simulation model. arXiv preprint arXiv:1610.00029.
- Tella, A. (2008). Teacher variables as predictors of academic achievement of primary school pupils' mathematics. International Electronic Journal of Elementary Education, 1(1):16-33. Available at http://iejee.com/files/1/articles/article_551316212fd29_last_article_5516c591 9ba0f.pdf.(Accessed17/4/ 2019).
- Thomas, G. (2009). How to do your Research Project: A Guide for Students in Education and Applied Social Sciences. London: SAGE Publications Ltd.
- Thorpe, A. (2014). why the leadership and management divide matters in education: the implications for and schools and social justice. International Journal of Education for Social Justice (REIJS) 3(2), 199-212.
- Thorpe, A. and Tran, D.T.H. (2015). The professional development of beginning teachers and the role of subject leaders: Vietnamese teachers' perspectives. International Journal of Educational Research 71(1), 108-117.
- Tikly, L. (2001). Globalisation and Education in a Postcolonial world: Towards a conceptual framework, Comparative Education, 37, (2) pp. 151-171.
- Tinio, V. (2002). Survey of ICT Utilization in Philippine Public High Schools: *Preliminary Findings* [Unpublished Manuscript].

- Tondeur, J., Keer, V. H., Braak, J.N., and Valcke M. (2008). ICT Integration in the Classroom: Challenging the Potential of a School Policy. Computers and Education 51(1): 212–223.
- Trends in International Mathematics and Science Study. (1999). TIMSS 1999 Science Assessment Public Released Items for Grade 8. Retrieved March 18, 2018, from <u>http://timss.bc.edu/TIMSS2007/items.html.</u>
- Trends in International Mathematics and Science Study. (2007). TIMSS 2007 Science Assessment Public Released Items for Grade 8. Retrieved March 18, 2018, from http://timss.bc.edu/TIMSS2007/items.html.
- Trifonas, P.P. (Ed.) (2008). Worlds of Difference: Rethinking the ethics of Global Education for the 21st Century. London: Paradigm Publishers.
- Trouche, L., (2005). Instrumental genesis, individual and social aspects. In: D. Guin, K. Ruthven & L. Trouche, eds. The didactical challenge of symbolic calculators. New York: Springer, p. 197–230.
- Trowler, P. (2003). Education Policy (L2 ed). London: Routledge.
- Truss, C., Gratton, L., Hope-Hailey, V., McGovern, P., and Stiles, P. (1997). 'Soft and hard models of human resource management: a reappraisal', Journal of Management Studies, 34, 1, 53-73.
- U.S. Department of Education. (2003). Federal funding for educational technology and how it is used in the classroom: A summary of findings from the Integrated Studies of Educational Technology. Office of the Under Secretary, Policy and Program Studies Service: Washington, D.C www.ed.gov/about/offices/list/os/technology/e valuation.html (Accessed: 06/06/2019)
- UNESCO (1999). Information and Communication Technology in Education–A Curriculum for Schools and Programme for Teacher Development. Paris: UNESCO.
- UNESCO (2005). Information and Communication Technology in Education–A Curriculum for Schools and Programme for Teacher Development. Paris: UNESCO.
- UNESCO, (2002). Open And Distance Learning Trends, Policy And Strategy Considerations, 14 UNESCO.
- UNESCO, (2003). Building Capacity of Teachers/Facilitators in Technology-Pedagogy Integration for Improved Teaching and Learning; Final Report Experts' Meeting on Teachers/Facilitators Training in Technology-Pedagogy Integration 18-20: Bangkok, Valtonen, T., Kukkonen, J., Kontkanen, S., Sormunen, K., Dillon, P. and Sointu, E.,

(2015). The impact of authentic learning experiences with ICT on pre-service subject leaders' intentions to use ICT for teaching and learning. Computers & Education, 81, pp.49-58.

- Van Hiele, P. M., (1986). *Structure and insight: A theory of mathematics education*. Orlando, Florida: Academic Press.
- Vannatta, R. A. and Fordham, N. (2004). Teacher dispositions as predictors of classroom technology use. *Journal of Research on Technology in Education*, 36, pp.253–272.
- Venkataraman, G. (2012). Innovative activities to develop geometrical reasoning skill in secondary mathematics with the help of open resource software "GeoGebra". *National Conference on Mathematics Education*. Mumbai. pp. 20–22.
- Venkatesh, Viswanath, Morris, M, G., Davis, G., B., and Davis F. D., (2003). "User Acceptance of Information Technology. Toward a Unified View." MIS Quarterly 27(3): 425–478.
- Vygotsky, L. S. (1978). Mind in Society: The Development of Higher psychological Processes. Cambridge. USA: Harvard University Press.
- Wallace, M.J. (1991). Training foreign language teachers: a reflective approach. 2nd edn. Cambridge: Cambridge University Press.
- Walliman, N. and Buckler, S. (2008) Your Dissertation in Education. London: Sage.
- Walsham, G. (1993). Interpreting Information Systems in Organizations. Wiley, Chichester.
- Waring, S. (2000). Can you Prove it? Developing Concepts of Proof in Primary and Secondary Schools, The Mathematical Association.
- Waters-Adams, S. (2006). Action research in education. Plymouth: University of Plymouth. [online] Available at: <u>http://www.edu.plymouth.ac.uk/RESINED/actionresearch/arhome.htm</u> (Accessed: 20/5/ 2019).
- Watson, D.M. (1993). IMPACT An evaluation of the IMPACT of the Information Technology on Children's Achievements in Primary and Secondary Schools. King's College London.
- Waxman, H., Connell, M., Gray, J. (2002). A Quantitative Synthesis of Recent Research on the Effects of Teaching and Learning with Technology on Student Outcomes, NCREL. Available at: http://www.coe.ufl.edu/Courses/eme5054/
 Foundations/Articles/waxman.pdf. (Assessed: 26/3/ 2019).

- Wheelahan L. (2014). Not just skills: what a focus on knowledge means for vocational education. *Journal of Curriculum Studies*, 47, (6) pp. 750-762.
- White , J. and Bramall, S. (2000). *Why learn maths?* London: London Institute of Education; Bedford Way papers.
- White, P. (1998). Understanding Prolepsis through Teacher Research. *Networks: An Online Journal for Teacher Research*, *1*(1) <u>https://dx.doi.org/10.4148/2470-6353.1249.</u>
- Whitty, G. (2002). 'School Improvement and Social Inclusion: limits and possibilities' in *Making Sense of Education Policy*, London: Sage.
- Willingham, D. T. and Lloyd, J. W., (2007). How educational theories can use neuroscientific data. *Mind, Brain, and Education, 1*, pp. 140-149.
- Willis, J. (1995) Recursive, reflective instructional design model based on constructivistinterpretist theory. Educational Technology, 35 (6), 5-23.
- Willoughby, T. and Wood, E. (2008). *Children's Learning in a Digital World*. New York, NY: Blackwell Publishing.
- Wong, A. F. L., Quek, C. L., Divaharan, S. and Williams, M. D. (2006). Singapore students' and teachers' perceptions of computer-supported Project Work classroom learning environments. *Journal of Research on Technology in Education*, 38 (4) 449-479.
- Wood, D. (2003). Think Again: Hindsight, Insight and Foresight on ICT in Schools, Brussele: European School Network (EUN), Brussels.
- Wood, E. and Bennett, N. (2000). Changing theories, changing practice: exploring early childhood teachers' professional learning. Teaching and Teacher Education, 16(5/6), 635-647.
- Woodcock, M. (1985). Teams Development Manual Gower: Aldershot.
- Woodrow, J. (1990). Locus of Control and student teacher computer attitudes. Computers Educ. Vol. 14. No. 4 pp 421-432.
- World Bank, (2004). Can Africa Claim the 21st Century? Washington DC, World Bank.
- Yaniv, H., and Crichton S. (2008). S-AI-L: Simulated, Adventure Intelligent, Learning: An Instructional Design Mode, IASK: TL 2008, Aveiro, Portugal.
- Yin, R. K. (2009). Case study research: Design and methods. London: SAGE.
- Yin, R.K. (2016). Qualitative Research from Start to Finish, Second Edition. New York: The Guilford Press.
- Young M. (2014). What is a curriculum and what can it do? *The Curriculum Journal*. 25, (1) pp. 7-13.

- Young, J.R. (2017). Technology integration in mathematics education: examining the quality of meta-analytic research. *International Journal on Emerging Mathematics Education*, 1, p.71.
- Young, M. (2013). Overcoming the crisis in curriculum theory: a knowledge-based approach. *Journal of Curriculum Studies* 45(2), pp. 101-118.
- Young, M. (2016). 'What are schools for?' in *Curriculum and the specialisation of knowledge: studies in the sociology of education*. Routledge.
- Yuen, H.K., Law, N.W.Y., Wong, K.C. (2003). 'ICT implementation and school leadership: case studies of ICT integration in teaching and learning', *Journal of Educational Administration*, 2003, 41(2) pp. 158–170.
- Yusuf, M.O. (2005). 'Information and communication education: analysing the Nigerian national policy for information technology', *International Education Journal*, 6(3), pp. 316–321.
- Zbiek, R.M., Heid, M.K. and Dick, T.P. (2007). Research on Technology in Mathematics Education: a perspective of constructs. In: F.K. Lester Jr., ed. 2007. Second handbook of research on mathematics teaching and learning: A project of the National Council of Teachers of Mathematics. Charlotte NC: Information Age Publishers, pp.1169-1207.
- Zulnaidi, H. and Zamri, S.N.A.S. (2017). The effectiveness of the GeoGebra software: the intermediary role of procedural knowledge on students' conceptual knowledge and their achievement in mathematics. *Eurasia Journal of Mathematics, Science and Technology Education*, 13(6), pp. 2155–2180.

Appendices

Appendix 1: Case study 1

Applying Technological, Pedagogical Content Knowledge to Lesson

TPACK framework is based on three primary forms of knowledge. So my first step was to understand the primary forms of knowledge in the context of the lesson.

- <u>Content Knowledge (CK)</u>—what am I teaching and what is my own knowledge of the subject? For this lesson, I needed to establish a solid understanding of circle theorem and processes.
- <u>Pedagogical Knowledge (PK)</u>—how do my students learn best and what instructional strategies do I need to meet and what are the requirements of the lesson plan? In this case, I needed to understand the best practices for teaching secondary school year 10 mathematics using small group collaboration and the preferred learning styles in my class.
- Technological Knowledge (TK)—what digital tools are available to me, which do I know well enough to use, and which would be the most appropriate for the lesson at hand? Geogebra. For this lesson, students will explore through investigating the proofs associated with circle theorems. The ability to calculate angles and justify the reasons for answers are important.

By looking at the primary knowledge and focusing on the intersection whilst viewing the goal of the lesson and the strategy through the lens of TPACK, I took a moment to consider the individual relationships.

- Pedagogical Content Knowledge (PCK)—understanding the best practices for teaching specific content to specific students in my class.
- Technological Content Knowledge (TCK)—knowing how the digital tools available to me can enhance or transform the content, how it's delivered to students, and how my students can interact with it.
- Technological Pedagogical Knowledge (TPK)—I had to think about using the digital tool as a vehicle to the learning outcomes and implement the desired experiences.

• Finally, the technological, pedagogical content knowledge (TPACK) was put together to enhance the activities of the lesson plan.

Below are examples of activities that were considered;

- After walking through the different parts of a circle theorem,
- Put students into small groups of three and asked them to collaborate on completing a Check for Understanding quiz via your GeoGebra; an interactive question that provides a diagram of a circle with blank labels and requires students to drag and drop the proper labels in place from an answer key.
- Students will compare the different circle theorem properties and make conclusions regarding the differences they find.
- Students will be asked to analyse the connections between different circle theorem. Each group will be asked to infer what might happen when different circle properties are considered. Each group will compile evidence to make their case.

Appendix 2: Activity Sequence:

Lesson 1

Learning objective: To learn by investigations;

- Task 1: The angle at the centre of a circle is twice the angle at the circumference of the circle from the same arc.
- Task 2: The angle formed in a semicircle is always a right angle. (Task 2)
- Task 3: Angles from the same arc in the same segment are equal. (Task 3)

Whole-class starter activity: Project Geogebra on the IWB to emphasize on the general functions of the Geogebra – (Students to work through starter activities in pairs) -15 minutes Class work: Task 1, Task 2 and Task 3 - 30 minutes

Feedback: Whole-class discussions on completed activities with misconceptions resolutions – 15 minutes

Extension: - Task 4: Opposite angles of a cyclic quadrilateral add up to 180^o.

Strategies and activities:

Teacher: Walk through the circle theorems and the basic properties of circle, noting the diagram in pupil's exercise book

Activities: Break the class into small groups. Task each group with labelling their own diagram of circle and researching a single theorem to present to the class later on. Ask each group to present the circle theorem on the IWB to the class.

Lesson 2

Learning objective: To learn by investigations;

- Opposite angles of a cyclic quadrilateral add up to 180⁰. (Task 4)
- Two tangents drawn from a point to a circle are equal. (Task 5)
- The angle between a tangent and a chord is equal to the angle at the circumference in the alternate segment. (Task 6)

Starter activity:

Project Geogebra on the IWB to consolidate the key operations of the Geogebra functions -15 mins

Students to work through starter activities in pairs – 15 minutes

Whole class work: 30 minutes (Task 4, Task 5, Task 6)

Extension: - Work sheet: To investigate; (The perpendicular line from the centre of a circle, chord and the bisection to the chord. The angle between a tangent and the radius, at the point where the tangent touches the circle. (Is it a right angle?)

Feedback: Whole-class discussions on the completed activities Major activities

Teacher: Walk through the circle theorems and the basic properties of circle, noting the diagram in the textbook

Activities: Break the class into small groups. Task each group with labelling their own diagram of circle and researching on the circle theorems to present to the class later on

Appendix 3: Task: Circle Theorem

Task 1: The angle at the centre of a circle is twice the angle at the circumference of the circle from the same arc.



Figure 1: Visualizing the inscribed angle theorem.

Task 2: Angles inscribed in a semicircle are right angles



Figure 2. Visualizing the Angles inscribed in a semicircle are right angles theorem

Task 3: Angles in the same segment are equal.



Figure 3: Visualizing the Angles in the same segment are equal.



Task 4: Cyclic Quadrilateral Theorem



Figure 4: Visualizing the Cyclic Quadrilateral Theorem



Task 5: Alternate Segment Theorem

Moving point A changes the position of the tangents. Moving points O or X changes the size of the circle.

Figure 5: Visualizing the Alternate Segment Theorem

Extension work:

Task 6. Perpendicular & chord

A perpendicular line from the centre, O, cuts the chord CD of the circle at point B.

So OB is perpendicular to the chord CD.

Move C or D, and note the connection between lengths CB and DB.

If you want to alter the size of the circle, move O or X.

Task 7. Angle between tangent and radius

What do you notice about the angle between tangent and radius, $\alpha \& \beta$? You can use the cursor to move point C, and you can change the size of the circle by moving point O or X.

Misconceptions:

Adapted from: (https://danpearcymaths.wordpress.com/2019/09/20/flipped-classroomcircle-theorems-and-common-misconceptions/)

The angle at the circumference is half the angle at the centre.

4 separate cases to consider understanding the misconceptions



Case 1







Case 1: Often presented in the textbooks and easier to understand by students

Case 2 Angles on the circumference sharing the same arc are equal. Most students cannot identify the two similar triangles when they associate the shape to the bow-tie angle theorem as seem below. Here the triangles are not similar.

Also it is difficult for students to tell which angles are equal. Because they fail to realise that when the triangles are similar the angles which lie on the same **arc** will be equal as seen below.



What are the misconceptions with the cyclic quadrilateral theorem?

Students make a mistake because the opposite angles in the quadrilateral does not always have to add up to 180 degrees. It is only possible when all the vertices of quadrilaterals lie on the circumference of the circle.

Appendix 4: Case study 2

Lesson plan 1

Introduction

Title: Demonstration of drawing lines

Recape: Review the aim of the lesson Awareness of purpose; Use GeoGebra to draw basic Lines and find the Equation of lines- Time 15minutes

Demonstrate the misconception in drawing lines and graphs: identifying errors; overt subject knowledge; theoretical underpinning of pedagogy; use of terminology; Class textbook to understand example on functions: use of textbook;

Class work: Practical Exercise: Controlling Characters with Graphs: reliance on procedures

Corner stone Activity 1: Coordinating algebraic, graphical, and tabular representations

Making connections between procedures;

making connections between concepts; anticipation of complexity; decisions about sequencing; recognition about conceptual appropriateness.

Speed as a context to introduce rates of change y= mx+c as a model of constant velocity motion – the meaning of m and c in the motion context

Choice of representations and explanations;

choice of examples, teacher demonstrations

Lesson plan 2

Review of previous lesson

Discuss Key words: use of terminology;

Lesson objectives: Graphing Linear Functions and finding the relations between tables and algebraic representations. Awareness of purpose;

identifying errors; overt subject knowledge; theoretical underpinning of pedagogy;

Demonstration: Creating a y-y1=m(x-x1) graph on GeoGebra (15minutes)

Completing examples from text book: use of textbook; reliance on procedures

Practice 2

Graphing Linear Functions with GeoGebra (10 mins)

- Awareness of purpose;

Activity2: Money remaining and the days left for the holiday

choice of examples, teacher demonstrations

Responding to children's ideas; deviation from agenda;

swift and correct analysis of student errors and difficulties

Creating a linear sequence in GeoGebra (using spreadsheets)-10 minutes

Demonstrate the relations between functions, tables and symbolic representations.

Choice of representations and explanations;

choice of examples, teacher demonstrations

use of opportunities;

Appendix 5: Coding using TPACK

Section 1: Teacher Lesson Plans for TPACK

The following themes have emerged:

Linear functions Overview of unit

Key to Knowledge Quartet codes		
Foundation, Fo AtT – adheres to textbook AP – awareness of purpose CoP – concentrates on procedures IE – identifying errors OSK – overt subject knowledge TUP – theoretical underpinning UT – use of terminology	Transformation, Tr CUE – choice and use of examples CUR – choice and use of representation TD – teacher demonstration UIM – use of instructional materials	
Connection, Co AC – anticipation of complexity DS – decisions about sequencing MCC – making connections between concepts MCP – making connections between procedures RCA – recognising conceptual appropriateness	Contingency, Cy DA – deviation from agenda RCI – responding to children's ideas UO – use of opportunities TI – teacher insight RAT – responding to (un)availability of tools and resources	
Practice 2		
Amount of money spent on vacation		
Writing an equation		
Money remaining and the days left for the holiday can be		
represented as a straight line		
View setting Drag and drop the graph		
Set specific windows		

Using function equations	Fo/AP- awareness of purpose
Improve the view using different colures	
Finding the interceptions of lines	Fo/UT-use of technology
Click on points and the segment	
Label axis	Tr/TD- teacher demonstration
Use the manual box to label axis	
Identifying the money remaining by labelling the axis	Tr/CUR-use of representation
Using the export function in GeoGebra	Co (MCC) making
Using copy and paste functions	Co/MCC- making
Controlling Characters with Graphs	
Graphs are mathematical representations of relationships	concepts
such as savings	Cv/TI- teachers insight
Play and pause a simulation.	
For graphs of savings (that is, money versus days), the	
steeper the line, the lesser the money savings.	
Demonstrate to students how to modify the graph to	
change the savings.	
The amount of savings can be determined from different	
parts of a graph and model.	
Adjust the graph to change the final position.	
Representing the Equations in the form of Graphs and	
tables.	
Play and pause a simulation.	
Writing equations based on tables or graphs.	
Modify the graph to change the savings.	
Translate between tables, graphs and equations.	
Demonstrate how to adjust the graph to change the final	
position.	
Representing days, money, holidays and savings	
differently in three demonstrations.	
Play and pause a simulation.	

For equations of the form y = mx + c, Saving y and money
and days respectively
Modify the graph to change the savings.
Demonstrate how to translate between graphs, tables
and algebraic expressions.
Adjust the graph to change the final position.

Section 2: lesson plan with Knowledge of Quartet

Foundation	Awareness of purpose; identifying errors; overt subject knowledge; theoretical underpinning of pedagogy; <mark>use of terminology</mark> ; use of textbook; <mark>reliance on procedures.</mark>
Transformation	Choice of representations and explanations; choice of examples, teacher demonstrations
Connection	Making connections between procedures; making connections between concepts; anticipation of complexity; decisions about sequencing; recognition about conceptual appropriateness.
Contingency	Responding to children's ideas; use of opportunities; deviation from agenda; swift and correct analysis of student errors and difficulties

Appendix 6: Snapshot from lessons

INTRODUCTION TO GRAPH DRAWING WITH GEOGEBRA
😨 GeuGebra							
File Edit View Options To	ois Window Help						
Move Drag or select objects (Esc)							
📕 Free Objects 🛛 🗡	Toolbar		Toolbar		A	В	
Dependent Objects	4-			1		1	<u>^</u>
			негр	2			
				3			
	5			4			
Algebra view				5	Spread	dsheet	
	2-			6	View		
				7			
	1 -			8			
		Graphics View		9			
	- <u> </u>		<u> </u>	10			
	-2 -1	0 1 2 5	4 5 5	11			
	-1 -			12			
				13			-
					 ■ 		F
🕐 Input:	Input Ba	r	• •	α -	Command .		-

Drawing graphs

The procedure is similar for all types of functions. First write the desired function expression in the input bar. Then, change the axes so that the graph show up in the graphics view.

We will now draw the graphs of the to following functions:

1. The straight line given by f(x)=3x+1

1. The straight line f(x)=3x+1

In the input bar, type f(x)=3x+1.

Input: f(x)=3x+1

f(x) is the name of the function f of the independent x. We can of course provide other names for the function like g(x), h(x), k(x). Note that we do not need the multiplication sign between 3 and x.

Press enter and the graph of the function will appear:



Adjusting the x- and y-axis: To get the best possible picture of the graph, we need to adjust the x- and y-axis. For this we use the Move Drawing Pad: File Edit View Options Tools Window Help

F	ie Ea	it view Op	tions loois	vvindow He	eip		
	₽][•	•] [~] [?	() [>] [•]] 🖸 🛋	a=2	+	Move Drawing Pad:

Press this button so that it is highlighted. Move the cursor so that it lies directly over one of the axes. Press left mouse button and drag the axis to change the units. Try to make the drawing pad look like this:

	8															
	6															
	4															
	2															
f -0.4	-0.2	0	0.2	0.4	0.6	0.8	1	1.2	1.4	1.6	1.8	2	2.2	2.4	2.6	2.8

Editing in the graphic view

	Free Objects	*	10				
	Dependent Objects		10	Drawing Pad			
			8	⊥ Axes ⊞ Grid			
			6	Zoom xAxis : yAxis	:		
			4	Show All Objects Standard View			
			2	B Drawing Pad			
Rig	ght click on an o oises on alread	object y drav	in the g	graphic <mark>vie</mark> w ar cts:	nd choose Object P	ropertie	es for more eo
Rig	ght click on an o oises on alread	object y drav	in the g vn objec	graphic view ar cts:	nd choose Object F	ropertie	es for more eo
Rig	ght click on an o oises on alread	object y drav	in the g vn obje	graphic view ar cts:	Basic Colour Style Algebra Advanced	Propertie	es for more ec
Rig	ght click on an o oises on alread	object y drav	in the g vn objed	graphic view ar cts:	Basic Colour Style Algebra Advanced Name: f Value: 3x+1	Propertie	es for more ed
Rig	Function f	object y drav	in the g ∨n obje¢	graphic view ar cts:	Paris Corose Object F		es for more ed
Rig cho	Function f Show Object Show Object Trace On Copy to Input Bar	object y drav	in the g vn objec	graphic view arcts:	Contain State Agence Advanced Norm: Contain State Agence Advanced Norm: Contain State Contain Contain State		es for more ed
Rig cho	Function f Show Object Show Object Trace On Copy to Input Bar Rename Delete	y drav	in the g ∨n objec	araphic view ar cts:	Basic Cotour Style Algebre Advanced Name: f Value: 3 × 1 Value: 3 × 1 Show Copet Show Copet Show Tase Fire Capet Ausliary Olget	Propertie	es for more ex
Rig cho	Punction f Function f Show Object Show Label Trace On Copy to Input Bar Copy to Input Bar Detete Sobject Properties	object y drav	in the g vn objec	graphic view ar cts:	Bank Catalar Bate Agentra Advanced Narra: // Value: 3 + 11 Capino: Bank Capest Bank Capest Bank Capest Bank Capest Advance Bank Capest Advance Advan		es for more ed

Linear Functions- Lesson and Exercise

Aims: 1. To understand the concept of Slope

2. Calculate gradients and comment on the nature of slope

Starter lesson:



The ten percent slope in the road signs above mean that if you move 100 m along the horizontal direction, then you move at an average 10 m along the vertical direction.



In order to find the percentage slope, you divide the movement along the *y*-axis with the movement along the *x*-axis, 10100=0.1=10%.

The road signs above assume that you drive from the left to the right, from the look of the pictures you can deduce whether it is a slope downwards or upwards.

When defining slope in mathematics, the definition is similar to the slopes on the road signs. The slope is defined to be the difference along the *y*-axis divided by the difference along the *x*-axis. In order to distinguish between a slope downwards and a slope upwards, the difference along the *y*-axis is counted as negative if the slope is downwards when going from the left to the right.



Instead of the words "difference along", the symbol Δ is used. The difference along the yaxis is denoted by Δ y and the difference along the x-axis by Δ x. If m denotes the slope, we get following formula: m= Δ y/ Δ x

Exercises

Exercise 1 - Slope by definition

Start a new GeoGebra sheet. Right-click on the drawing pad and check both Axes and Grid.

Choose Options->Point Capturing->On (Grid).

Use the tool **Line through Two Points** *it* to make four lines through the points:

- 1. (-1,1) and (3,4)
- 2. (0,-1) and (4,-1)
- 3. (2,-2) and (-2,2)
- 4. (1,1) and (1,4)

Find the slope of the four lines by using the definition $m=\Delta y\Delta x$ and write down your answers!

Exercise 2 - Using the tool Slope

Use the tool **Slope** \checkmark on the four lines. Click on the tool and then on a line.

Compare the GeoGebra-slopes to your answers.

Comment

The GeoGebra-way of defining slope could be seen as an alternative definition:

If the change along the x-axes (moving from left to right) is 1,

then the change along the y-axes is the slope.

Load GeoGebra worksheet



Slope

Exercise 3

Start a new GeoGebra sheet. Make sure that the input bar is shown; if not, pick View->Input Bar.

Enter following three functions in the input bar (one at the time):

- y=x
- y=x+2
- y=x-1

Without plotting the graph, answer these questions about the function y=x+4.

- 1. What is the slope?
- 2. What is the *y*-intercept?
- 3. What is the *x*-intercept?

Exercise 4

Delete the three functions and instead enter following functions in the input bar:

- y=-2x
- y=-2x+3
- y=-2x-2

Without plotting the graph, answer these questions about the function y=-2x+1.

- 1. What is the slope?
- 2. What is the *y*-intercept?
- 3. What is the *x*-intercept?

Exercise 5



Delete the three functions and enter the equation y=2x in the input bar.

The equation is shown in the algebra view and the line is shown in the drawing pad.

Use the mouse to drag the line and watch the equation in the algebra view. The second number shown in the expression, changes as you moves the graph. Describe how you can find the second number of the expression by looking at the graph!

Exercise 6

Without drawing a line, answer these questions about the line defined by y=3x-2.

- 1. What is the slope?
- 2. What is the *y*-intercept?

Comment

GeoGebra distinguishes between functions and equations. Functions are written like this: f(x),f1(x),g(x),

Main Function (x),... Equations are written using the variables x and y. Each equation is given a unique name, but all equations use the variables x and y.

```
Algebra

Free Objects
a: y = 2x + 4.54
b: y = 5
c: x = 3
Dependent Objects
three equations with the names
a, b, and c
```

An equation can be used to define a line like this:

Only the points whose coordinates x and y satisfy the equation

(makes the equality true),

lie on the line

A longer explanation is given on next page.

Gradients of Perpendicular Lines

- Draw a line through points A and B.
- Draw a line perpendicular to the first line through the point A.
- Use the *A*Slope tool on both lines.
- Rename the gradients to m₁ and m₂ respectively.



• Make a new variable as in the picture below. Move the points!



• Make a conjecture! Prove it!

Hint: Look at the picture and use similar triangle.

reference: the pictures of the road signs are

from: http://commons.wikimedia.org/wiki/Category:Slope_signs

Source of lesson: http://www.malinc.se/math/functions/linesasfunctionsen.php

Appendix 7: Interview scripts -students

Knowledge Quartet emphases questions

Question 1:

What do you understand by a function as a rule

Response:

A function assigns to each input exactly one output. The graph of a function is the set of ordered pairs consisting of an input and the corresponding output

Question 2:

Give me an example of why a function is a rule?

Response:

For example, the rule that takes *x* as input and gives 5x+4 as output is a function. Using y to stand for the output we can represent this function with the equation y = 5x+4, and the graph of the equation is the graph of the function will be a straight line.

Question 3

What do you understand by linear function?

Response:

Is a rule that assigns to each input exactly one output. The graph of a function is the set of ordered pairs consisting of an input and the corresponding output.

Question 4

Can you give me an example of a function?

Response:

For example, the rule that takes *x* as input and gives 5x+4 as output is a function. Using y to stand for the output we can represent this function with the equation y = 5x+4, and the graph of the equation is the graph of the function.

Question 5

How would you Interpret the equation y = mx + b as defining a linear function, whose graph is a straight line; give examples of functions that are not linear.

Response

For example, the function $A = s^2$ giving the area of a square as a function of its side length is not linear because its graph contains the points (1,1), (2,4) and (3,9), which are not on a straight line

Question 6

Can you determine which of the functions listed below are linear and which are not linear and explain your reasoning.

o $y = -2x^2 + 3$	non-linear because it has a square
o $y = 2x$	linear, because it has 2 as gradient and 0 intercept
	153

o $A = \pi r^2$ non-linear, nonlinear function, because it has a square o y = 0.25 + 0.5(x - 2) linear

TPACK Framework emphasised

Question 1: How is your understanding in linear functions enhanced by the teachers use of Geogebra

Response(S1); I am able to complete more work and stay focused throughout the lesson

Response (S2); I am able to see and observe what am doing so my interest is sustain throughout the learning

Response (S3): The use of GeoGebra assisted me to understand the work better because I can drag the graphs and understand what is happening

Response (S4): I am to follow what the teacher is teaching so I don't drift in my learning. This made me to follow and stay with the lesson. I like the pace of the lesson

Response (S5): I am able to practice more work and if am stuck I able to get help quickly and then I move on with my learning. The teacher is able to demonstrate and show me how to overcome my difficulties. The pace of the lesson is good and not boring.

Question 2: What do you like most when the teacher was teaching with GeoGebra?

Response (S1): using the software made my work so nice and neat, it was very user friendly since if I made a mistake, am able to erase my work and do it again. Something I cannot do when learning with pen and paper.

Response (S2): It saves time when you are using GeoGebra to do your class work.

Response (S3): The teacher is able to give clear instruction on what we have to do without sometimes talking too much

Response (S4):The teachers is able to walk around the class and support pupils having troubles and also provide extension work for the quick students.

Response (S5): The teacher is able to use different questions to access my learning ability on the class work.

Question 3: Would you recommend and be happy for the teacher to use GeoGebra to teach all the topics?

Response (S1): Yes, because I think the use of GeoGebra made the learning more fun and everyone in the class was paying attention to the teacher

Response (S2): learning with the software made the learning visual for me so I would like it to be used to present all topics in most lessons

Response (S3): Although I had some problems in starting with the work, my friends helped me to overcome my problems. I think it will be good to use it in all lessons to help children understand the learning better.

Response (S4): Using the software made the lesson very challenging for me. I was able to do most of the work alone by myself once I had been given clear <u>instructions and</u> <u>demonstrations</u> by the teacher. I think it will be ok for me for the software to be used always in my lessons.

Response(S5): I like to use GeoGebra in my all my lessons because I think teachers can give us homework and everyone will be able to do their homework at home if you have a computer since everyone like working on the computers.

Appendix 8: Letter to Head teacher

Dear Sir, I would like your permission to conduct a research project in the school for my dissertation as part of my PhD in Mathematics Education.

The title of the research: The impact of ICT on Mathematics education

The aim of the research: To explore my own classroom practice when teaching and learning linear functions using digital technologies. The focus of the research is on my interaction with the students' mathematical activity with digital technologies.

My year 9 maths group will be involved in the research study. By taking part in this research, the school will discover new approaches to the teaching of linear functions using digital technology. 10 students who have consistently demonstrated interest in integrated technology learning will be approached and provided with an information

sheet explaining the purpose of this research. They will also be given a consent form for their parents to sign to show they are happy for their child to participate in the project. All students and parents will be informed about the expectations of the research.

The obligation expected from the school is to support me and permit me to use my year 9 maths group to take part in the study during one week starting in the first week of September, 2019. As part of the study, students will be recorded completing their class work in the maths lesson. All participants will be granted anonymity in the research reports. I will keep all the data collected strictly confidential and only make it available to the designated supervisor and myself.

If permission is granted to proceed with the project, I will send a letter to parents informing them about the project and their right to withdraw their child's data if they wish to.

Please contact me if you need further information about my study in the school.

Many thanks for your cooperation as usual and I look forward to hearing from you soon.

Yours sincerely,

D. Ata-Baah

Appendix 9: Letter to Teacher

1st September 2019

Dear Parent/Guardian

As part of my final module to complete my PHD Degree in Maths Education, I am undertaking a research project into how students learn linear functions using dynamic technology. I am delighted to inform you that your son/daughter is among six students selected to take part in a research project due to his/her interest in learning with technology and his/her articulated expressive reasoning in mathematics.

The research will take place in his/her normal teaching group and does not require any changes to his/her normal learning or timetable. During their normal lesson with me, I will observe how the pupils interact with technology when learning linear functions in mathematics. All participants will be invited to discuss their reasoning during the session

with me at the end of the lesson. I will also ask them to complete a task that will require written responses.

Also after the session, students will be invited for a short interview to discuss their learning with the technology. Two lessons are planned, each lasting for about 60 minutes with 10 minutes of individual interviews during the lunch break. The interviews will not have any impact on their lunch break since pupils will be allocated extra time for their lunch.

The data collected from pupils' responses together with the response from the interviews will form the basis for the research. All pupils will remain anonymous and your son's/daughter's details will not be disclosed to a third party without your consent. All parents will have the right to withdraw their child's data from the project at any stage of the research.

This is an exciting opportunity for your child to take part in the research since I believe the experience will be enriching and worthwhile to develop pupils' deeper understanding of mathematics.

Please complete and sign the paper below to let me know whether you would like your son/daughter to participate in the research and return it to me by 30th September,2019. I will assume that you have consented to your child taking part in the research if I do not hear from you by then.

Many thanks.

D. Ata-Baah

Head and Teacher of Mathematics
Pupils Name----I do agree...
I do not agree that my son/daughter can participate and be interviewed
(Please tick whichever applies)
Parent Name: ------Date: -----

Appendix 10: Consent form

Title: What is the impact of ICT on mathematics education?

(Using GeoGebra to teaching linear functions)

01.09.2019-01.10.2019

Please complete this consent form and return to <u>atabee@hotmail.com</u> by 30.08.2019.

I am consenting to this element of the study by initialling/ ticking each box below Any unticked box will be assumed that I DO NOT consent to that part of the study. I understand that by not giving consent for any one element that I may be deemed ineligible for the study.

I agree to take part in this research,	Yes	No
I am free to withdraw at any point without giving a reason		
I agree that my name will be protected in the publication of any report		
I am happy for my interview to be audio		
I agree that the data will be collected and processed in accordance with the		
University's Data Protection Policy.		
Name		

 Signed _____
 Date _____

Researcher's name: Daniel Ata-Baah

Signed _____

Appendix 11: Participation Consent Form

Title of Research Project:

What is the impact of integrating ICT on mathematics education?

Brief Description of Research Project,

The purpose of the project is to find out the impact of teachers on the use of ICT in teaching and learning mathematics. This research is carried out in fulfilment of my PhD degree in Mathematics education dissertation.

What Participation Involves.

10 participants will be selected for this project. I will conduct a semi structured interviews over a period of two weeks and then use about a month to analyse my data. The findings will be presented in a report. Each interview will last for about 15 minutes.

Limits on confidentiality

The interview will be tape recorded, and transcribed with any identifying details removed. The transcript, or extracts from, may appear in my report and in publications arising from it. The tapes may be heard by my supervisor and others who might be involved in examining the report.

Everything you say will be treated confidentiality, but there is a limit to this: if you disclose a risk of serious harm then I may need to take appropriate action (this adheres to the ethical guidelines of the).

Student Investigator Contact Details:

Name: Daniel Ata-Baah,

Email: Telephone

Consent Statement:

I agree to take part in this research, and am aware that I am free to withdraw at any point without giving a reason, although if I do so I understand that my data might still be used in a collated form. I understand that the information I provide will be treated in confidence by the investigator and that my identity will be protected in the publication of any findings, and that data will be collected and processed in accordance with the Data Protection Act 1998 and with the University's Data Protection Policy.

Name	Signature	Date -

Please note: if you have a concern about any aspect of your participation or any other queries please raise this with the investigator or the Supervisor. However, if you would like to contact an independent party please contact the Head of Department.

Supervisor Contact Details:	Head of Department Contact Details		
Name:	Name		
University Address:	University Address:		
Email:	Email:		
Telephone:	Telephone:		

Appendix 12: Participant Debrief

Participant Number: _____

Title of Research Project:

What is the impact of ICT on Mathematics education?

Thank you very much for taking part in our study, we greatly appreciate your contribution.

This study is designed to examine the factors that might improve the integration of ICT by subject teacher and your participation is extremely valuable.

All data gathered during this study will be held securely and anonymously. If you wish to withdraw from the study, contact us with your participant number (above) and your information will be deleted from our files.

Should you have any concern about any aspect of your participation in this study, please raise it with the investigator in the first instance or with the Project Supervisor or Head of Department.

Investigator	Project Supervisor	Head of Department
-	• -	_

Daniel Ata-Baah	Name	Name	
If you are a student at Roeham	pton University and ar	e troubled or worried about any as	spect
of the study, or issues it may h	nave raised, you may fi	nd it helpful to contact one of the	
following who will be able to	advise you on agencies	s that can deal with your particular	•
concern:			

Student Welfare Officers:

.....

If you feel your concerns are more serious or complex you may wish to contact your General practitioner.

If you are a non-student you may find it helpful to contact your GP or one of the following who will be able to advise you on agencies that can deal with your particular concern:

[Agree with supervisor on appropriate agencies]

Appendix 13: Health & Safety Risk Assessment

Health & Safety Risk Assessment for Ethics Applications

RISK ASSESSMENT FORM						
Risk Assessment H	Reference Number					
Title of Project	What is the impact of ICT in math	ematics educati	ion?			
Description of	• Interviews	Date	From: September 2020			
activity	• Recording of interviews		Until:,			
Area/Locations	London					

Project team	Name	:		Job Title	Signed
Hazards	<u> </u>	1.Non	e		
		2.			
		3.			
		Etc.			
Who can be harm	ed?	1. Nor	ne		
		2.			
		Etc.			
How can someone	e be	1. Nor	ne		
harmed?		2.			
		Etc.			
Number of people	<u>.</u>		Rate	H=Hourly, D=D	aily. W=Weekly. M=Monthly.
affected				Q=Quarterly, S=	=Six monthly, A=Annually
Consequence		1. Nor	ne	·	
		2.			
		.Etc.			
Existing Control		1. Nor	ne		
Measures		2.			
		Etc.			

Comments								
Risk rating		VH=Very High, H=High, M=Medium, L=Low, VL=Very Low						
Further possible	None							
control measures								
Any further actions required								
Responsible person	Descriptio	tion of Details of		Date	C	Completed		
	hazard		action taken					
The signatures below confirms that a meeting/discussion has taken place, the Hazard, Risks and appropriate Control measures outlined above have been read and understood.								
Signed (Applicant)			Print Name	Da	niel Ata-Baah			
							Date	
Student Application	5		Print Name				Date	
Signed								
(Supervisor/PI)								
Staff Applications			Print Name				Date	
Signed (Peer								
Review)								

Appendix 14: Interviews and Documentary Analysis Schedules

Dissertation title: What is the impact of ICT on Mathematics education?

Key Questions	Prompts and Probes
Part 1: Beliefs on ICT	Is the use of ICT integration a necessity in your
1. What do you believe to be the benefits of	mathematics? Why do you say that?
using ICT in your subject area?	
2. What do you believe are the	
Obstacles/barriers to integrating ICT into your	
Mathematics pedagogy?	
Part 2 ICT Facilitators	How do you think the use of ICT can be
	supported the teaching of mathematics ?
3. What do you perceive are the facilitators to	supported the teaching of mathematics :
integrating ICT into your department's	
pedagogy?	What Strategies for using ICTs would change
4. What are the specific tools used to facilitate	pedagogical practices.
ICT integration in mathematics?	
Part 3: The impact of ICT	Comment on your personal ICT experience in
5. How does your experience of using ICT	your pedagogy?
impact the use of ICT in your teaching?	Can you mention your current implementations
6. What is the impact of ICT on learning and	of ICT tools in your teaching?
achievement ?	Do you see any improvement in pupils learning
	using ICT?
Part 4: Support to use ICT	What do you think are the barriers to ICT
	integration in your mathematics?
7. What on-going professional development is	integration in your mathematics:
available to support the use of ICT?	
8. What do you perceive to be the effective	
strategies for using ICT to change pedagogical	
practices?	

Part 5: Leadership role on	Do you think the leadership in your school				
integrated ICT	support the use of ICT in teaching				
9: How do you view whole school leadership	mathematics?				
on ICT integrated learning?	How do you feel about the leadership within				
10: How would you view your leadership role on integrated ICT pedagogy in your department?	your department on ICT integration? Are teachers supportive?				

Appendix 15: Interview Transcript 1 – Teacher 1

00:00-00:14

Interviewer: This interview is in fulfilment of my Phd degree programme.

The tile of the research is 'What are the impact on ICT in Mathematics education?

00:15-00:20

Interviewer: You can always pull out of the interview if you don't feel comfortable.

00:21-00:21

Respondent: Ok.

00:21-00:27

Interviewer: Any information that I gather here will not be used externally without your permission.

00:2-00:28

Respondent: Ok.

00:28-00:56

Interviewer: Question: what do you believe to be the benefits of using ICT in in maths?

00:59-01:55

Respondent: I think that it brings life into the lesson. The development of concepts can be quite boring for some students, especially when it comes to reading and writing. I think, especially the internet has a lot of potential for young people to be engaged in different ways in the classroom and also it makes the learning more fun...

01:56-02:09

Interviewer: Question 2 Alright, very good. Now what do you believe are the obstacles, or the barriers, the problems to integrating ICT into maths pedagogy?

02:12-03:09

Respondent: I suppose that it would always depend on the flexibility of the curriculum and how the curriculum can be approached because, well, there are certain topics and certain concepts that have to be covered and, I think that maybe for the sake of following the curriculum it is impossible to integrate ICT. Also sometimes there is lack of resources. It will be great if the school can provide us the resources that we wasting time looking for resources. 03:09–03:03

Interviewer: Is there any more problems that you can think of?

03:09-03:11

Respondent: Yeah, exactly, the Hardware and software problem makes it difficult to use ICT with poor internet problems. It is very frustrating using the internet with slow internet connections.

04:12-04:27

Interviewer: Ok, now, question 3: what do you believe or perceive are the facilitators to integrating ICT into maths pedagogy?

04:31-05:30

Respondent: I think that the best way to facilitate ICT integration would be for the teachers to be ICT literate. I think that most teachers ultimately do not have the knowledge; so on the job training will help to facilitate ICT integration.

05:30-05:34

Interviewer: How does your experience in using ICT impact on your ICT use in your teaching? Your ... I'm talking about your own personal experience in using ICT. 05:59–06:30

Respondent: Well, when it comes to teaching, in particular, I think that there are some videos, films, adaptations of books that sometimes help a lot to the children to understand the background or put into context all the things that we imagine when we're reading, so I think for me, II think my good experience in using ICT has helped me to integrate ICT tools into my lessons. I like to use ICT to promote visual learning through the interactive white boards.

06:30-06:32

Interviewer: Ok, good. Question what is the impact of ICT on learning and achievement? Do you think the use of ICT brings about higher achievement in learning?

08:22-09:28

Respondent: I couldn't say for sure because I think that it makes a great contribution into engaging students better and I think that the interactive nature of ICT tools makes it more enjoyable and promotes independence.

09:28-09:39

Interviewer: Ok, question: what ongoing professional development that you know is available to support you to use of ICT?

09:44-10:20

Respondent: Well, at school we have ICT person that is in charge of smart boards and computers and we have, online safety course organised by external agencies.

10:20-10:20

Interviewer

What do you believe to be the support on ICT integrating in your school? 12: 00-12:30

There is no support for staff. Because guess when you are struggling and want some help, you will find it difficult to get the needed training since most of the IT personals may be working on other projects for the school or sometimes supporting children with passwords and trivial IT issues with teachers such as the white board is not working, classroom speakers are not working and so not any major training takes place. There is no external support as well and I can say that it is a struggle with most teachers.

Respondent: I think that there should be a strong school policy on in-service teacher training initiatives to ensure that ICT is integrated in every lesson. I mean unfortunately for teachers as well as leaders, this is something very recurrent I think. It's like the training never stops - we're always trying to catch up with something that we need to be, you know, up to date with. So I would say that training is everything... is key to incorporating this, I know other strategies will include changing the culture of the school by giving incentives to teachers who regularly integrate ICT in their lessons will help.

14:04-14:09

Interviewer: Are there any habits or attitudes or perceptions that need to be changed?

14:07-15:04

Respondent: Yeah. Yeah ... yeah ... yeah, and also there are so many constraints and so many pressures, from ... from ... from a budget perspective, for the coming years. I don't know what is going to happen with there are Ofsted visits, I mean schools that ... that need to ... that are, you know, that have that cloud above their head waiting for the call and praying for a Wednesday to cancel they didn't receive the call and they're now being Ofsted and there is a lot of pressure there and I think that there is such a focus on achievement and on league tables and being at the top and advancing x number of levels per year and we lose that creative part, that ... the ... the freedom to ... to try to explore in different ways of teaching that could be more enjoyable but they are maybe less accountable for ...

15:04-15:24

Interviewer: That's very good, very good. Ok how do you perceive ICT integrated learning in your lessons? Do you think it's very necessary to promote it, is it a necessity, do you think it makes no difference or ...?

16:59-17:24

Interviewer: Now, how do you view your leadership role for ... on integrating ICT pedagogy in your department?

17:24-18:54

Respondent: I suppose that there's a lot to be said about example setting and examples. I think the leader should lead by example. is being an example and being a team player as well, so trying to bring out the best of everyone involved, so I think that the fact that I am a very ICT person ... I use apps for pretty much everything in my daily life ... makes it more believable when I say like, well, we can use this to enhance that or this activity we could incorporate the tablet for this or dictionaries for that or whatever. I think that in this particular case, if you're not an example, then you're not a true leader. Yeah.

18:54-18:56

Interviewer: Alright. Thank you very much, is there anything you would like to add?

Appendix 16: Interview Transcript 2 – Teacher 2

00:00-00:31

Interviewer: These are questions for my Phd degree in using ICT in mathematics

Education. You are free to pull out of this interview at any point in time No information from you will be given to a third party without your permission.

Ok, so what do you believe to be the benefits of using ICT in teaching maths?

00:32-00:54

Respondent: It's interactive side, mainly. I'm as a teacher, I use a lot of ICT to teach maths, purely because its interactive for the young children, you can make it as animated as possible for the children, in particular. You can change a lot of things, so it's not like books and stuff where every-, all the information is static, its actually things you can change when and if you want to, so it's really, really flexible.

00:56 - 01:04

Interviewer: Alright, what do you believe are the obstacles or barriers to integrating ICT in teaching maths?

01:05-01:18

Respondent: Yeah, I think cost is one of the ones, secondly its reliability. You know if the system fails or internet fails or power fails, then you have those issues. So, yeah, my main two ones would be ... would be cost and reliability.

01:19 - 01:29

Interviewer: Right, well, what do you believe or perceive as the facilitators to integrating ICT in teaching maths?

01:37-02:00

Respondent: Yeah, I think generally make it more widespread. I know my school is using it a lot more and, so I think if everyone gets on the same platform and understands the benefits of using this, it would really help all the the children. So I think just making sure everyone's aware why we use ICT, and the benefits of it, and once everyone understands that it'll be a lot ... lot clearer to everyone to be supported.

02:00 - 02:17

Interviewer: Ok, so do you think your expertise or knowledge in ICT would really be a facilitator in you helping to use ICT in your lessons?

02:17-02:43

Respondent: Yeah, of course, I think ... I think everyone one needs to have a certain degree of ICT skills to be able to teach the children. I think with ICT being advanced in so many ways, you know using a computer or using a tablet, you know everyone has to have a certain degree of skills, so I think having that basic knowledge at least in ICT you can then help children develop their skills. And with ICT you're always learning, there's always things like upgrades that come along and ... and your various development. I must inform you that am not a fun of training provided by external consultants. We have not been lucky with this due to lack of follow ups.

02:44 - 02:50

Interviewer: Good. Thank you. How does your experience, lets say your background impact the use of ICT?

02:50-03:14

Respondent: Yeah so my interest in ICT in the past is crucial, I think my past good experience helps to use ICT. Of cause my background as an IT consultant has made me appreciate the need to use ICT in most of teaching. This motivates me to plan, teach and always find ways to incorporate ICT.

04:47 - 04:55

Interviewer: Ok, is there any professional development that you think you would ... you'll benefit from, to help you to use ICT?

04:58-05:23

Respondent: Yeah, I mean I myself attend a lot of trainings which involve ICT and to do with my maths teaching and I think that will benefit everyone, I think for all, you know, children are using computers from a very young age and some adults are barely using computers. So, I think it's very important that they also keep up with ... with technology to then keep up with their children and it will help their teaching aspect of the children as well.

05:54 - 06:08

Interviewer: Ok, now, how do you view the role of ICT integrated leadership on learning? Is it very, very important, in your view to integrate ICT ...? Is it a necessity or can ... is it like a luxury? Can the school just do without it or does it make any difference at all?

06:08-06:18

Respondent: I think it's a necessity to have a strong leadership on ICT in the school and I think it's a great tool for school to improve student achievements with a strong leader who will develop the entire school to integrate ICT.

06:18 - 06:35

Interviewer: Ok. Thank you, the last question now. How would you view your leadership role on integrated ICT pedagogy in the maths department? I mean your leadership skills, how do you think will impact the teachers in the department to be encouraged to use ICT?

06:35-06:56

Respondent: Yeah, I think with, you know, the background knowledge of how ICT can help and my background skills in IT, I continue to show the other teachers how to integrate ICT through training meetings, you know such training will boost teachers confidence. You can't just say ICT is going to help, you need evidence, you need proof and with that you have a stronger argument, so I think having that base knowledge, it's going to put teachers in the department in a stronger position to use ICT.

Interviewer: Thank you. The outcome of this will be publicised, so in case you're interested in it, I can let you have a copy of the research.

07:07-07:08

Respondent: Yeah, alright, yep.

Appendix17: Interview Transcript 3 _Teacher 3

00:00-00:03

Respondent: I can't guarantee that I'll be able to answer your questions, but ... I will try.

00:04-00:08

Interviewer: Please try your very best. These questions are to assist me complete my dissertation in the Phd degree in Mathematics Education. Please feel free to let me know

when you want to pull out of this interview at any stage. No information from you will be given to a third party without your permission.

00:09 - 00:18

Interviewer: Ok, so, what do you believe to be the benefits of using ICT in mathematics?

00:19-00:40

Respondent: It connects well to the majority of student's preferring to learning using ICT due to the interactivity and instant feedback to class work...

00:48 - 01:04

Interviewer: Good. What do you believe are the obstacles then for ... integrating ICT in your pedagogy?

01:04-01:13

Respondent: Yeah, I don't see any obstacles. The only obstacle is when it's not working ...

01:13 - 01:17

Interviewer: You mean the hardware isn't working or is it the issues with the software?

01:17-01:54

Respondent: Yeah, when you think about the work to be done and the poor internet connection is not functioning or there's been a, some kind of internet problem, yeah, obviously there are problems with monitoring children's work due to lack of the software to monitoring them. Children are carried away by doing other task or playing games on the computer.

01:54 - 02:10

Interviewer: But in terms of the software or, lets say, a teachers ability, if the teacher hasn't got the expertise in using ICT, don't you see it as an obstacle?

02:10-02:19

Respondent: Yes, that's an obstacle as well, yeah. Yes, there is quite a problem with expertise, yeah. So that is yes.

02:19 - 02:29

Interviewer: But in terms of the curriculum, what do you think? Doesn't it affect the way the curriculum is delivered?

02:29-03:10

Respondent: No, no it doesn't, it actually does the opposite. Not only that, but when it comes to real-life examples you can show video clips like 'science in the real world', just to ... just to help the students connect better than ..., because they can see its application in the real world.

05:55 - 06:11

Interviewer: Ok, so, do you think a good leadership on ICT is really a necessity in raising people's achievement and how do your think teachers can be supported to integrate ICT?

06:12-07:25

Respondent: As I've said before it's really down to personal preference. Preference but, I guess it's to do with preference and as I said, the joys of the leader in using ICT. An interactive whiteboard in itself is, you know, it's an interactive activity that a download going on between yourself and the students. There can be good at learning using ICT interactive tools with a teacher who knows what he is doing.

09:56 - 09:14

Respondent: Training sessions help such teachers to overcome their difficulties. Yeah, that's one thing that needs to be acknowledged than to discourage them from using ICT. Although it's very rare that teachers haven't got the experience of ICT that can't be used. For example, using the spreadsheet or to analyse data or to input test scores onto the computer. These are simple tasks that every teacher can do. The more completed task should be learned through training sessions provided by the subject leaders if they know or ICT expects can be invited to train all teachers who are lacking the required skills needed for teaching science.

10:49-10:55

Interviewer: But have schools really got the time... got those training programmes in place, to help subject teacher?

10:55 - 11:25

Respondent: I think that maybe there are schools that don't really know where to begin and they kind of, lack motivation, so that is schools coming together and working together. Our school sometimes tap into experienced teacher's expertise and ask them to offer insets and share good practice.

11:25-11:3

Interviewer: But the problem with schools is that there's no follow-up, so teachers don't really benefit from those insets or what do you think the impact of these training sessions are?

11:36 - 12:20

Respondent: Yeah, the follow-up is on time, yeah. It can be a problem. The follow-up can be a problem. And sometimes that's why working in teams or pairs does help us teachers. Working in pairs does help, like sometimes, I feel like I ... I might know how to think about how the lesson will go and then my colleagues however, there might be a wizard on a computer, so we will work together all the time and then we'll get together and work on something substantial together, you know.

Appendix 18: Interview Transcript 4 – Teacher 4

00:01-00:10

Interviewer: Good morning. These questions are to assist me complete my dissertation on a Phd degree in Mathematics Education. Please feel free to let me know when you want to pull out of this interview at any stage. Also be informed that no information from this interview will be given to a third party without your permission.

Interviewer: What do you believe to be the benefits of using ICT, in teaching mathematics?

00:13 - 00:49

Respondent: Well, personally I believe the benefits of integrating ICT is uncountable. It makes my planning of lessons easier, I can teach using power points and online resources. I can make my lessons more interactive and by using ICT tools am able to plan and teach lessons to meet all the different learning styles in my class. Through ICT new skills and concepts are introduced. Also, when using ICT, I am able to promote more interest in the learning and this helps me with the classroom management and reinforces the learning.

Respondent: And because of that, they will have studied better, they have working knowledge of what they are doing, and they, it will increase their attainment.

01:23 - 01:30

Interviewer: Okay. So do you think without ICT, the people's attainments cannot be increased?

01:31 - 01:52

Respondent: Well, without ICT guess what I could to as a teacher? I will not be able to assess and obtain instant feedback from completed student work. You know learning can be very boring. ICT gives me the chance to differentiate my teaching and so it is a good tool to raise attainment of most children at the same time. Raising attainment would not be rapid without the use of ICT tools in lessons.

02:03-02:10

Interviewer: Ok my next question. What do you believe are the obstacles for integrating ICT as a subject leader of Mathematics in your school?

02:13 - 02:14

Respondent: The software. In my experience the barriers include insufficient number of computers, insufficient number of internet-connected computers, Insufficient Internet bandwidth or speed, sometimes and the lack of adequate skills of personals to support teachers. Also there is the restriction on the school's timetable. The time is fixed with lesson time and the inability to find time for training is a concern for most teachers.

03:20 - 03:25

Interviewer: What do you perceive are the facilitators to integrating ICT into your department's pedagogy?

03:25 - 03:40

Respondent: I think among the supporters or drive of ICT in my school will be the personal or any person who is more experienced and knowledgeable to drive the initiative forward. The technology co-ordinator is able to design programs to drive the agenda. The school can employ people externally to drive the initiative. This can be done through online coaching services.

03:40 - 03:50

Interviewer: what about teachers' experience? How does your experience of using ICT impact the use of ICT in your school?

04:30 - 04:53

Respondent: I have had difficult times using ICT due to lack of sufficient knowledge in ICT from the teacher training school that I attended. Less ICT was used when learning maths so I find it more comfortable just sticking to what I already know. In my school, using ICT in teaching and learning has not being a priority. I guess as long as the children achieve their targets, the school was happy with you. For me as a leader of maths, I am also happy as long as the department does well.

04:53-05:59

Interviewer: So, teachers who are not confident... Is there any program to support them?

05:00 - 05:05

Respondent: Sometimes we do have a program in the department to support teachers. For that we do, but there is the lack of time to learn and gain the required confidence. Occasionally we train teachers to use new software properly.

Interviewer: How do you view whole school leadership on ICT integrated learning?

05:30 - 06:40

Respondent: The priority of the school is to perform well against schools in the catchment area so I guess integrating ICT is not a big drive in the school. I may be wrong, but often the focus is on outstanding lessons and it does not matter how a teacher does it. Usually, at the start of the term, there are series of training including using ICT across the curriculum. However, you don't hear anything about it again during the term and there is no clear direction on how the focus on ICT is champion in the school.

06:30 - 06:50

Interviewer: How would you view your leadership role on integrated ICT pedagogy in your department?

06:50 - 07:20

Respondent: I have the passion to integrate ICT but I don't have the skills and confident to provide the leadership drive on it. I have promoted department trainings to motivate the team in the department. However, lack of consistencies with the whole school policy on ICT makes it difficult to measure the outcomes of ICT use.

07:42-07:45

Interviewer: Thank you and that's the end of the interview.

Appendix 19: Interview Transcript 5 – Teacher 5

00:01-00:15

Interviewer (**A**): Hi. These are interview questions to aid me write a dissertation, in partial fulfilment of my PhD degree in mathematics Education. You may feel free to pull out of this interview at any stage if you are not comfortable with the questions. I can assure you that no information from this interview will be given to a third party without your permission.

00:15-00:25

A: What do you – What do you believe to be the benefit of using ICT in your Mathematics education?

00:25-00:51

B: OK, I would say the benefit; the main benefit of using ICT is that it offers a greater degree of interaction with the students. It offers a greater range of interaction than a conventional whiteboard and black pen. I would say to you that it offers a greater variety and, if you like, breadths and depths of interaction.

02:23-02:31

A: What are the obstacles or barriers to integrating ICT?

02:31-03:06

B: Oh, great! So, you can put that—you can say one of the obstacles, the barriers to ICT, is even though an interactive whiteboard has many functions, it's not personalized enough. One

of the barriers is there should be greater use of radio-controlled pads, so that—or linked notebooks for the children so that they can directly interact with the teacher at the front. Also, the time and space available on the schools time table and the lack of interest of the school leaders are creating barrier to the ICT integration in the secondary schools' pedagogy.

03:24-03:52

B: No, I mean the national curriculum could facilitate—I mean you could use ICT in *any* Mathematics lesson. For example, if you're teaching geometry, yeah? If you're introducing, you know, straight line theorems, intersecting line theorems at Key Stage 3— you can use, you can use Geo—GeoGebra what are they called?

06:56-07:10

A: Yeah, but with this, what would you think about the teachers and personal experience? If a teacher isn't so confident in using ICT, do you think that would promote ICT integration?

07:50-08:08

A: But as a subject leader, if you are not confident with ICT, are you sure that your leadership on ICT in the department would be beneficial to other members of staff?

08:09-09:32

B: Yeah...I think as a head of department, you don't necessarily have to be the best person on every piece of software in the department. Yeah? I think it helps if you have those skills, but if you don't have those skills, then I would have champions within my department. So, for instance, if you had, Daniel, I don't know, me in your department, and I was hot on Cabri, you'd let me run the training and take charge of training people within the department. You know? And I think sometimes it doesn't happen because people are – we are too ambitious sometimes, and what we should do is maybe start off with, like, planning a small part of the curriculum using ICT, and developing those materials and storing it, rather than trying to do the whole lot. We have to start small and then build big. But I think within – that's where leadership of learning is important. Because as a leader, you're not the person that necessarily does execute, but you're the person that makes sure those things get done.

09:33-09:39

A: But if you're a leader who doesn't believe in...you see, there are some leaders who don't

believe in ICT, so would they be able to provide leadership on something they don't believe in?

09:40

B: Leading doesn't mean that you do everything.

09:43-09:45

A: You have to believe in it before you use it. Isn't?

09:47-10:52

B: I'm saying good leadership means you've got to distribute the leadership within your team. Now, you know, if you want somebody to be an expert in Geometer's Sketchpad, you could give them time to do some training, give them time to do... to sort of send them on a course, do you understand what I'm saying? And then they have that time back in here to pay it back somehow by taking part of a departmental meeting, by rolling out some new materials well in advance, getting some feedback from the kids and from the students about how it went. There's no point doing it if it's not working. The whole point of doing it, of implementing the change, is that it's going manifest itself in the children's understanding, isn't it? So, you know it's – but I do agree with you that being a good role model is good, but I know guys who are good themselves, but nobody in the department ever can keep up with them, and it doesn't happen. And I've seen situations where people have been in the department, and been more distributive leaders, and it's been far more effective.

10:53-11:09

A: But here is a case where the leader hasn't got any motivation to use ICT. So do you think he would promote any lead at all on teachers using ICT? He wouldn't.

11:10-11:15

B: No, he wouldn't, but that would be a problem for Senior Management, wouldn't it?

11:15-11:26

A: So it takes a leader who has got the charisma or who are interested in promoting ICT to really promote ICT in the department, otherwise ICT will be dormant. Is that right?

11:26-12:00

B: Basically, ICT is a tool. Just like pens, it's a tool. It's not a god, it's a tool. And I would

say anybody who's interested in the teaching of maths to children has got to being using these tools. I mean, look at all of us, we're using mobile phones now. You go to the – you can't even pay your bus fare. If you've got money, you can't even go on the bus. We're using technology more and more in society, and I think they're in danger, if we don't use it in the classroom, we're going to actually lose the children.

14:52-15:17

A: That is very good. So those departments which haven't got leaders, who are so much into ICT, are there any professional developments which are taking place, in your opinion, which are preparing leaders to be able to integrate ICT in their lessons? Do you know of any professional developments happening around schools?

15:18-16:01

B: Oh, yeah, I mean in schools, I mean really there was a little bit of a wave but it seems to have died a death, and one of the reasons why it's died a death is because there isn't the money around. But I think, ideally, new schools are doing a lot of ICT training because they're buying products that are new, aren't they? They're buying products that are new. So sometimes when you go to a school, if you get a job in the school – and this is something you need to think about when you move schools – it's good to move to a school which just introduced some ICT. Because it means that nobody knows how to use it, and you will get a certain amount of time for training.

16:02-16:17

A: But wouldn't you just say that subject leaders who have had poor training from using ICT, example through PGCE or wherever, would have poor impact on the way they would use ICT in lessons?

16:17-17:36

B: Oh, yeah, obviously, if – I'd be very surprised if at PGC E they hadn't talked about, on a mass education course, they hadn't talked about the use of ICT. In the 21st century. I'd be very surprised. But I'm not saying that they all do. But I would have thought that on a teacher training course, they're going to discuss ideas like group work, individualized learning programs – which seems to be making a comeback. In the old days it was Smile but now it's MyMaths, they are making a comeback but they're kind of making a comeback in a supportive role. You know what I mean? They're used as homework. And that's ICT. I was
at a school in Southwold today, and they actually had the MyMaths book. I'd never seen that before. MyMaths actually produce a book. And what they've cleverly done, they've matched their chapters to the topics they already have. So the idea is, if you buy into MyMaths, the book, you get exactly the same topics on the system.

17:59–18:23

A: Yeah, they've made lots of money. And lastly, how would you assess the impact of, let's say, local training, mentoring leaders who haven't got the ICT expertise? How would you assess the impact against schools who would be interested in bringing in external trainers?

18:24-21:37

B: I think one of the problems that there has been – and it's like anything – is having a coherent plan of implementation. We know that ICT has tremendous potential, but the problem that arises is how people, you know, how they implement the professional development, and do they actually review it, and then complete that cycle? Do you know what I mean? Reviewing the plans are the problem. People often introduce something, and they don't think carefully about how to monitor it. They just rush to introduce something. I mean, what I would say is, within the department, maybe, depends on how big the department, but you could start off with a champion, you know what I mean? Like, one person in the department who has some responsibility for that. And the way you could sell that to them as something that they could put on their CV. Yeah?

21:39-21:42

A: We have come to the end of the interview. Is there anything you would like to add?

21:43-22:45

B: I've given you a few ideas, isn't?

22:46-22:48

A: Alright. Yeah, that's fine, thank you.

Appendix 20: Teachers questionnaires

Objective:-

To collect information on the impact of using ICT in mathematics teaching in secondary schools in London, partial fulfilment of PhD Degree in Mathematics education.

Note:

- All answers will be kept confidential.
- You will be given the copy of the analysis if interested for your information.
- All teachers' identities would be protected.

Section A: Please tell us About Yourself.

Please tick one answer in each question.

Q1.Gender: [] MALE [] FEMALE

Q2. Age:	Q3 Position/teaching role:
----------	----------------------------

Under 30 years	30-40	Above 40

Q4.Teaching Experience

Under 5 years	5 – 10 years	Above 10

Q5. Educational Qualification

Diploma	Degree	Masters

Section B: use of ICT for learning mathematics.

Q6. How will you rate you ICT experience?

Below average	Average	Above Average

Q7. How often do you use ICT in teaching Mathematics?

Never	Sometimes	Always

<u>Please tick one Strongly SA = Agreed</u>, A = Agreed, D = Disagreed SD=Strongly Disagreed

Q8. What do you perceive to be the benefits of integrating ICT pedagogy in mathematics?

Sustained student						Aid le	sson	planr	ning		Impro	ved t	he		Enhance interactivity					
interest.				with wide range of						attainı	nent	of		and exploration.						
						resour	ces.				childr	en.								
	SA	Α	D	SD		SA	А	D	SD		SA	А	D	SD	SA	А	D	SD		

Q9 What do you perceive to be the barriers of integrating ICT pedagogy in

mathematics?

Lack of curriculum					Software and						of exp	perier	nce	Negative attitudes.				
resources				Hardware problems						with ICT tools								
SA	A	D	SD	S	SA	А	D	SD		SA	A	D	SD	~ 1	SA	А	D	SD

Q10. What do you perceive to be the facilitators of integrating ICT pedagogy in

mathematics?

The A	vaila	bility	v of		Exper	ience	and			Schoo	ol lead	dersh	ip	Opportunities for					
ICT tools					persor	nal kr	nowle	edge						trainir	ng				
					of usin	ng IC	T												
SA	А	D	SD		SA	А	D	SD	T	SA	А	D	SD	SA	А	D	SD	ſ	

Tables

Table 1 Mathematics teachers: interviews

 Table 2 Mathematics teachers:
 Questionnaires

 Table 3: Computer sharing

Table 4: Summary data analysis using SPSS